

## Registration of a Charge

Company Name: JANE STREET EUROPE LIMITED

Company Number: 05903707



Received for filing in Electronic Format on the: 11/04/2022

**Details of Charge** 

Date of creation: 01/04/2022

Charge code: **0590 3707 0008** 

Persons entitled: THE BANK OF NEW YORK MELLON, LONDON BRANCH

Brief description: N/A

Contains fixed charge(s).

Contains negative pledge.

#### **Authentication of Form**

This form was authorised by: a person with an interest in the registration of the charge.

### **Authentication of Instrument**

Certification statement: I CERTIFY THAT SAVE FOR MATERIAL REDACTED PURSUANT

TO S. 859G OF THE COMPANIES ACT 2006 THE ELECTRONIC COPY INSTRUMENT DELIVERED AS PART OF THIS APPLICATION FOR REGISTRATION IS A CORRECT COPY OF THE ORIGINAL

INSTRUMENT.

Certified by: ED PARKER



# CERTIFICATE OF THE REGISTRATION OF A CHARGE

Company number: 5903707

Charge code: 0590 3707 0008

The Registrar of Companies for England and Wales hereby certifies that a charge dated 1st April 2022 and created by JANE STREET EUROPE LIMITED was delivered pursuant to Chapter A1 Part 25 of the Companies Act 2006 on 11th April 2022 .

Given at Companies House, Cardiff on 19th April 2022

The above information was communicated by electronic means and authenticated by the Registrar of Companies under section 1115 of the Companies Act 2006





# The Bank of New York Mellon Framework Agreement



## **Table of Contents**

Cont	tents	Page
1	Scope	3
2	Definitions	3
3	Appointment of Service Providers	3
4	Constitution of Service Agreements	3
5	Amendments to the Service Form	4
6	Authorisations	4
7	Representations and Warranties Applicable to all Service Agreements	5
8	Fees, Charges, Costs and Expenses	6
9	Termination on Notice	7
10	Exclusion and Restriction of Liability	8
11	Use of Affiliates	9
12	Force Majeure	9
13	Waiver of Immunity	9
14	Severability	9
15	Notices	9
16	Conditions Precedent to Custody	11
17	Amendments to Eligible Collateral	11
18	Further Assurances	11
19	Notice of Security	11
20	Miscellaneous	12
21	Governing Law and Jurisdiction	14
22	Process Agent Appointment Error! Bookmark not defi	ned.
Sche	edule 1 to the Framework Agreement - Service Form	17
Sche Term:	edule 2 to the Framework Agreement - BNYM Legal Entity Specific Framework Agreement is 19	
	edule 3 to the Framework Agreement - Amendments and Supplements to the Service ements	21

Schedule 4 to the Framework Agreement - Agency	Terms

This Framework Agreement is made on 1st April 2022 between:

(1) JANE STREET EUROPE LIMITED, ("you");

#### (2) THE BANK OF NEW YORK MELLON, LONDON BRANCH ("BNYM, LB")

and relates to the provision of services to you (the "Services").

By executing and (to the extent applicable) delivering this deed, each party specified above other than you has entered into a separate deed with you on the terms set out herein as amended by any BNYM legal entity specific terms applicable to such party as set out in Schedule 2 (BNYM Legal Entity Specific Framework Agreement Terms) to this deed (each a "Framework Agreement" and the party to such Framework Agreement other than you, the "Service Provider").

This deed witnesses and it is declared as follows:

#### 1 Scope

This Framework Agreement sets out the basis on which the applicable Service Provider will provide the relevant Services to you.

#### 2 Definitions

Terms used in this Framework Agreement but not otherwise defined herein shall have the meaning given to them in the applicable Service Module (as defined below).

#### 3 Appointment of Service Providers

You hereby appoint the Service Provider to perform its obligations in respect of the relevant Services as described in the applicable Service Agreements (as defined in Clause 4 (Constitution of Service Agreements) entered into by you and the Service Provider pursuant to Clause 4 (Constitution of Service Agreements).

#### 4 Constitution of Service Agreements

You and the Service Provider agree that by execution and (to the extent applicable) delivery of this Framework Agreement, a service agreement shall come into effect between you and the Service Provider in respect of each of the service modules elected to apply to you in respect of the applicable Service Provider as set out in the service form (the "Service Form") in Schedule 1 (Service Form) to this Framework Agreement, as amended from time to time pursuant to Clause 5 (Amendments to the Service Form) (each a "Service Agreement"). The terms of the Service Agreement in respect of each applicable service module shall consist of:

- (a) such service module as amended from time to time (each a "Service Module");
- (b) Clauses 6 (*Authorisation*) to Clause 18 (*Further Assurances*) of this Framework Agreement inclusive, as amended from time to time (the "**Central Terms**");
- (c) the regulatory terms set out in the Service Form as amended from time to time (the "Regulatory Terms"); and
- (d) the operational terms set out in the Service Form as amended from time to time (the "Operational Terms"),

as amended by the amendments and supplements set out in Schedule 3 (Amendments and Supplements to the Service Agreements) and Schedule 4 (Agency Terms) to this Framework Agreement.

In addition, the provisions of this Framework Agreement relating to construction of certain references and the terms of Clause 20 (*Miscellaneous*) shall apply to the Service Agreement as if set out therein.

Subject to Clause 5 (*Amendments to the Service Form*), each such Service Agreement shall be effective as of the applicable Effective Date set out in the Service Form.

#### 5 Amendments to the Service Form

5.1 To add or remove Services with existing Service Provider: To add or remove Services to be provided to you from time to time, you and the Service Provider shall agree an amendment and restatement or a supplement to the Service Form in writing.

Where an amendment and restatement or a supplement to the Service Form is agreed in writing, you and the Service Provider agree that a service agreement shall come into effect between you and the Service Provider in respect of each of the new service modules elected to apply to you in respect of the applicable Service Provider as set out in such amendment and restatement or a supplement to the Service Form.

The terms of the Service Agreement in respect of each applicable new service module shall be as set out in Clause 4 (*Constitution of Service Agreements*).

Subject to Clause 5 (*Amendments to the Service Form*), each such Service Agreement shall be effective as of the applicable Effective Date set out in the Service Form.

Service Agreements in respect of Services not subject to the amendment and restatement or a supplement shall continue in full force and effect.

**5.2 To add a new Service Provider**: To add a new service provider not party to this agreement (each a "**New Service Provider**"), you and the New Service Provider shall execute a new deed on the terms of this deed (an "**additional deed**").

By executing and (to the extent applicable) delivering such an additional deed as a deed, each New Service Provider shall have entered into a separate deed with you on the terms set out herein as amended by any BNYM legal entity specific terms applicable to such party as set out in Schedule 2 (BNYM Legal Entity Specific Framework Agreement Terms) to this deed (each a "Framework Agreement" and the party to such Framework Agreement other than you, the "Service Provider").

Service Agreements in respect of you and the existing Service Providers shall continue in full force and effect.

#### 6 Authorisations

You authorise the Service Provider to give instructions and provide information (including financial information) concerning you to any of its affiliates or any third party who the Service Provider reasonably believes to be acting on your behalf, and such affiliate or third party shall be entitled to rely on any such instructions or information without further enquiry.

#### 7 Representations and Warranties Applicable to all Service Agreements

- **7.1 General representations and warranties applicable to all Services**: you and the Service Provider each represent, warrant and undertake to the other that:
  - (a) it is duly organised and existing and in good standing under the laws of its jurisdiction;
  - (b) it has all necessary power and is duly authorised to execute and deliver, and perform its obligations under, the Service Agreement and has taken all necessary action to authorise such execution, delivery and performance;
  - (c) the person accepting the Service Agreement, and (with respect to you only) any Authorised Person providing the Service Provider with Instructions in accordance with the Service Agreement, is duly authorised to do so on its behalf;
  - (d) it has obtained all authorisations of any government or regulatory body required in connection with the Service Agreement and all such authorisations are in full force and effect;
  - (e) the execution, delivery and performance of the Service Agreement will not violate any law, statute, regulation, order, ordinance, charter, by-law or rule applicable to it or any agreement by which it is bound or by which any of its assets are affected; and
  - (f) the Service Agreement constitutes its legal, valid and binding obligations enforceable in accordance with its provisions.
- **7.2** Additional representations and warranties applicable to a recipient of custody services: Where the Service Module is a Custody Service Module (as defined in the Service Form), you represent, warrant and undertake that:
  - (a) the Securities and Cash are, and will remain during the term of the Service Agreement, free and clear of all liens, charges, security interests and encumbrances (except for those granted or disclosed in the Service Agreement and applicable Service Module);
  - (b) where applicable, your client has consented to the liens, charges, security interests and encumbrances granted or disclosed in the Service Agreement and applicable Service Module; and
  - (c) you:
    - (i) have established and maintain policies and procedures (a copy of which will be provided to the Service Provider on request) which require that you obtain and verify information about the identity of persons on whose behalf you are acting and which are reasonably designed to ensure that you are not being used by any such other person as a conduit for money laundering or other illegal or illicit purposes; and
    - (ii) have verified and recorded the identity of each person on whose behalf you are acting and where reasonably required (and to the extent permitted) by the Service Provider or any legal or regulatory authority, shall upon request furnish the Money Laundering Reporting Officer of the Service Provider with all relevant documentary evidence of the identity of each such person. To the best of your knowledge, no transaction undertaken in respect of the Account is prohibited by applicable law, regulation or rule and no Property held in the

Account is derived from any activity prohibited by applicable law, regulation or rule.

- 7.3 Additional representations and warranties applicable to a recipient of collateral management services: Where the Service Module is a Collateral Service Module (as defined in the Service Form), the Collateral Provider or the Collateral Receiver, as applicable, further represents, warrants and undertakes that:
  - (a) the appointment of the Service Provider as its agent pursuant to the Service Agreement has been duly authorised and no other corporate action is required prior to the Service Provider acting pursuant to the Service Agreement;
  - (b) subject to the application of Schedule 4 (Agency Terms) to this Framework Agreement, it is entering into the Service Agreement and the Deals for itself as principal and not as agent for another person;
  - (c) any Authorised Person providing the Service Provider with Instructions in accordance with the Service Agreement, is duly authorised to do so on its behalf;
  - (d) except as set out in Clause 20.1 (*Variation*) of this Framework Agreement (as applied to the Service Agreement), it will not transfer or assign its interest in, or rights with respect to, any Deal and is acquiring the same for its own account; and
  - (e) at the time of transfer of any Eligible Collateral or Securities (as applicable), it will have full and unqualified right to make such transfer and, upon such transfer, the Collateral Receiver or Collateral Provider (as applicable) will receive such Eligible Collateral or Securities (as applicable) free and clear of any lien, claim, charge, encumbrance or other security interest.
- 7.4 Additional representations and warranties of the Service Provider: Where the Service Provider is providing services pursuant to a Collateral Service Module (as defined in the Service Form) or a Custody Service Module (as defined in the Service Form) the Service Provider represent, warrants and undertakes that it shall perform such custody and collateral management services using the level of skill and care that would reasonably be expected of a professional custodian (in accordance with CASS 6.3.1 of the FCA's Handbook) or collateral manager operating in the relevant market.

#### 7.5 Repetition of representations and warranties

The representations and warranties in Clause 7.1 (General representations and warranties applicable to all Services) and Clause 7.3 (Additional representations and warranties applicable to a recipient of collateral management services) shall, where they are applicable to the relevant Service, be deemed to be repeated on each day.

#### 8 Fees, Charges, Costs and Expenses

Where applicable, you shall be responsible for the Service Provider's proper and reasonable fees and charges and you shall reimburse the Service Provider for all costs and expenses incurred by the Service Provider in connection with the Service Agreement, unless otherwise agreed.

Fees and charges will be determined in accordance with the Service Provider's rates in effect at the time the charges are incurred and in accordance with the Fee Schedule set out in Part A (Fee Schedule) of Schedule 3 to the Operational Terms or as otherwise notified to you, verbally or in writing. The Fee Schedule may be amended from time to time by the Service

Provider upon thirty (30) days' prior written notice to you. Any alteration to these charges will be notified to you prior to such change becoming effective.

All amounts payable by you to the Service Provider shall be due and payable on demand without set-off, counterclaim or deduction.

Fees and charges and reimbursement for costs and expenses shall be paid periodically as agreed in writing between the parties.

The Service Provider may debit any Cash Accounts (if any) for such fees, costs and expenses.

#### 9 Termination on Notice

**9.1 General termination right**: You or the Service Provider may terminate the Service Agreement, or the provision of a Service, by giving notice of termination to the other party in writing, which will take effect 30 calendar days after the date on which the other party receives such notice.

The Service Provider may terminate the Service Agreement, or the provision of a Service, immediately upon notice to you upon your dissolution, if you are a body corporate or partnership, or upon the commencement of any action or proceedings seeking administration, liquidation, winding-up, insolvency, reorganisation or other similar relief in respect of you or your debts under any insolvency or analogous proceedings in any jurisdiction where such commencement or initiation of actions or proceedings has not been stayed, dismissed or cured within 15 days of initiation or commencement.

Upon termination under the Service Agreement, the Service Provider and you will fulfil any outstanding obligation agreed to but not settled before the date of the termination.

Unless the Service Agreement expressly provides otherwise, any termination of the provision of a Service will not affect the continuation of any other Service not terminated and the Service Agreement in respect of such Service will continue in full force and effect.

Any termination of the Service Agreement will not affect accrued rights under the Service Agreement, or any provision of the Service Agreement intended to survive termination which for the avoidance of doubt will include the fees, the indemnity and limitation of liability provisions set out in the Service Agreement and the terms of the Custody Service Module until such time the delivery of all Property held by the Service Provider has been carried out pursuant to clause 9.2 below.

9.2 Additional consequences of termination with respect to custody services: Where the Service Module is a Custody Service Module (as defined in the Service Form), upon termination of the Service Agreement and payment of all amounts due and owing to the Service Provider, the Service Provider shall, upon receipt of your Written Instructions (such Written Instructions to include details of the account to which the relevant Property shall be delivered), deliver the Property and all records relating to the Property pursuant to your Written Instructions within a reasonable timeframe following receipt of such Written Instructions. You shall be responsible and liable for any shipping and insurance costs associated with such delivery.

Upon termination, provisions set out in the relevant Fee Schedule that are conditional on termination shall apply. These may include an increase to the amount of fees we charge you.

#### 10 Exclusion and Restriction of Liability

- Agreement, the Service Provider shall not be liable for any and all losses including but not limited to claims liabilities, damages, costs, expenses, penalties, actions, demands, settlements, investigations, proceedings and judgments (including legal and other professional advisers' fees and expenses) sustained by a person ("Losses") resulting from its action or inaction in connection with the Service Agreement except for those Losses arising out of the negligence, fraud, or wilful default of the Service Provider, an affiliated company or a nominee company controlled by the Service Provider or an affiliated company. In no event, whether for negligence, breach of contract, misrepresentation or otherwise, shall the Service Provider be liable to you or any third party for any special, indirect or consequential damages, or any loss of profits, business or opportunity (whether direct or indirect in nature), arising under or in connection with the Service Agreement.
- **10.2** Losses: Without prejudice to the generality of Clause 10.1, none of the Service Provider, or any affiliate, shall be responsible or liable for any Losses arising out of or relating to, directly or indirectly:
  - (a) (i) any adverse tax, deductions, accounting or other implications of any transaction whatsoever and (ii) any inability of the Service Provider, an affiliate of the Service Provider, any Sub-custodian or any of their respective agents to file claims for exemptions or refunds or otherwise obtain relief from Tax Obligations due to (A) your failure to provide, or delay in providing, Tax Information to the Service Provider, (B) any failure of yours to comply with applicable tax laws, or (C) any failure or refusal of any taxing authority to provide such relief;;
  - (b) any act or omission of yours, including any error, negligence or misconduct of yours;
  - (c) the accuracy of any information provided to you which has been obtained from or provided to the Service Provider by any other entity; or
  - (d) receiving or transmitting any data to or from you, your Authorised Person or your investment manager via any non-secure method of transmission or communication.

The Service Provider may, with respect to questions of law, apply for and obtain the advice and opinion of counsel, and shall not be liable for any Losses suffered as a result of anything done or omitted by it in good faith in accordance with such reasonable advice or opinion.

- **10.3** Additional limitation of liability applicable to custody services: Where the Service Module is a Custody Service Module (as defined in the Service Form):
  - 10.3.1 The Service Provider shall take appropriate action to recover any Losses incurred by you as a result of the acts or failure to act by a Sub-custodian and any liability of the Service Provider for the acts or failure to act of a Sub-custodian (that is not an affiliated Sub-custodian) shall be limited to the amounts so recovered, after deduction of costs and expenses incurred by the Service Provider.
  - 10.3.2 Limbs (a) to (d) of Clause 10.2 shall be supplemented by the following as a new limb (e):

"the receipt or acceptance of fraudulent, forged or invalid Securities (or Securities which are otherwise not freely transferable or deliverable without encumbrance in any relevant market)".

#### 11 Use of Affiliates

You agree that the Service Provider may use other The Bank of New York Mellon entities and other third parties in connection with its performance of the services and any other obligations under the Service Agreement and in certain other activities, including, without limitation, audit, accounting, tax, administration, risk management, credit, legal, compliance, operations, sales and marketing, relationship management, information technology, records and data storage, performance measurement, data aggregation and compilation and analysis of information and data regarding you, your affiliated companies and associates and any accounts you hold with The Bank of New York Mellon. This Clause 11 shall survive the termination of the Service Agreement.

#### 12 Force Majeure

Neither the Service Provider nor any affiliate shall be liable to you, or have any responsibility for any Losses incurred or suffered by you, for the non-performance, partial performance or delay in performance of any of its obligations thereunder caused by any cause beyond the control of the Service Provider or any of its affiliates (a "Force Majeure Event"). Where a Force Majeure Event occurs that prevents or delays the performance by the Service Provider and/or any of its affiliates of any of their obligations under the Service Agreement, (i) all such obligations shall be suspended for the duration of the Force Majeure Event; and (ii) any such obligation that would, but for the Force Majeure Event, have fallen due shall not fall due, or be deemed for any purpose to fall due, for the duration of the Force Majeure Event.

#### 13 Waiver of Immunity

You irrevocably waive, with respect to you and your revenues and assets (irrespective of their use or intended use), all sovereign or other immunities and privileges to which you or your revenues or assets might otherwise be entitled in the courts of any jurisdiction in any suit, action or proceeding relating to any dispute (including, without limitation, immunity from (a) suit and legal process, (b) jurisdiction of any court, (c) relief by way of injunction or order for specific performance or recovery of property, (d) attachment or seizure of your assets whether before or after judgment, and (e) execution or enforcement of any judgment or award by any means). You consent to the grant of such relief in any form and irrevocably agree that you will not claim any such immunity or privilege in any suit, action or proceeding relating to any dispute.

#### 14 Severability

If at any time any provision of the Service Agreement becomes, or is deemed by an authority of competent jurisdiction to be, invalid, unenforceable or contrary to Applicable Regulations (as defined in Clause 20.1), neither the legality, validity or enforceability of the remaining provisions of the Service Agreement nor the legality, validity or enforceability of such provision under the law of any other jurisdiction shall in any way be affected or impaired by such provision.

#### 15 Notices

Unless otherwise specified in the Service Agreement, any notice required to be delivered by you or the Service Provider to the other party pursuant to the Service Agreement shall be

sent by electronic delivery or by courier delivery service (return receipt requested), to the other party to the relevant contact details specified below.

(a) Notices should be sent to the Service Provider at:

The Bank of New York Mellon
One Canada Square London, E14 5AL

Attention: Head of EMEA Legal - Markets

(b) Notices should be sent to you at the address specified in the Service Form or as otherwise notified by you in writing.

Any party may by notice to the other change the address at which notices, or other communications, are to be given to it.

Any notice shall be effective only upon the receipt thereof by the party to whom sent and shall be effective only for the purpose and in the specific instance for which it is given.

Unless otherwise expressed to the contrary, notices shall only apply to the specific Service Agreement that they relate to.

- Notices related to collateral management services: Where the Service Module is a Collateral Service Module (as defined in the Service Form), any notice authorised or required by the Service Agreement shall be sufficiently given if addressed to the receiving party and hand delivered or sent by post or e-mail to the individuals at the addresses specified in Part Al (Notices) Schedule 1 (Collateral Management Master Agreement (Collateral Provider) Service Module Forms of Notice) of the Operational Terms (as amended from time to time) or Part Al (Notices) Schedule 2 (Collateral Management Master Agreement (Collateral Receiver) Service Module Forms of Notice) of the Operational Terms (as amended from time to time) respectively.
- **15.2 Notices related to security**: Where the Service Module is a Security Service Module (as defined in the Service Form):
  - 15.2.1 Any notice or demand served on the Company by the Service Provider hereunder must be written in English and delivered or sent by post or e-mail process to be served in accordance with the contact details provided for the service of notices in Schedule 1 to this Framework Agreement (Service Form) or at any substitute address or department or officer as the Company may notify to the Service Provider by not less than seven days' notice. Any notice sent by post shall be sent by prepaid first class recorded delivery post (if within the United Kingdom) or by prepaid airmail (if elsewhere).
  - **15.2.2** Any notice or demand shall be deemed to have been served:
    - (a) if delivered, at the time of delivery;
    - (b) if posted from within the United Kingdom, at 10 a.m. on the next business day in London following the date of posting; and
    - (c) if sent by e-mail, at the time the e-mail is sent by the Service Provider.
  - 15.2.3 In proving service of a demand or notice it shall be sufficient to prove that delivery was made or that the envelope containing the notice or demand was properly addressed and posted (either by prepaid first class recorded delivery post or by

prepaid airmail, as the case may be) or that the e-mail message was properly addressed, as the case may be.

In the event of any inconsistency with this Clause 15.2.3 and the rest of Clause 15, this Clause shall govern.

#### 16 Conditions Precedent to Custody

Where the Service Module is a Custody Service Module (as defined in the Service Form), the Service Agreement is conditional upon you providing to the Service Provider or the Service Provider obtaining, as the case may be, the documents set out in Part C (Conditions Precedent Documents) of Schedule 3 to the Operational Terms (Dealer Custody (English/Belgian Law) Service Module and Receiver-only Custody (English/Belgian Law) Service Module). In the event that such conditions are not fulfilled, the Service Provider may elect to terminate the Service Agreement whereupon the Service Agreement shall have no further effect and all the liabilities and obligations of the Service Provider and you shall cease.

#### 17 Amendments to Eligible Collateral

Where the Service Module is a Collateral Service Module (as defined in the Service Form):

- (a) A schedule for Eligible Collateral may be amended by matching Written Instructions from the Collateral Provider and the Collateral Receiver, or an agent on their behalf, to the Service Provider specifying the amendments to be made and the date on and from which such amendments are to be effective;
- (b) Where such amendments and effective date are also agreed to by the Service Provider, the Service Provider, may send a confirmation notice to the Collateral Provider, the Collateral Receiver and the Collateral Receiver's Collateral Manager or the Collateral Provider's Collateral Manager, as applicable, of its agreement to such amendment to a schedule for Eligible Collateral and any such amendment to a schedule for Eligible Collateral shall take effect without further formality from:
  - (i) the effective date, if specified within such notice; or
  - (ii) if no effective date is specified, the date that the Service Provider provides such confirmation notice.

#### 18 Further Assurances

You shall do and execute, or arrange for the doing and executing of, each necessary act, document and thing reasonably within your power to implement the Service Agreement.

You will promptly provide to the Service Provider such information as the Service Provider reasonably requests from time to time to enable the Service Provider to comply with any applicable law or regulation, in addition to any information specifically required or requested pursuant to the Service Agreement.

#### 19 Notice of Security

You hereby give notice, and by execution and (to the extent applicable) delivery of this Framework Agreement, each other party hereby acknowledges that it has notice, of the security created pursuant to any applicable Security Service Module.

#### 20 Miscellaneous

**20.1 Variation**: No variation of the Service Agreement or this Framework Agreement shall be effective unless in writing and signed by, or on behalf of, each party.

Notwithstanding the foregoing, where reasonably considered by the Service Provider to be necessary or desirable for compliance with Applicable Regulations, the Service Provider may amend the Service Agreement, the Regulatory Terms or this Framework Agreement with such amendments to be effective immediately upon written notice to you.

"Applicable Regulation" means the rules and regulations of any applicable Regulator, the rules of any relevant exchange and any laws or regulations applicable to the Service Provider in the provisions of the Services to you.

"Regulator" means each of the European Securities and Markets Authority, the European Central Bank, the Financial Conduct Authority, the National Bank of Belgium or any applicable European Economic Area or other competent regulatory authority regulating the Service Provider in any jurisdiction.

Notwithstanding the foregoing, the applicable Service Provider may amend the Operational Terms (together with any documents incorporated by reference therein) at any time upon written notice to you and any such amendment shall become effective on the date specified in the notice, provided that any such amendment will not have any unreasonably detrimental effect on you.

By continuing to accept Services from the Service Provider after receipt of any notice of amendment, you agree to such amendments. Any such amendments will also apply in respect of any obligations owed by either party to the other. If you do not accept any such amendments, you must notify the Service Provider in writing and refrain from dealing with the Service Provider until the terms on which we provide our Services to you are agreed.

- 20.2 Assignment: Neither party may assign, novate, transfer or charge any of its rights or obligations under the Service Agreement or this Framework Agreement without the written consent of the other party (with such consent not to be unreasonably withheld or delayed by the consenting party) provided that the Service Provider may assign or novate its rights, benefits and/or obligations under or in connection with the Service Agreement or this Framework Agreement provided to be assigned or transferred to any affiliate, subject to giving you reasonable notice thereof.
- 20.3 Waiver: No failure to exercise, nor any delay in exercising, any right, power or remedy under the Service Agreement or this Framework Agreement or by law shall operate as a waiver, nor shall any single or partial exercise of any right or remedy prevent any further or other exercise or the exercise of any other right or remedy. The rights and remedies provided in the Service Agreement or this Framework Agreement are cumulative and not exclusive of any rights or remedies (provided by law or otherwise). Any waiver of any breach of the Service Agreement or this Framework Agreement shall not be deemed to be a waiver of any subsequent breach.
- 20.4 Partial Invalidity: If at any time any provision of the Service Agreement or this Framework Agreement is or becomes illegal, invalid or unenforceable in any respect under any law of any jurisdiction, neither the legality, validity or enforceability of the remaining provisions nor the legality, validity or enforceability of such provision under the law of any other jurisdiction will, in any way, be affected or impaired.

**20.5 Counterparts**: This Framework Agreement may be executed in counterparts which, when taken together, shall constitute one instrument.

#### 20.6 Construction of Certain References: References to:

- 20.6.1 an agreement, deed, instrument, licence, code, constitution, legislation, regulation, statute, treaty, list or other document, or to a provision contained in any of these, shall, except to the extent that the context requires otherwise, be construed, at the particular time, as a reference to it as it may then have been amended, varied, supplemented, modified, suspended, replaced, assigned or novated;
- 20.6.2 costs, charges, remuneration or expenses include any value added, turnover or similar tax charged in respect thereof;
- 20.6.3 a Directive includes any relevant implementing measure of each Member State of the European Economic Area which has implemented such Directive and, in relation to the United Kingdom, any relevant provisions of the European Union (Withdrawal) Act 2018;
- 20.6.4 a "person" include any company, partnership or unincorporated association (whether or not having separate legal personality);
- **20.6.5** a company shall include any company, corporation or any body corporate, wherever incorporated;
- 20.6.6 any party (howsoever referred to) include its successors, permitted assigns and permitted transferees, unless the context otherwise requires;
- **20.6.7** a "**judgment**" includes any order, injunction, determination, award or other judicial or arbitral measure in any jurisdiction; and
- 20.6.8 a "law" includes common or customary law and any constitution, decree, judgment, legislation, order, ordinance, regulation, statute, treaty or other legislative measure, in each case of any jurisdiction whatever (and "lawful" and "unlawful" shall be construed accordingly).
- **20.7 Headings**: Headings shall be ignored in construing the Service Agreement or this Framework Agreement.
- 20.8 Schedules: The Schedules to this Framework Agreement, the Regulatory Terms, the Operational Terms or the Service Module are part of such Framework Agreement, Central Terms, Regulatory Terms, Operational Terms or Service Module and shall have effect accordingly.
- **20.9 Inconsistency**: In the event of any inconsistency between:
  - 20.9.1 the Central Terms and the Regulatory Terms, the Central Terms will govern;
  - 20.9.2 the Central Terms and/or the Regulatory Terms and a Service Module, the Service Module will govern; and
  - 20.9.3 the Central Terms, the Regulatory Terms and/or a Service Module and this Framework Agreement (other than the Central Terms), this Framework Agreement will govern.
- **20.10 Applicable Regulations**: Any Services provided to you under the Service Agreement shall be subject to Applicable Regulations such that in the event of any inconsistency or conflict

- between the Service Agreement and any such Applicable Regulations, the Applicable Regulations will govern.
- 20.11 Unfettered discretion: Where the Service Module is a Security Service Module (as defined in the Service Form), then any liberty or power which may be exercised or any determination which may be made under the Security (English law) Service Module by the Service Provider may be exercised or made in the absolute and unfettered discretion of the Service Provider which shall not be under any obligation to give reasons therefor.

#### 21 Governing Law and Jurisdiction

- **21.1 Governing Law**: This Framework Agreement and any non-contractual obligations arising out of or in connection with it are governed by, and shall be construed in accordance with, English law, or, in the case of the Central Terms, the governing law specified in the applicable Service Module.
- 21.2 Jurisdiction: The courts of England, or, in the case of the Central Terms, the courts specified in the applicable Service Module, are to have exclusive jurisdiction to settle any disputes that may arise out of or in connection with this Framework Agreement (including non-contractual disputes or claims) and accordingly any legal action or proceedings arising out of or in connection with this Framework Agreement ("Proceedings") shall be brought in such courts. The parties irrevocably submit to the exclusive jurisdiction of such courts and waive any objection to Proceedings in such courts whether on the ground of venue or on the ground that the Proceedings have been brought in an inconvenient forum.

This deed is delivered on the date stated at the beginning.

JANE STREET EUROPE LIMITED
Ву:
Title: Robert Granieri, Director
By:
Title: Ian Shea, Director
Date:

## EXECUTED AS A DEED BY THE BANK OF NEW YORK MELLON, LONDON BRANCH

ACTING BY	
	Digitally signed
	by Mark Higgins
	Date: 2022.04.01
	16:39:33 +01'00'

#### SIGNATURE OF DULY AUTHORISED SIGNATORY

## Schedule 1 to the Framework Agreement - Service Form

#### The following Service Modules shall to apply to you:

Service Module	Service Provider	Effective Date	
Dealer Custody (English law) Service Module	The Bank of New York Mellon, London branch	As of the date of this Framework Agreement	
Security (English law) Service Module	The Bank of New York Mellon, London branch	As of the date of this Framework Agreement	
Collateral Management Master Agreement (Collateral Provider) Service Module	The Bank of New York Mellon, London branch	As of the date of this Framework Agreement	

#### Relevant Regulatory Terms and Operational Terms

The Regulatory Terms: The regulatory terms dated the date of this Framework Agreement

The Operational Terms: The operational terms dated of this Framework Agreement

#### **Agency Terms**

The Agency Terms in Schedule 4 (Agency Terms) to the Framework Agreement shall not apply

#### **US Securities**

In relation to Securities issued in the United States of America, the Shareholders Communications Act of 1985 (the "Act") requires the Service Provider to disclose to issuers, upon the issuer's request, the name, address and securities position of our customers who are the "beneficial owners" (as defined in the Act) of the issuer's Securities, provided that the beneficial owner does not object to such disclosure. The Act also requires the Service Provider to disclose to issuers, upon the issuer's request, the name and address of our customers who are acting as a "respondent bank" (as defined in the Act) in relation to the Securities. (Under the Act, "respondent banks" do not have the option of objecting to such disclosure upon the issuer's request). The Act defines a "beneficial owner" as any person who has, or shares, the power to vote a security (pursuant to an agreement or otherwise), or who directs the voting of a security. The Act defines a "respondent bank" as any bank, association or other entity that exercises fiduciary powers which holds securities on behalf of a beneficial owner and deposits such securities for safekeeping with a bank, such as the Service Provider. Under the Act, you are either a "beneficial owner" or a "respondent bank."

I am a "beneficial owner" of the Securities to be held by the Service Provider.	
I am not a beneficial owner of the Securities to be held by the Service Provider, but I am acting	g
as a "respondent bank" in relation to the Securities to be held by the Service Provider.	

IF NO BOX IS CHECKED, BNYM SHALL ASSUME THAT YOU ARE THE BENEFICIAL OWNER OF THE SECURITIES.

For beneficial owners of the Securities only:

- [] I/we object to such disclosure
- [] I/we do not object to such disclosure

of the Customer's name, address and securities position to an issuer which requests such information pursuant to the Act for the specific purpose of direct communications between such issuer and Customer.

IF NO BOX IS CHECKED, BNYM SHALL RELEASE SUCH INFORMATION UNTIL IT RECEIVES A CONTRARY INSTRUCTION FROM THE CUSTOMER.

In relation to Securities issued outside of the United States of America, information shall be released to issuers only if required by law or regulation of the particular country in which Property is located.

#### Notices to be sent to you should be sent to:

Client: Jane Street Europe Limited

1) Company Name: Jane Street Europe Limited

2) Address: 2 & A Half Devonshire Square, London EC2M 4UJ

3) Attention: Head of Legal

4) Email: legalnotices@janestreet.com

5) Phone: +12125486640

#### Sovereign Immunity

For Governmental Entities Only: To the extent that, in any jurisdiction, you have acquired or hereafter may acquire, or hereafter may be entitled to claim, for yourself or your assets, immunity (sovereign or otherwise) from suit, execution, attachment (before or after judgment) or any other legal process, you irrevocably agree not to claim, and hereby waive, such immunity.

#### **US Special Resolution Regime**

Notwithstanding Clause 19 of the Framework Agreement or any further terms of the security created pursuant to any applicable Security Service Module, in the event:

- (i) BNYM, LB becomes subject to a proceeding under a U.S. special resolution regime, the transfer of the Framework Agreement and any applicable service modules (and any interest and obligation in or under, and any property securing, the Framework Agreement and any applicable service modules) from BNYM, LB will be effective to the same extent as the transfer would be effective under the U.S. special resolution regime if the Framework Agreement and any applicable service modules (and any interest and obligation in or under, and any property securing, the Framework Agreement and any applicable service modules) were governed by the laws of the United States or a state of the United States; and
- (ii) BNYM, LB or any of its affiliates become subject to a proceeding under a U.S. special resolution regime, default rights with respect to the Framework Agreement and any applicable service modules that may be exercised against BNYM, LB are permitted to be exercised to no greater extent than the default rights could be exercised under the U.S. special resolution regime if the Framework Agreement and any applicable service modules were governed by the laws of the United States or a state of the United States.

## Schedule 2 to the Framework Agreement - BNYM Legal Entity Specific Framework Agreement Terms

Where the Service Provider is BNYM, LB, the following provisions shall apply:

1 Clause 20 (*Miscellaneous*) shall be supplemented by the following provisions:

"Contracts (Rights of Third Parties) Act 1999: Subject only to the extent explicitly set out in the Service Agreement or this Framework Agreement, a person who is not a party to the Service Agreement or this Framework Agreement has no right under the Contracts (Rights of Third Parties) Act 1999 (the "Act") to enforce any term of the Service Agreement or this Framework Agreement respectively.

You acknowledge that any affiliate of any Service Provider may enforce the Service Agreement or this Framework Agreement subject to, and in accordance with, this Clause and the provisions of the Act. The parties to the Service Agreement or this Framework Agreement do not require the consent of any affiliate to rescind or vary the Service Agreement or this Framework Agreement respectively at any time."

2 References to a time shall mean the time in effect on that day in London.

Where the Service Provider is **BNYM SA/NV** the following provisions shall apply:

1 Clause 20.2 (Assignment) shall be amended to include, following the second use of the term "novate" the addition of:

"(in which case, such novation shall benefit from the application of Article 1278 of the Belgian Civil Code)".

- 2 References to a time shall mean the time in effect on that day in Belgium.
- Notwithstanding and to the exclusion of any other term of the Framework Agreement or any other agreements, arrangements, or understanding between the Service Provider and you, you acknowledge and accept that a BRRD Liability arising under the Framework Agreement may be subject to the exercise of Bail-in Powers by the Relevant Resolution Authority, and acknowledge, accept, and agree to be bound by:
  - (i) the effect of the exercise of Bail-in Powers by the Relevant Resolution Authority in relation to any BRRD Liability of the Service Provider to you under the Framework Agreement, that (without limitation) may include and result in any of the following, or some combination thereof:
    - (a) the reduction of all, or a portion, of the BRRD Liability or outstanding amounts due thereon;
    - (b) the conversion of all, or a portion, of the BRRD Liability into shares, other securities or other obligations of the Service Provider or another person, and the issue to or conferral on you of such shares, securities or obligations;
    - (c) the cancellation of the BRRD Liability; and

- (d) the amendment or alteration of any interest, if applicable, thereon, the maturity or the dates on which any payments are due, including by suspending payment for a temporary period; and
- (ii) the variation of the terms of the Framework Agreement as deemed necessary by the Relevant Resolution Authority, to give effect to the exercise of Bail-in Powers by the Relevant Resolution Authority.

For the purposes of this paragraph 3, the following terms shall have the meanings set forth below:

"Bail-In Legislation" means the law of 25 April 2014 on the status and supervision of credit institutions, and any other law or regulation relating to the transposition of BRRD under Belgian law.

"Bail-in Powers" means any write-down, conversion, transfer, modification or suspension power existing from time to time under, and exercised in compliance with, any law or regulation in effect in Belgium, relating to the transposition of BRRD, including but not limited to the Bail-In Legislation and Regulation (EU) No 806/2014 and the instruments, rules and standards created thereunder, pursuant to which:

- (a) any obligation of a bank or investment firm or affiliate of a bank or investment firm can be reduced, cancelled, modified or converted into shares, other securities or other obligations of such entity or any other person (or suspended for a temporary period); and
- (b) any right in a contract governing an obligation of a bank or investment firm or affiliate of a bank or investment firm may be deemed to have been exercised.

"BRRD" means Directive 2014/59/EU establishing a framework for the recovery and resolution of credit institutions and investment firms.

"BRRD Liability" means a liability in respect of which the relevant Bail-in Powers in the applicable Bail-in Legislation may be exercised.

"Relevant Resolution Authority" means the resolution authority with the ability to exercise any Bail-in Powers in relation to the Service Provider.

# Schedule 3 to the Framework Agreement - Amendments and Supplements to the Service Agreements

N/A

## Schedule 4 to the Framework Agreement - Agency Terms

Where you elect to apply the 'Agency Terms' in the Service Form, the applicable Framework Agreement and Service Agreement shall be supplemented by, and you shall have hereby agreed and accepted the Service Agreement on the basis of, the following terms:

- All references in the Service Agreement (other than these Agency Terms) to "you", or a "party" with reference to you, are deemed to be references to the relevant Underlying Principal (or, where the context permits, you acting for and on behalf of such Underlying Principal).
- In relation to the Service Agreement, if you are acting as agent on behalf of a third party (including, but not limited to, where you act as an investment manager, agent or trustee for a third party and whether disclosed to the Service Provider or not, including, without limitation, a natural person or corporate entity or person acting in the capacity as trustee, sub-fund or protected cell or analogous segregated component, partnership or unincorporated association) (an "Underlying Principal") then, on a continuing basis you represent and warrant and undertake to the Service Provider that:
  - (a) you have full power, authority and capacity from each of your Underlying Principals to enter into and perform your obligations under the Service Agreement, or a transaction entered into pursuant to the Service Agreement, on your Underlying Principal's behalf;
  - (b) you are expressly authorised by your Underlying Principal to instruct the Service Provider in relation to any transaction pursuant to the Service Agreement and each transaction is entered into by you on the Underlying Principal's behalf and consequentially the Underlying Principal shall be liable in respect of all obligations and liabilities to be performed in respect of any such transaction;
  - (c) you have carried out all due diligence required under relevant laws, including without limitation, all applicable prevention and detection of money laundering, client identification, sanctions (for the avoidance of doubt, including any prevention and detection of terrorism legislation), laws and regulations, to satisfy yourself of the good standing of your Underlying Principal and that each Underlying Principal is not involved in any money laundering or criminal activity;
  - (d) you assume full responsibility for, and shall ensure compliance with, without limitation any and all suitability, supervision control, registration, credit review, market abuse laws, rules and regulations and other requirements and restrictions of Applicable Regulations in respect of your Underlying Principals;
  - (e) you will use all reasonable endeavours to ensure that any Underlying Principal on whose behalf you act as agent complies with and fulfils all of its obligations under any transactions entered into pursuant to the Service Agreement; and
  - (f) each Underlying Principal is able to, and you as agent of the Underlying Principal do hereby on its behalf, make the representations in the Service Agreement (other than these Agency Terms) as if all references to "you" in the Service Agreement are references to each Underlying Principal, and you have carried out the requisite due diligence to satisfy yourself of this.

#### 3 You represent and warrant that:

- (a) You have due consent and authority to make the representations, warranties and undertakings set out above or, as the case may be, on behalf of each Underlying Principal;
- You have sufficient experience and expertise either from your own professionals or advisers to reasonably understand the characteristics of derivative contracts relating to currencies and the classification of such contracts as financial instruments for the purposes of Directive 2014/65/EU of the European Parliament and the Council, and the relevant exclusions that apply ("Financial Instrument Exclusions"), and have not relied and will not rely on the Service Provider for such understanding;
- (c) You have processes and procedures in place to consider the applicability of Financial Instrument Exclusions and that such processes and procedures can either identify transactions that do not qualify for a Financial Instrument Exclusion, or ensure that such transactions are not executed with the Service Provider; and
- (d) You shall promptly provide such further information in relation to the processes and procedures referred to in (c) above as may be reasonably required by the Service Provider from time to time to enable the Service Provider to satisfy itself of the existence and extent of such processes and procedures.

#### 4 You agree and acknowledge that:

- (e) each representation and warranty above shall be deemed repeated on each day on which you enter into a transaction with the Service Provider; and
- (f) that the Service Provider shall rely on the accuracy and truthfulness of each representation and warranty above in entering into a transaction with you.
- 5 Clause 7.3(b) of the Framework Agreement shall be disapplied.

The Service Provider may terminate the Service Agreement, or a Service in relation to any such Underlying Principal pursuant to the Service Agreement, without affecting the continuation of the Service Agreement in relation to you and/or any other Underlying Principal.

#### ELECTRONIC ACCESS TERMS AND CONDITIONS (English Law)

These Electronic Access Terms and Conditions (the "Terms and Conditions") set forth the terms and conditions under which The Bank of New York Mellon Corporation and/or its subsidiaries or joint ventures (collectively, "BNY Mellon") will provide the undersigned entities and its (their) affiliates listed on Schedule A ("You" and "Your") with access to and use of BNY Mellon's electronic information delivery site known as "BNY Mellon Connect" and/or other BNY Mellon-designated access portals ("Electronic Access"). Access to and use of Electronic Access by You is contingent upon and is in consideration for Your compliance with the terms and conditions set forth below. Electronic Access includes access to BNY Mellon web sites accessible via BNY Mellon Connect and/or other BNY Mellon-designated access portals ("Sites"), pursuant to which You are able to access products and services provided by BNY Mellon as well as data regarding Your accounts and/or Your clients' accounts. You may amend Schedule A by delivering a revised version to BNY Mellon. If You are BNY Mellon's Client: Any particular product or service accessed by You through Electronic Access may be subject to a separate written agreement between You and BNY Mellon with respect to such products and services (each a "Services Agreement"). In addition terms and conditions and restrictions with respect to any under which The Bank of New York Mellon Corporation and/or its subsidiaries or joint ventures (collectively,

services (each a "Services Agreement"). In addition, terms and conditions and restrictions with respect to any particular product or service accessed through Electronic Access (such as privacy and internet security matters), together with any disclaimers related to the specific products or services, may be set forth on the Sites (hereinafter referred to as "Terms of Use"). If You wish to allow any third party or any employee of a third party (such as an investment manager, consultant or third party service provider) to have access to Your account information through Electronic Access and be included as a "User" under these Terms and Conditions, You may designate that person as Your Authorized User or Authorized Transactional User (as defined below) under these Terms and Conditions and on accompanying account set-up documentation. Otherwise you must send us an authorization letter in the form set out in Schedule B and that third party must enter its own Terms and Conditions for Electronic Access with BNY Mellon. Whether You designate a third party under these Terms and Conditions or whether You permit a third party to access data through Electronic Access under a separate agreement they enter into with BNY Mellon, they will be considered Your third party under these Terms and Conditions.

If You are not BNY Mellon's Client: BNY Mellon's client has signed an authorization letter in the form set out

in Schedule B, providing their consent to You accessing information, data and services regarding their accounts via Electronic Access. Where You outsource functions You perform for BNY Mellon's client, You may designate your outsourced service providers or their employees as additional Users and You represent and warrant to BNY Mellon that you have the consent of BNY Mellon's client to do so.

Applicable to all who receive these Terms: By Your signature below, You agree to the Terms and Conditions and the Terms of Use. By any of Your Users accessing the Sites, and the products and services available through Electronic Access, You agree to any Terms of Use and acknowledge and accept any disclaimers and disclosures included on the Sites and the restrictions concerning the use of proprietary data provided by Information Providers (as defined below) that are posted on the Data Terms Web Site (as defined below). For the avoidance of doubt, the execution of these Terms and Conditions will not after or amend or otherwise affect any Services Agreement whether such Services Agreement is executed prior to or after the execution of these Terms and Conditions

#### Access Administration:

- To facilitate access to Electronic Access, You will furnish BNY Mellon with a written list of the names, and the extent of authority or level of access, of persons You are authorizing to access the Sites, products and services and to use the Electronic Access ("Authorized Users") on a read-only basis. In addition, You may also designate Authorized Users who will have authority to enter transactions and provide instructions to BNY Mellon that cause a change in or have an impact on assets held by BNY Mellon for Your accounts and/or Your clients' accounts ("Authorized Transactional Users"). Where appropriate, Authorized Users and Authorized Transactional Users are collectively referred to You also give authority to Users who will be designated as administrators to herein as "Users." designate future Users
- designate future Users.

  Upon BNY Mellon's approval of Users (which approval will not be unreasonably withheld), BNY Mellon will send You a user-id, temporary password and, where applicable, a security identification device for each User. You will be responsible for providing to Users the user-ids, temporary passwords and, where applicable, secure identification devices. You will ensure that any User receiving a secure identification device returns such device immediately following the termination of the User's authorization to access the products and services for which the secure identification device was provided to such User. You are solely responsible for Users' access to Electronic Access, and You and Users are solely responsible for the confidentiality of the user-ids and passwords and secure identification devices that are provided to them and will remain responsible for each secure identification device until it is returned to BNY Mellon. You, on behalf of You and Your affiliates, acknowledge and agree that, BNY Mellon will have no duty or obligation to verify or confirm the actual identity of the person who accessed Electronic Access using a validly issued user-id and password (and, where applicable, security identification device) or that the person who accessed Electronic Access using such validly issued user-id and password (and, where applicable, security identification device) is, in fact, a User (whether an Authorized User or an Authorized Transactional User).
- Transactional User).
  You shall not, and shall not permit any User or third party to, breach or attempt to breach any security measures used in connection with Electronic Access or Proprietary Software. Any attempt to circumvent or penetrate any application, network or other security measures used by BNY Mellon or its suppliers in connection with Electronic Access is strictly prohibited.
  You are also solely responsible for ensuring that all Users comply with these Terms and Conditions and are Terms of Ites included on the Step the Society Society of Society Soc
- and any Terms of Use included on the Sites, the Services Agreement for each product or services accessed through the Sites and their associated services and all applicable terms and conditions, restrictions on the use of such products and services and data obtained through the use of Electronic Access. BNY Mellon reserves the right to prohibit access or revoke the access of any User to Electronic Access whom BNY Mellon determines has violated or breached these terms and conditions or any Terms of Use on a Site accessed by the User, including the Data Terms Web Site (as defined below), or whose conduct BNY Mellon reasonably determines may constitute a criminal offense, violate any applicable local, state, national, or international law or constitute a security risk for BNY Mellon, its third party service providers, its clients or any Users of Electronic Access. BNY Mellon may also terminate access to all Users following termination of all Services Agreements.

#### 2.

- Agreements.

  Proprietary Software:

  a. Depending upon the products and services You elect to access through Electronic Access, You may be provided software owned by BNY Mellon or licensed to BNY Mellon by a third party ("Proprietary Software"). You are granted a limited, non-exclusive, non-transferable license to install the Proprietary Software on Your authorized computer system (including mobile devices registered with BNY Mellon) and to use the Proprietary Software solely for Your own internal purposes in connection with Electronic Access and solely for the purposes for which it is provided by Your Meyer Market and Proprietary Software solely for Your Openitor Software for the purposes for which it is provided. to You. You and Your Users may make copies of the Proprietary Software for backup purposes only, provided all copyright and other proprietary information included in the original copy of the Proprietary Software are reproduced in or on such backup copies. Any attempt to circumvent or
- Popularly solving a required in the following and state states of the security of Electronic Access is strictly prohibited.

  You agree that if You require the Proprietary Software to be installed on a computer system not under Your control: (i) You will notify BNY Mellon in writing prior to such installation; (ii) You under Your control: (i) You will notify BNY Mellon in writing prior to such installation; (ii) You will be responsible and liable for the installation and protection of the Proprietary Software on such system in the same manner as though You had installed the Proprietary Software on Your internal computer system; and (iii) BNY Mellon may require the individual or entity that controls such computer system to sign a separate agreement with BNY Mellon. You agree that You will not (i) except as provided in these Terms and Conditions, make additional copies of Proprietary Software; (ii) disclose Proprietary Software to, or allow Proprietary Software to be used by or for the benefit of, any third party; (iii) alter, decompile, disassemble, reverse engineer, modify, or attempt to determine the source code for, any Proprietary Software and/or merge Proprietary Software with another software program. Version: September 2015.

  \*\*Data\*\*

#### Use of Data:

Electronic Access may include information and data that is proprietary to the providers of such information or data ("Information Providers") or may be used to access Sites that include such information or data from Information Providers. This information and data may be subject to restrictions and requirements which are imposed on BNY Mellon by the Information Providers and

- which are posted on http://www.bnymellon.com/products/assetservicing/vendoragreement.pdf or an successor web site of which You are provided notice from time to time (the "Data Terrus Web Site"). You will be solely responsible for ensuring that Users comply with the restrictions and requirements
- concerning the use of proprietary data that are posted on the Data Terms Web Site.

  You consent to BNY Mellon, its affiliates, and its third party service providers disclosing to each other and using data received from You and Users and, where applicable, Your third parties in connection with these Terms and Conditions (including, without limitation, client data and personal data of Users) (I) to the extent necessary for the provision of Electronic Access; (2) in order for BNY Mellon and its affiliates to meet any of their obligations under these Terms and Conditions to
- provide Electronic Access; or (3) to the extent necessary for Users to access Electronic Access, and You have the necessary consents of Your Users and Your third parties to give effect to this clause. You agree that BNY Mellon may use other BNYM entities and third parties in connection with its performance of the services and any other obligations under these Terms and Conditions, the Terms performance of the services and any other obligations under these Terms and Conditions, the Terms of Use, each Services Agreement and in certain other activities, including, without limitation, audit, accounting, tax, administration, risk management, credit, legal, compliance, operations, sales and marketing, relationship management, information technology, records and data storage, performance measurement, data aggregation and compilation and analysis of Your Information (collectively, the "Activities"). Notwithstanding anything to the contrary in these Terms and Conditions, the Terms of Use or each Services Agreement, each BNYM entity may, in connection with the Activities or for any other purpose permitted under these Terms and Conditions, the Terms of of Use or each Services Agreement, collect, use, store and disclose, within and outside of the European Economic Area (including but not limited to the United States, Australia, Canada, Cayman Islands, Hong Kong, India, Japan, Republic of Korea and Singapore) Your Information to: Cayman islands, Hong Rong, India, Japan, Republic of Rofera and Singapore; Your information to:

  (a) other BNYM entities; and (b) bird party service providers who are required to maintain the
  confidentiality of Your Information. In addition, BNYM may aggregate Your Information (other
  than Personal Data) with other data collected and/or calculated by BNYM, and BNYM will own all
  such aggregated data, provided that BNYM shall not distribute the aggregated data in a format that
  identifies You or any particular individual after such aggregation. You represent You have lawful
  grounds and BNYM relies on Your representation for BNYM's collection, use, storage and
  disclosures of Your Information. disclosure of Your Information, including Personal Data, as set out in these Terms and Conditions You consent to the disclosure of Your Information to governmental, tax, regulatory, law enforcement and other authorities in relevant jurisdictions where BNYM operates and otherwise as required by law, rule or guideline (including tax reporting regulations) or requested by such authorities.
- In relation to the collection, use, storage and disclosure of Personal Data by BNYM, to the extent In relation to the collection, use, storage and disclosure of Personal Data by BNYM, to the extent that each BNYM entity is required to obtain consent under the applicable Personal Data laws in any jurisdiction, You confirm that by providing such Personal Data to BNYM, You have lawful grounds to allow each BNYM entity to collect, use, store and disclose Personal Data in accordance with these Terms and Conditions and the notice contained at https://www.bnymellon.com/emea/en/privacy.jsp ("Personal Data Notice"). For the avoidance of doubt, where consent is not required in the particular jurisdiction, each BNYM entity is providing notice of its collection, use, storage and disclosure of Personal Data in accordance with these Terms and Conditions and the Personal Data Notice, receipt of which is acknowledged by You. You agree that BNYM may make amendments and additions to this Personal Data Notice by posting a revised version of this Personal Data Notice at the abovementioned website link (or such other link as BNYM may advise You from time to time). You agree that Your maintenance and/or continued use of any service provided by any BNYM entity and continued provision of Personal Data to a BNYM after any such revised version is posted constitutes Your deemed confirmation that You use to have lawful grounds to permit each BNYM entity to collect, use, store and disclose Personal Data in accordance with the revised Personal Data Notice.
- Personal Data in accordance with the revised Personal Data Notice.

  In paragraphs c, and d, above, "BNYM" and "BNYM" entity" means The Bank of New York Mellon Corporation and/or each of its subsidiaries, affiliates (including each of their respective branches and representative offices, individually and/or collectively) and joint ventures, acting either as the contracting entity under these Terms and Conditions or as service provider or intermediary to BNY Mellon, or otherwise in a relationship with You; and "Your Information" means data regarding You and Your affiliates and subsidiaries and the account(s) established and/or maintained pursuant to each Services Agreements, including Personal Data; "Personal Data" means personal data of employees and representatives of Your and Your affiliates and subsidiaries, and, where You are the trustee of a trust, also includes personal data of individual beneficiaries and employees and representatives of beneficiaries of that trust. This paragraph and paragraphs c. and d. above will survive termination of these Terms and Conditions.

#### Ownership and Rights:

- Electronic Access, including any database, any software (including for the avoidance of doubt Electronic Access, including any database, any software (including for the avoidance of doutr, Proprietary Software) and any proprietary data, processes, scripts, information, training materials, manuals or documentation made available as part of the Electronic Access (collectively, the "Information"), are the exclusive and confidential property of BNY Mellon and/or BNY Mellon's suppliers. You may not use or disclose the Information except as expressly authorized by these Terms and Conditions. You will, and will cause Users and Your third parties and their users, to keep the Information confidential by using the same care and discretion that You use with respect. to Your own confidential information, but in no event less than reasonable care.
- to Your own confidential information, but in no event less than reasonable care. The provisions of this paragraph will not affect the copyright status of any of the Information which may be copyrighted and will apply to all Information whether or not copyrighted. Nothing in these Terms and Conditions will be construed as giving You, Users, or third party users any license or right to use the trade marks, logos and/or service marks of BNY Mellon, its affiliates, its licensors, its Information Providers or its third party service providers. Any Intellectual Property Rights and any other rights or title not expressly granted to You or Users
- under these Terms and Conditions are reserved to BNY Mellon, its affiliates, its licensors, its Information Providers and its third party service providers. "Intellectual Property Rights" includes all copyright, patents, trademarks and service marks, rights in designs, moral rights, rights in computer software, rights in databases and other protectable lists of information, rights in confidential information, trade secrets, inventions and know-how, trade and business names, domain names (including all extensions, revivals and renewals, where relevant) in each case whether registered or unregistered and applications for any of them and the goodwill attaching to any of them and any rights or forms of protection of a similar nature and having equivalent or similar effect to any of them which may subsist anywhere in the world.
- Fees: In consideration for Electronic Access as set out in the Terms and Conditions, You shall pay the fee of £1 to BNY Mellon (the receipt of which is hereby acknowledged by BNY Mellon). All fees are exclusive of applicable value added tax (and other similar taxes) and all such taxes will be payable by You. Confidentiality:
- Electronic Access (including, without limitation, the design, programming techniques, algorithms and codes contained within the Electronic Access), Proprietary Data (as defined in the Data Terms Website) and any other information provided by BNY Mellon or any of its affiliates to You are confidential property of BNY Mellon, its affiliates, its licensors, or third party service providers in relations to the Electronic Access Service under these Terms and Conditions.

  You shall not, and shall procure that each User must not, without the prior written consent of BNY
- You shall not, and shall procure that each User must not, without the prior written consent of BNY Mellon, disclose to any third party or make unauthorized use of the proprietary Information, the Sites and the content of these Terms and Conditions, and will take reasonable care to protect the confidential property in relation to these Terms and Conditions from examination by anyone except for its agents or employees who have a need to know. You shall be responsible for the consequences of any misuse of, or unauthorized use of or access to, the Proprietary Software and for the disclosure of any confidential property or information by You or a User, provided that this clause 6 shall not restrict any disclosure by You pursuant to any applicable law, or by order of any court or government agency. You will promptly notify BNY Mellon in writing if such disclosure occurs.

  Notwithstanding any other provision in these Terms and Conditions, BNY Mellon may at its discretion, retain the records of Your and each User's Commands for any applicable legal or regulatory requirement and, for any other reason, including, without limitation, monitoring the
- regulatory requirement and, for any other reason, including, without limitation, monitoring the quality of service that You receive, Your compliance with these Terms and Conditions and the security of the Information.

- Reliance:

  a. BNY Mellon will be entitled to rely on, and will be fully protected in acting upon, any actions or a second elastification device issued to a User until such instructions associated with a user-id or a secure identification device issued to a User until such time BNY Mellon receives actual notice in writing from You of the change in status of the User and receipt of the secure identification device issued to such User. You acknowledge that all commands, directions and instructions, including commands, directions and instructions for transactions issued by a User, or Your third party are issued at Your sole risk. You agree to accept full and sole responsibility for all such commands, directions and instructions and that BNY Mellon, will responsionly not an such commands, directions and institutions and that not without when of liability for, and you hereby release BNY Mellon from, any losses, liabilities, damages, costs, expenses, claims, causes of action or judgments (including attorneys fees and expenses) (collectively "Lasses") incurred or sustained by You or any other party in connection with or as a result of BNY Mellon's reliance upon or compliance with such commands, directions and
- All commands, directions and instructions ("Commands") involving a transaction entered by Authorized Transactional User will be treated as an authorized instruction under the applicable Activative Transactional Uses with the trade as an artificial instance the applicable Services Agreement(s) covering accounts, products and services and products provided by BNY Mellon with respect to which Electronic Access is being used whether such Services Agreement is executed prior to or after the execution of these Terms and Conditions.
- BNY Mellon shall use reasonable endeavours to comply with any Commands received from User or any requests from You regarding the revocation and cancellation of such any Commands, provided that such a request: (a) is correctly entered by the User; (b) is received by BNY Mellon in sufficient time for BNY Mellon to act upon such a request; and (c) complies with BNY Mellon's
- BNY Mellon each agree that the time at which Commands are deemed to be received by BNY Mellon shall be the time of the sending of the confirmation of receipt by BNY Mellon to the originator of the relevant transaction. BNY Mellon shall have no obligation to act on any Commands unless BNY Mellon have sent (or otherwise made available) the User (from whom the Command originated from) a transaction reference number. BNY Mellon may in their absolute discretion decline to act upon any Commands which are insufficient or incomplete or are not received by BNY Mellon in sufficient time for BNY Mellon to act upon or in accordance with such Commands, or any other circumstances prescribed by the relevant Services Agreement.
- Commands, or any other executances preserviced by the reveals services agreement.

  To the fullest extent permitted by law, there is no warranty of merchantability, no warranty of fitness for a particular purpose, no warranty of quality and no warranty of title or noninfringement. There is no other warranty of any kind, express or implied, regarding electronic access, the sites, any proprietary software, information, materials or client data.

  Notwithstanding the prior paragraph, BNY Mellon or an affiliate designated by it, will defend You and pay any amounts agreed to by BNY Mellon in a settlement and damages finally warded by a court of compared by the final party against the properties in the party against the party against the properties in the party against the party agai
- and pay any amounts agreed to by BNY Mellon in a settlement and damages finally awarded by a court of competent jurisdiction, in an action or proceeding commenced by a third party's patent, copyright, or trade secret, provided that You (i) notify BNY Mellon promptly of any such action or claim (except that the failure to so notify BNY Mellon will not limit BNY Mellon, or its hereunder except to the extent that such failure prejudices BNY Mellon); (ii) grant BNY Mellon, or its designated affiliate, full and exclusive authority to defend, compromise or settle such claim or action; and (iii) provide BNY Mellon, or its designated affiliate, all assistance reasonably necessary to so defend, compromise or settle. The foregoing obligations will not apply, however, to any claim or action arising from (i) use of the Proprietary Software Information or Electronic Access in a manner not authorized under these Terms and Conditions, the Terms of Use, or the Data Terms Web Site; or (ii) use of the Proprietary Software or Electronic Access in combination with other software or ervices not supplied by BNY Mellon.

#### Disclaimers:

- Although BNY Mellon uses reasonable efforts to provide accurate and up-to-date information through Electronic Access, BNY Mellon, its Content Providers and Information Providers make no warranties or representations under these Terms and Conditions as to accuracy, reliability or comprehensiveness of the content, information or data accessed through Electronic Access. Without limiting the foregoing, some of the content on Electronic Access may be provided by third party sources ("Content Providers") and by Information Providers. For that content BNY Mellon is a distributor and not a publisher of such content and has no control over it. Information provided by Information Providers has not been independently verified by BNY Mellon and BNY Mellon makes no representation as to the accuracy or completeness of the content or information provided. Any opinions, advice, statements, services, offers or other information given or provided by Content Providers and Information Providers (including merchants and licensors) are those of the respective authors of such content and not that of BNY Mellon. BNY Mellon will not be liable to You, Your clients, Users, or Your third party for such content or information in any way nor for any action taken in reliance on such information nor for direct or indirect damages resulting from any actor tach in remarks on a set information for our decormance damages testing from the use of such information. For purposes of these Terms and Conditions, all information and data, including all proprietary information and materials and all client data, provided to You through Electronic Access are provided on an "AS-IS", "AS AVAILABLE" basis.

  BNY Mellon makes no guarantee and does not warrant that Electronic Access or the information
- and data provided through the Electronic Access are or will be virus-free or will be free of viruses, worms, Trojan horses or other code with contaminating or destructive properties. BNY Mellon will employ commercially reasonable anti-virus software to its systems to protect its systems against
- ome Sites accessed through the use of Electronic Access may include links to third party websites. BNY Mellon will not be liable to any person for the content found on such third party websites. BNY Mellon will not be responsible for third party websites that collect information from parties who visit their web sites through links on the Sites. BNY Mellon will not be liable or responsible for any loss suffered by any person as a result of their use of any third party web sites that are 14. linked to the BNY Mellon Sites.
- BNY Mellon retains complete discretion and authority to add, delete or revise in whole or in part Electronic Access, including its Sites, and to modify from time to time any Proprietary Software provided in conjunction with the use of Electronic Access and/or any of the Sites. To the extent provided in conjunction with the use of Electronic Access and/or any of the Sites. To the extent reasonably possible, BNY Mellon will provide notice of such modifications. BNY Mellon may terminate, immediately and without advance notice, and without right of cure, any portion or component of Electronic Access or the Sites.

  To the fullest extent permitted by the state of the Sites.
- To the fullest extent permitted by law, there is no warranty of merchantability, no warranty of fitness for a particular purpose, no warranty of quality and no warranty of title or noninfringement. There is no other warranty of any kind, express or implied, regarding electronic access, the sites,
- notes is no other warrancy or any kind, express or impliced, registering states any proprietary software, information, materials or client data.

  Notwehistanding the prior paragraph, BNY Mellon or an affiliate designated by it, will defend You and pay any amounts agreed to by BNY Mellon in a settlement and damages finally awarded by a court of competent jurisdiction, in an action or proceeding commenced by a third party against court of competent jurisdiction, in an action or proceeding commenced by a third party aganty's You based on a claim that Electronic Access or the Proprietary Software infringe such third party's patent, copyright, or trade secret, provided that You (i) notify BNY Mellon promptly of any such action or claim (except that the failure to so notify BNY Mellon will not limit BNY Mellon's obligations hereunder except to the extent that such failure prejudices BNY Mellon; (ii) grant BNY Mellon, or its designated affiliate, full and exclusive authority to defend, compromise or settle such claim or action; designated arrinate, run and exclusive authority to dereitd, compromise or settle such craim or action; and (iii) provide BNY Mellon, or its designated affiliate, all assistance reasonably necessary to so defend, compromise or settle. The foregoing obligations will not apply, however, to any claim or action arising from (i) use of the Proprietary Software Information or Electronic Access in a manner not authorized under these Terms and Conditions, the Terms of Use, or the Data Terms Web Site; or (ii) use of the Proprietary Software or Electronic Access in combination with other software or services not supplied by BNY Mellon.

Limitation of Liability:
a. In no event will BNY Mellon, its licensors, content providers, Information Providers or third party service providers be liable to you or anyone else under these terms and conditions for any losers, liabilities damages, costs or expenses including but not limited to, any direct damages, consequential damages, reliance damages, exemplary damages, incidental damages, special damages, punitive damages, indirect damages or damages for loss of profits, good will, business interruption, use, data, equipment or other intangible losses (even if we have been advised of the possibility of such damages) that result from (1) these Terms and Conditions; or (2) the use of or

- inability to use electronic access. Proprietary Software or client data. BNY Mellon will not be liable for any loss, damage or economic injury resulting from voluntary shutdown of the server, electronic access or any of the sites to address technical problems, computer viruses, denial-of-service messages or other similar problems.
- in Source and similar properties. BNY Mellon, its licensors, the Information Providers and the third party service providers arising out of any kind of claim, demand, proceedings or action ("Claim"), whether in contract, negligence or otherwise, in any way connected with Electronic Access, Information, client data and these Terms and Conditions shall not exceed the greater of the amount of the fees paid by You for Electronic Access over the 12 months immediately preceding the incident or £18,000. A number of Claims arising out of the same event or a series of connected events shall be deemed to be a single Claim for these purposes. Nothing in these Terms and Conditions shall exclude or limit any liability which cannot be excluded or limited by any
- The limitation of liability set forth in this Limitation of Liability section and in other provisions in these Terms and Conditions is in addition to any limitation of liability provisions contained in any Services Agreements and will not supersede or be superseded by limitation of liability provisions contained in such Services Agreements, whether executed prior to or after the execution of these Terms and Conditions, except to the extent specifically set forth in such other Services Agreements containing a reference to these Terms and Conditions.

#### Indemnification:

- You agree to indemnify, protect and hold BNY Mellon, its affiliates, its licensors, Content Providers and Information Providers harmless from and against all liability, claims damages, costs and expenses, including reasonable attorneys fees and expenses, resulting from a claim by a third party that arises out of (i) any breach by You or Users, or any third parties of these Terms and Conditions, that arises out of the part Terms of Use or the Data Terms Web Site; (ii) BNY Mellon's use of client data in accordance with these Terms and Conditions; and (iii) any person obtaining access to Electronic Access through You, Users, Your third parties or through use of any password, user-id or secure identification device issued to one of Users, whether or not You, a User, or Your third party authoritied such access. For the avoidance of doubt, and by way of illustration and not by way of limitation, the forgoing indemnity is applicable to disputes between the parties, including the enforcement of these Terms and Conditions. The rights and remedies conferred hereunder will be cumulative and the exercise or waiver of any such right or remedy will not preclude or inhibit the exercise of additional rights or remedies or the subsequent exercise of such right or remedy.
- The indemnity provided in herein is in addition to any indemnity and other remedies contained in any Services Agreements and will not supersede or be superseded by such Services Agreements, whether executed prior to or after the execution of these Terms and Conditions, except to the extent specifically set forth in such other Services Agreements and expressly stating an intent to modify this Terms and Conditions. Nothing contained herein will, or be deemed to, alter or modify the rights and remedies of BNY Mellon as set forth in the Services Agreements.

  Governing law and jurisdiction: Unless otherwise agreed and specified herein, these Terms and Conditions and any issues or disputes (whether such disputes are contractual or non-contractual (if any) in
- nature) arising out of or in connection with them are governed by and construed in accordance with English law and You and BNY Mellon each expressly and irrevocably agree to submit, to the exclusive jurisdiction of the courts of England.

#### 12.

- Term and Termination:

  a. Either BNY Mellon or You may terminate these Terms and Conditions and the Electronic Access upon thirty (30) days written notice to the other party. Either BNY Mellon or You may terminate this Agreement to coincide with a termination of a Services Agreements between You and BNY
- In the event of any breach of the provisions of these Terms and Conditions or a breach by any Authorized User of the Terms of Use or the restrictions and requirements concerning the use of Information Providers' proprietary data that are posted on the Data Terms Web Site, the non-breaching party may terminate these Terms and Conditions and the Electronic Access immediately upon written notice to the breaching party if any breach remains uncured after ten (10) days written
- notice of the breach is sent to the breaching party.

  BNY Mellon may immediately terminate access through an Authorized User's user-id and password and may, at its discretion, also terminate access by an Authorized User, without right of cure, in the event of an unauthorized use of an Authorized User's user-id or password, or where
- BNY Mellon believes there is a security risk created by such access.

  BNY Mellon may terminate, without advance notice, Your access or the access of Users to any portion or component of Electronic Access or the Sites in the event a third party service provider, Content Provider or Information Provider prohibits BNY Mellon from permitting You or Users to have access to their information or services.
- Promptly upon receiving or giving notice of termination, You will notify all Users of the effective
- date of the termination.

  Upon termination of Your access to Electronic Access, You shall return all manuals, documentation, workflow descriptions and the like that are in Your possession or under Your
- documentation, workflow descriptions and the like that are in Your possession or under Your control and all security identification devices. The foregoing obligations shall not be constituted to require You to return, delete or destroy client or personal data.

  g. The Proprietary Software, Use of Data, Ownership and Rights, Reliance, Disclaimers, Limitation of Liability Indemnification, Force Majeure, General and Confidentiality provisions of the Terms and Conditions (and other provision of these Terms and Conditions containing disclaimers, limitation of liability and indemnification) shall survive the termination of these Terms and Conditions, none of BNY Mellon, its still nature service provides required provides and Conditions.
- third party service providers, licensors, or Information Providers shall be liable if they are prevented from, or delayed in performing their obligations, or for any loss resulting from a cause that is beyond the control of that entity

- of that entry.

  General

  BNY Mellon has adopted an incentive compensation program designed (i) to facilitate customers gaining access to and being provided with explanations about the full range of products and services offered by BNY Mellon and (ii) to expand and develop customer relationships. This program may lead to the payment of referral fees to employees of BNY Mellon who may have been involved in a referral that resulted in Your obtaining of products or services covered by the Terms and Conditions or which may be ancillary or supplemental to such products or services. Any such referral fees are funded solely out of fees and commissions paid by You under these Terms and Conditions or with respect to such ancillary or supplemental products or services. Further details of the payment of referral fees will be provided to the extent required by applicable laws or regulations and other legal requirements and limitations. Should You require such further details, You should contact its designated BNY Mellon relationship manager.

  These Terms and Conditions contain the entire agreement between You and BNY Mellon relating
- These Terms and Conditions contain the entire agreement between You and BNY Mellon relating to the Electronic Access and supersedes all prior written or oral communications between the parties on the subject. In the event any one or more of the provisions of these Terms and Conditions shall for any reason be held to be invalid, illegal or unenforceable, the remaining provisions shall remain valid and enforceable. Furthermore, the unenforceable provision(s) shall be modified to carry out to the full extent possible the intent of the provision deemed unenforceable.

  Unless otherwise specified in these Terms and Conditions, these Terms and Conditions may be amended through a written notice by BNY Mellon to You, together with written acceptance by You of such amendment to his Acceptance.
- of such amendment to this Agreement.
- You may not assign these Terms and Conditions without the prior written consent of BNY Mellon. BNY Mellon may assign these Terms and Conditions to any affiliate or successor in interest without Your prior consent. These Terms and Conditions shall be binding upon and inures to the
- without rout prior consent. These retrins and Conditions stand to bringing upon and mutes to the benefit of You and BNY Mellon and each of your respective successors and assigns. You and BNY Mellon each acknowledge the intention that these Terms and Conditions do not confer or purport to confer on any third party any benefit or any right to enforce any provisions under these Terms and Conditions, except that: (a) BNY Mellon's affiliates may benefit from and enforce the provisions under these Terms and Conditions; and (b) each Information Provider may benefit from and enforce any right or benefit given to a Information Provider under these Terms and Conditions.
- BNY Mellon and You may each exercise, without the consent of any third party, any rights you each may have to amend or rescind these Terms and Conditions.

g. A notice to either party shall be given in written English and may be delivered personally or sent by first class post, pre-paid recorded delivery or by facsimile to the other party. Any proceedings arising out of these Terms and Conditions will be sufficiently served if served on the relevant party at its given address. Each notice will be effective, in the case of a notice delivered by facsimile, when the sender receives a fax receipt confirming transmission of the fax and, in any other case, will be effective upon actual receipt.

You represent and warrant to BPN Mellott that these Terms and Conditions and the indemnity contained herein bave been daty authorized, executed and delivered on Your behalf, that You have full mathority to execute these Terms and Conditions, both for the undersigned entities and for any affiliate with Electronic Access, and that the individual executing these Terms and Conditions has the requisite authority to bind the undersigned entities and each such affiliate to these Terms and Conditions, and that these Terms and Conditions constitute Your binding obligation enforceable in accordance with its terms.

Chiseri, Client's legal entity name)
while A.
By:
Name: Ian Shea
Title:_Director
Date:
By: _energonaeveroveroveroveroveronaevero
Name:
Tale:
Dase:
signed Higgins 1 +01'00'

#### SCHEDULE A List of Client's affiliates

#### SCHEDULE B Client authorization letter [ON CLIENT LETTERHEAD]

#### [insert date]

THE BANK OF NEW YORK MELLON, LONDON BRANCH One Canada Square London E14 5AL, U.K.

Signature of Client's authorised signatory

Client's authorised signatory name

Title

To whom it m	nay concern					
Authorisatio	n of a third pa	irty to receive	information and provide instructions by mean o	f Electronic Access under the BNY Me	llon Electronic Access Terms and Conditio	ns ("Terms and Conditions")
			more Services Agreements that we (the "Client") latso have signed the EAA once] ("Third Party") to		ent authorises	[insert
I. 2.	Receive clien	t data under the or after the d	Access by BNY Mellon in accordance with the Te ne Terms and Conditions in relation to any of the Cl ate of this authorisation ("Accounts") unless this a	lient's accounts existing at any time in co	nnection with the Services Agreements, irres	pective of whether such accounts are
3. 4. 5.	data to BNY (applicable or	Mellon that ca	nd as an Authorised Person as defined in the Servi use a change in or have an impact upon assets held I Third Party outsources functions it performs for the RNY Mellon:	by BNY Mellon for the Client's account.	•	
5.	(a) (b)	that the above receive infor	we authorisations by the Client include but are not li mation, valuations and data ("Commands") to BN I give effect to the above authorisations so that eac	Y Mellon, in connection with any of the A	Accounts; and	
		Electronic A (i) (ii)	ccess shall:  be deemed to be an "authorised instruction" given constitute a representation and warranty by the C that BNY Mellon may rely on the representations stated below.	Hient that the Third Party continues to be	duly authorised by the Client to so act on the	
6.	on terms that		nat the Third Party will not be provided Electronic at the BNY Mellon and (ii) the Client has instructed E			
7.		this Client aut supersede an remain effec on which BN		authorisation is to be revoked, where such	notification of revocation will be effective	
8.	The Client ac		n executing this authorization letter that it is accepting	ng responsibility and liability for Losses (	as defined in the Services Agreements result	ing from the acts or omissions of the
9.		erms used in th	is letter that are not defined have the same meaning	as given to them in the Terms and Condit	ions.	
Yours faithfu	lly,					
			JINSERT FULL CLIENT LEGAL	ENTITY NAME] (the "Client")		
			IINSERT FULL CLIENT LEGAL	ENTITY NAME] (the "Client")		

# The Bank of New York Mellon Operational Terms



#### **Table of Contents**

Conte	ents Pa	age
1	Introduction	0
2	Definitions	0
3	Forms of Notices	1
4	Reports and Statements	1
5	Notices and Other Communications	3
6	Governing Law and Jurisdiction	3
	Forms for BNYM LB	
	dule 1 to the Operational Terms - Collateral Management Master Agreement (Collate	
Provid	der) Service Module – Forms of Notice	4
	dule 2 to the Operational Terms - Collateral Management Master Agreement (Collate ver) Service Module – Forms of Notice	
Sched	dule 3 to the Operational Terms - Dealer Custody (English / Belgian law) Service	
Modul	le and Receiver-only Custody (English / Belgian law) Service Module	. 04
Sched	dule 4 to the Operational Terms - Security (English law) Service Module	. 10
	dule 5 to the Operational Terms - Security [English law] / [Belgium Law] Service	
Modul	le	. 16
	Forms for BNYM SA/NV	
Sched	dule 1A to the Operational Terms - Collateral Management Master Agreement	
(Colla	teral Provider) Service Module – Forms of Notice	. 2.
Sched	lule 2A to the Operational Terms - Collateral Management Master Agreement	
(Colla	teral Receiver) Service Module – Forms of Notice	. 3.
Sched	dule 3A to the Operational Terms - Dealer Custody (English / Belgian law) Service	
Modul	le and Receiver-only Custody (English / Belgian law) Service Module	. 4.
Sched	lule 4A to the Operational Terms - Security (English law) Service Module	. 44
	lule 5A to the Operational Terms - Security [Belgian law] / [English law] Service	. 51

In the above Table of Contents, Schedules 1, 2, 3, 4 & 5 are relevant to entities contracting with BNYM LB; Schedules 1A, 2A, 3A, 4A & 5A are relevant to entities contracting with BNYM SA/NV.

#### 1 Introduction

RfcDp]kcumpi epockclramlqrogsrcqrfcyqcptgac]epockclr—s[qbcdgcb\_cjmu'\_crucclwns]|b@LWK]|bglampnmp]rcq\_wpcocbpclacrfc]nnjga]\_jc cptgacKmbsjcqs[qbcdgcb\_cjmu'dmprfcqcptgacwnspcosgocmeerfcpugfrfcPcesj]mpw]|bMncp]rgml]jRcpkq,@wcptgacKmbsjcuckc]|rfcpcjct]|r]epockclr]qn]prmdrfcDp]kcumpi epockclrs@jajsbgle\_srlmrjgkgocbrmrfcAmKKNpmtgbcpmpPcacglcp) caspogwepockclrmprfcBc]jcpAsqmbwepockclr',

Ric Mncp]romi] jRcpk qamir] oliq ric chopk qmdl mropacqchop] jj cpt gacq &liq b cobjecb\_cjmu'rf] r k]w\_c snb]rcb chonk rojk cmmrojk c\_w @LWK u opf msr pc cvca sromi] lb npmt obcq bcqaponromi molqr] lb]pb mncp]romi] j npmacqqcq srojongcb rm bcjot cp ric amjj]rcp]j qcpt gacq rm wns,

Ricco ] pc object af cbsjcq rm ric Mncp]romi]j Rock q benel boje mi u ferfep wms ] pc amirp]andje u ogf Jmibmi @p]laf mpu ogf -LT Njc]qc pcobprm rick]oj oj bev rmk]ie ric pcjct]lr cjcaromiq, af cbsjcq/)0)1)2 3 ] pc pcjct]lr rmclrogocq amirp]andje u ogf @LWK J@9 af cbsjcq/)0)1)2 3 ] pc pcjct]lr rmclrogocq amirp]andje u ogf @LWK -LT,

#### 2 Definitions

Rfc dnjjmu oj e rcpk q qf] jj f] tc rfc dnjjmu oj e k c] l oj e8

yCustody Service Module-8 cpt gac Kmbsjc uf cpc\_wwms] nnmograf c cpt gac Npmt opcp] q asqrmbog | modrf c caspogocq bennnqojeb dmpq] object ugf rf c cpt gac Npmt opcpmpu gaf ] | w s\_asqrmbog | ] nnmograb \_wrf c cpt gac Npmt opcp] | b rmf mjb A] qaf og ] aampb] | ac u gaf rf c repk q modrf c cpt gac epeck clr,

yService Agreement-8Rfc] epock clrrf]ra]k.c.g/rmcobbar\_cruccl wms]lbrfc cpt.gac Npmtobocpol poquo armolo] af motific qopt.gack mbsjcq cjcarcbm]nnjw/rmwmsol poquo armolific ]nnjga]\_jc cpt.gacNpmtobocp]q qormsrol rfcqopt.gacdmpk,

yServices-8Rfc cpt gacq rf]r]nnjwrmwns,

"Service Form-8Dmpk ufcpc rfc pcjct] Ir cpt gac K mb sjc gq b cobj cb,

"Service Module-8C] af modrfcolopological j cptoacq rf]r]nnjwmwns,

"Service Provider–8 YRfc@[li MdLcu Wmpi Kcjjml -LTZmp YRfc@[li mdLcu Wmpi) Jmlbml@m]lafZ,

yStatement of Account-8 r]rck clrq md] aamslrrf]rrfc cpt gac Npmt opcpqf]jjnpmt opc wms ml]os]propjw\_]qoqqfmu oje]jjpp|lq]aromlq ojrfc aamslr,

yStatement of Assets-8 r] rck clrq md] qqcrq rf] r rf c cpt gac Npmt obcpqf] jj npmt obc wms ml os] prepiw ] qqq ob rf c aamsl r,

yThird Party Security Interest-8 Gebnyk ]rominf]rojajsbcq & srogal mrjok ojcb mibcr]ojq md ]lw qcaspogwojropcqredjirob \_w wmsoj qitmsp mdifopb n]propcq & mrojajsboje nfc o cptopac Npmtopop,

Iw rcpk q sqcb of Aj] sqc 2 & Reports and Statements' mp Aj] sqc 3 & Notices and Other Communications' ufgaf] pc Imrbcotof cb of rfcqc Mncpl rognl] j Rcpk q qf] jj f] tc rfckc] I of e eqtcl mmfck of rfc] nnjoa] jc cpt gac Kmbsjc,

FMA1 September 2021

<sup>&</sup>lt;sup>1</sup> Appropriate selection to be made depending on whether the Client is contracting with London Branch or with SA/NV

## 3 Forms of Notices

Ric ompk q mod Imogac of pocquorar modific optopacq]pocqormosrof ric qafobsjoq morificqo Mnop[romi]jRopk quifogaf]pocrio omjjmuoje8

Name of the Form	London Branch	SA/NV
Addresses for Service of Notice	Schedule 1, Parts A(i),	Schedule 1A, Parts A(i),
and Instructions (Provider)	<u>A(ii)</u>	<u>A(ii)</u>
Confirmation Instructions	Schedule 1, Part B(i)	Schedule 1A, Part B(i)
(Transactions) (Provider)		
Confirmation Instructions (Loans /	Schedule 1, Part B(ii)	Schedule 1A, Part B(ii)
Other Transactions) (Provider)		
Transfers of Cash from Collateral	Schedule 1, Part C(i)	Schedule 1A, Part C(i)
Manager to Collateral Provider		
Transfers of Cash from Collateral	Schedule 1, Part C(ii)	Schedule 1A, Part C(ii)
Provider to Collateral Manager		
Addresses for Service of Notice	Schedule 2, Parts A(i).	Schedule 2A, Parts A(i).
and Instructions (Receiver)	A(ii)	A(ii)
Confirmation Instructions	Schedule 2, Part B(i)	Schedule 2A, Part B(i)
(Transactions) (Receiver)	Schodulo 2 Doct P(ii)	Sahadula 2A Bart D(ii)
Confirmation Instructions (Loans / Other Transactions) (Receiver)	Schedule 2. Part B(ii)	Schedule 2A. Part B(ii)
Transfers of Cash from Collateral	Schedule 2, Part C(i)	Schedule 2A, Part C(i)
Manager to Collateral Receiver	Ochleddie 2, ? are Ott	Sofredule 2A, Car On
Transfers of Cash from Collateral	Schedule 2, Part C(ii)	Schedule 2A, Part C(ii)
Receiver to Collateral Manager		Summing the sum of the
Fee Schedule	Schedule 3, Part A	Schedule 3A, Part A
Notices	Schedule 3, Part B	Schedule 3A, Part B
Conditions Precedent Documents	Schedule 3, Part C	Schedule 3A, Part C
Global Custody Network	Schedule 3, Part D	Schedule 3A, Part D
Countries and Sub-Custodians	- 2007	
English law: Form of Notice of	Schedule 4	Schedule 4A
Pledge where Collateral is		
Korean Collateral		
English law: Form Of	Schedule 4	Schedule 4A
Acknowledgment And Consent		
	Schedule 5	Schedule 5A
Security (Belgian Law) (English	man overweed to the state of th	WAS ELECTRONIC ON THE TRANSPORTED TO CONTRACT OF THE TRANSPORT
Law) Service Module		

#### 4 Reports and Statements

Where the applicable Service Module is a Custody Service Module, this Clause 4 (Reports and Statements) (as amended herein) shall also apply.

- 4.1 Rfc cpt.gac Npmt.gbcp.qf]jj npmt.gbc wms u.gf Statement of Assets]|b Statement of Account, Rfc r]rck.clr mod qqcrq]|b r]rck.clr mod aamslr qf]jj glajsbc) u.gfmsr jgk.gn]rgml) gl.dmpk]rgml gbclrogdwoje rfc Npmcprw.fcjb)]|b rfc rpj|q]argmlq coolcarcb, Rfc cpt.gac Npmt.gbcp"q mljwm\_jge]rgmlq rmn.pmt.gbc wms u.grf gl.dmpk]rgml pcj]rdje rmr.fc]qqcrq gl. rfc aamslrq qf]jj\_c rfmqc cvnpcqqjwqcrmsr.gl.nfc cpt.gac epcek.clr,
- 4.2 Whosk]w)\_w Upogrel Georpsaromiq) peosoper ric epitogae Npmtobep minpmtobe mie mpk mpe rirek eirop mod qqerqol peqnear mod Ajopeir easpogoeq]q]r]b]re mpb]req qneagonopeb ol qsaf Upogrel Georpsaromiq &\_eole]b]re mpb]req mimp]drepric epitogae Npmtobeporgaeonor mod qsaf Upogrel Georpsaromiq', Ric epitogae Npmtobeporgijjnpmtobe qsaf rirek eiroka' mod qqerq uorfol 3 &ologic'@sqoleqqB]wq]drepricb]re]q]r ufogaf qsaf rirek eiroganpenipeb, Whosqfjjj

pcojx\_spacerfc cptopac Npmtojbcpomprfcamqrmdnpmtojbojle]|wqsaf r]rckclr&q'md qqcrqnspas]|rmmrfognAj]sqc2,0,

- 4.3 Whis qf]jjnpmknrjwev]kojce]af qsaf r]rekelrmd qqerq]lb r]rekelrmd aamslr&llb) od rfe eptogae Nophtologep nophtologeq] amloogk]rogni md] caspogocq rp]lq]arogni Ymp]l DV Rp]lq]arogni)Ze]af qsaf amloogk]rogni', Whis qf]jjlmrogdwrfe eptogae Nophtologepoliu pogolje md]lw eppmp mpmkogqogni oli rfe r]rekelrmd qqerq mpr]rekelrmd aamslru ogrfoj 7. b]wq md peaeogni mdqsaf r]rekelrmpamloogok]rogni,
- 4.4 Gidmpk ] right poosignob ml whosp quasipgwight reporting

Where the applicable Service Module is the Dealer Custody (English law) Service Module or the Receiver-only Custody (English law) Service Module, the following paragraph (a) applies:

Uf cpc rfc cpt gac Npmt obcopf njbq ] qqcrq oli q] obci ccnolie chopwors) rfc cpt gac Npmt obcop u gji npmt obco oli chopk ] rogni rmwons oli rfc chopk modqr] rck clrq ] qpcocpc lacb ] \_mtc oli Aj] sqc 2,/, Rf og oli chopk ] rogni oli ajsbcq & sr og l mr joly opcb rmi ] Rf opcb N] pw caspogworcpcqr, qcaspogworcpcqr chop rf og nspnmopc oli ajsbcq ] lwjocl ) af] pec) njcbec mpk mpme] ec, of mposp mr oli ajsbc rf og oli chopk ] rogni oli wnsp] aamsl r qr] rck clrq) rfc cpt opcb Npmt obcop l ccbq rm pocacoj c bcr] ojcq opmk wns mod rfc] aamsl r u opf rfc cpt opcop u fopef Doli] laolij of oppsk clrq qs kocar rm] lwRf opcb N] prw caspogworcpcqr] pc f cjb,

Where the applicable Service Module is the Dealer Custody (Belgian law) Service Module or the Receiver-only Custody (Belgian law) Service Module, the following paragraph (a) applies:

- Uf cpc rfc cpt gac Npmt opcpf mjbq ] qqcrq ot q] dcic condie dmpwms) Kop DoB Gedurpmbsacq ] pcosopock clr ml rfc cpt gac Npmt opcp rm ot ajsbc of dmpk ] roml of wmsp ] aamsl r qr] rck clrq &qclr mwms ] rjc] qr ml ] os] prepjw\_] qogn' ml acpr] ot qcasopow of repcqrq apc] rcb mt cp wmsp Dot] ladji Goqrpsk clrq &X op ObB Bcjce] rcb Pcesji roml) progajc 4180' &c'', saf qr] rck clrq ] pc ] q pcocpcl acb ] \_mtc of Aj] sqc 2,/, Rfog of dmpk ] roml of ajsbcq &\_srog | mrjok opcb rml ] Rfog b N] prw casopow Grepcqr, qcasopow of repcqr dmp rfog nspnmqc of ajsbcq ] lwjocl) af] pec) njebec mpk mpre] ec, Gompbep rmod ajsbc rfog of dmpk ] roml of wmsp] aamsl r qr] rck clrq) rfc cpt gac Npmt opcp loebq mmpcacolic bcr] opc dpmk wms md rfc ] aamsl r u opf rfc cpt gac Npmt opcp of uf gaf Dot] ladji Goqrpsk clrq qs hcar rm] lw Rfog b N] prw casopow Grepcqr] pc f cjb,
- (c) Whis ] po poosocqrob Imrimiqo Ib] Iwojichnok ] rogni u fogaf og amiobo cirojij mpu fogaf whis ] po Imrino pk girob imiqo Ib imir fic o cptogac Npmtobo cps Ibcp] Iw] nnjoga] \_jcj] u mppocesj] rogni, Dospfopk mpo) whis ] po poosocqrob Imrimiqo Ibojichnok ] rogni pcj] rojne imigogefirq moliqor modo) porro Irogni molirogic) rogjo rojli qobp amjj] ropnj j ppj Ieck cir) qr] Ib\_w jorrop moliapobog) es] pj Iroc mpojibok Iogw) mp] Iwojichnok ] rogni pcj] rojle imir pj Iq] arogniq rmi u fogaf if coptogac Npmtobo pogrejologij n] prw
- Ufcpc wms direllormnpmt object copt gase Npmt objecpu gef ber ] oigq md] Rf geb N] pw caspogw Grepcqr) griqfmsjb\_cqclrrmmfc cpt gase Npmt objecp] q beqapog\_cb oli Aj] sqc 3 & Notices and Other Communications') mpmmqsaf mmfcp] bbpcqq mpoli qsaf mmfcpk] II cp] q mfc cpt gase Npmt objecpk] w Immoglwmmwns objeck rolls cmmolic & objectq gase w Immoglogian militire Godnpk] rognt Uc qoge',
- (e) Ufcpc wns f]to npmtoboch fo cptopac Npmtobocpu grf olichpk ]rogni olippi]rogni rm.] Rfopb N]pw caspogwoGropcqr)wns qfmsjbllmoplwrfo cptopac Npmtobocpmd]lwaf]leo oliqr]rsq) dmpcv]k njo ]lw ropkolj]rogni mpbogqaf]peo mdrfo pojot]lr qoaspogwoj ropcqr, Odrfo cptopac Npmtobocp poacotocq Imiqsaf Imroptopal tomk wns)rfo cptopac Npmtobocpu oji

] qqsk c rf]r rfc glothpk] rgmi j] qr qsnnjopcb\_wwns pck] glq amppocar, Ufcpc wms bml mr ]| b f]tc | mr npmtopcb rfc cpt gac Npmtopcpu grf glothpk] rgmi gl pcj] rgmi rm] | wRfgdb N] prw caspogrw Grcpcqr) rfc cpt gac Npmtopcpu gij] qqsk c rf]r | mRfgdb N] prw caspogw Grcpcqr f]q\_ccl epi|rcb\_wwns) gl ufgaf a]qc | mqsaf glothpk] rgmi u gij] nnc]p mi wnspqr]rck cl rq,

#### 5 Notices and Other Communications

- Sljcqqmmfcpu.gncqncagoglocbolg.nfc cpt.gac epockclr)]jj r]rckclrqmdqqcrq) r]rckclrqmdaamslr]lbamloglocbolg.ngmlqqf]jj\_colglupgglocbolglib]jj lmgacqmpmfcpamkkslga]rgmlq\_crucclrfcn]progrqqf]jj\_colglClejgnf]lbk]w\_colglcogfcpmd]jjwmpoglupgglocbolgacqi,Whsaacnrrf]rqmkckcrfmbqmdamkkslga]rgml]ckgrggmlnfpmsefAmknsrcpcptgacqi,Whsaacnrrf]rqmkckcrfmbqmdamkkslga]rgml]pclmrqcaspc]lbrfccptgacNpmtobepqf]jjoglasplmjogl\_gignwchnp&snmlwnsppcoscqr'rpjlqkgrgglocrfprkclrqmdqqcrq) r]rckclrqmdaamslrmpmfcpb]r]rmwmstogl]lwqsaflmlqcaspckcrfmb, r]rckclrqmdqqcrq]lbrjrckclrqmdaamslrugjj\_cqclrrmwmsml]ncpombga\_]qogl)qbcqapq\_cbolgAj]sqc2,/,jjr]rckclrqmdqqcrq) r]rckclrqmdaamslrugjj\_cqclrrmwmsml]ncpombga\_]qogl)qbcqapq\_cbolgAj]sqc2,/,jjr]rckclrqmdqqcrq) r]rckclrqmdaamslr)amloglocpbmrfc]bbpcqqqcrmsrogl\_afcbsjc\_nblprogramfcqcMncpjrgml]jPcpkq&Votices')ufgaf]bbpcqqk]w\_caf]lecbsnml3@sqoglcqqB]wqnpompupgrcllmgacrmrfcmfcpm]pwy
- Whis qf]jjnpmtobocnfc cptoac Npmtobocpu.opf] acprosoba] redinforchpk qcrmsroli \_afcbsjc1 mm rfcqcMncp]romi]jRcpk qamir]didienfcl]kcq]|bqncaokclqoel]rspcqmdc]af snfmpoqcbNcpqmi,Rfc cptoac Npmtobocpogl]snfmpoqcbmamknjwu.opf]|bpcjwsnmi]|wqsaflmopacqmpmnfcpamkksloal]rogmiq\_cjoctcb\_wogmnf]tc\_cclqclrmpeotcl\_w]| snfmpoqcbNcpqmi,Wnsk]w]kclbqsafacprogoba]rcmp]bb]|wncpqminmmpbcjcrc]|wncpqmiotmkqsafacprogoba]rc\_wbcjotcpde]pcnj]ackclracprogoba]rcmnfc cptosacNpmtobocpFmuctcp)slrojinfccptoacNpmtobocp]ars]jjwpcacotocqqsafpcnj]ackclracprogoba]rcmfccptosacNpmtobocpNpmtobocpk]wpcjwsnmi]|bqf]jjolasplmjol\_ojogwontppcjwolesnminfcmposoba]jacprogoba]rc,
- 5.3 Ufclotopric optigac Npmtobopogrposogob mm] ic] Iw] argmi sloopric optigac epockol r
  \_] qob snmi ric poacogni md] Imogac mpoliothyk ] rogmi optik ] Iogqasopmd] caspogwimp mmi opn] pw
  ufgaf ogni mm] n] pwmmric optigac epockol r) ric optigac Npmtobopojom \_ioge] rogmi mmqm] ar ogn
  ami bogomi] j snmi ric poacogni modqsaf Imogac mpoliothyk ] rogmi \_wifo bon] pokolii modric optigac
  Npmtobop poqnimi qojo othon pmacqqolie qsaf Imogac mpoliothyk ] rogmi impothon pmtobolie qsaf Imogac
  rmwns,

#### 6 Governing Law and Jurisdiction

- EmtcploteJ] u 8Rfc cptoac epock clr) rfcqc Mncplroml] jRcpk q] lb] lwl mi amlrplars] j m\_joe] romlq] pogole msr modmo of aml l caroml u onf rfck ] pc emtcplote\_w)] lb qf] jj\_c amlqrpscb of ] aampb] lac u onf) rfc emtcplotej] u qncaoodocb of rfc] nnjoe] \_jc cptoac Kmbsjc,
- 6.2 Hspoglogargm 8 Rf camsprq qncaogobolo of rfc] nnjga] \_jc cpt gac Kmbsjc] pc rmf] to cvajsqoloc hspoglogargm rm qorrjc] I w bognsrcq rf] rk] w] pogloc msr mod mpolog amilicargm u ogf rfc cpt gac epock clr mprfcqc Mncpl rgml] pcpk q & oglajsbolog I mi amirplars] j bognsrcq mpajjok q'] I b ] aampbologiw] I w jce] j] argmi mpnpmaccbologieq] poglogie msr mod mpologiami I cargmi u ogf rfc cpt gac epock clr mprfcqc Mncpl rgml] j Rcpk q & proceedings-'qf] jj \_c \_pmsefroliq saf amsprq, Rfc n] procq oppoct maj \_jwqs\_k og rmrfc cvajsqoloc hspoglogargmi mod saf amsprq] I b u] otc] I wm\_hcargmi rm Npmaccbologieq og qsaf amsprq u fcrfcp mi rfc epmsI b mod tolsoc mp.mi rfc epmsI b rf] r fc Npmaccbologieq f] to col pmsefroli] I otgam tologolir ompsk,

# **SCHEDULE 1 TO THE OPERATIONAL TERMS**

# **Forms of Notice**

for

Collateral Management Master Agreement (Collateral Provider) Service Module

#### PART A(i) - NOTICES

# (Collateral Management Master Agreement Service Module for the Collateral Provider under [English] / [Belgian] law)

#### ADDRESSES FOR SERVICE OF NOTICE AND INSTRUCTIONS

Iw Imrogac ] srfmoon, consonable wife contogace epockel rechnoming for normal grown med an ji] replj i k ] I] eckel repet gace man ji] replj normal go pocket sekel repet gace man ji] replj normal go pocket gacel rjwedjel gd] bept go pocket gacel gd gacel rjwedjel gd] rife ] bept go pocket gacel gd gacel rjwedjel gd gacel gd gacel gacel gd g

#### TO THE COLLATERAL PROVIDER:

#### TO THE SERVICE PROVIDER:

#### Notices relating to an Event of Default:

Rfc@limdLcuWmpiKcjjml)Jmlbml@pjlaf
One Canada Square
Jmlbml)C/23 J
Slogcblojebmk
rnl8Ajc]pjlac]lbAmjjjrcpjjK]l]eckclr
Ck]oj8Triparty\_default@bnymellon.com

#### All other Notices:

Rfc@[limdLcu Wmpi Kcjjml]Jmlbml@p]laf
One Canada Square
Jmlbml)C/23 J
Slogcblojebmk
rnl8Ajc]p]lac]lbAmjj]rcp]jK]l]eckclr
Ck]oj8Amjj]rcp]jK]l]eckclr;\_lwkcjjml,amk

FMA1 September 2021

- 7 -

# PART A(ii) - INSTRUCTIONS

# TO THE SERVICE PROVIDER:

Rfc@[limdLcuWmpiKcjjml)Jmlbml@p]laf

MlcA]l]b] os]pc

Jmlbml)C/23 J

Slogcblojebmk

rrl8Ajc]p]lac]lbAmjj]rcp]jK]l]eckclr
Ck]oj8Amjj]rcp]jK]l]eckclr;\_lwkcjjml,amk

# PART B(i) - CONFIRMATION INSTRUCTIONS (TRANSACTIONS)

To: The Bank of New York Mellon, London Branch

# (Collateral Management Master Agreement Service Module for the Collateral Provider)

#### **Global Collateral Management**

# Transaction (including Reverse Stock Loan)

(Cash processed within the Custodian)

BNY Mellon expects clients to use BNYM's electronic platform to message instructions, or to instruct manually through the GUI. Only in cases where contingency measures are applied should email be an acceptable medium for instructions. The data points in the template below continue to be required for email instructions,

Attn:	The Bank of New York	k Mellon, Lor	idon Branch		
	Ajc] p] I ac ] I b Collate	eral Managen	nent		
Tel no:	To be provided during	onboarding			
Transaction de	etails:				
Collateral Rece	eivers name:	[COLLATE	RAL RECEIVER	I	
Collateral Prov	iders name:	[COLLATE	RAL PROVIDER	]	
Cash amount:			currency:		
Rate on cash a	mount:				
Cash amount c	on termination:				
Transaction co	mmencement date:				
Transaction ter	mination date:				
Cash Instructio	n (if different from defa	ault option)			
On behalf of:					
Signature of Authorised Person:					
Name of Authorised Person:					
Date:					

FMA1

September 2021

# PART B(ii) - CONFIRMATION INSTRUCTIONS (LOANS)

# To: The Bank of New York Mellon, London Branch

# (Collateral Management Master Agreement Service Module for the Collateral Provider)

#### **Global Collateral Management**

#### [Loan (including Reverse Stock Loan)] [Other Transactions]

(Cash processed outside of the Custodian)

BNY Mellon expects clients to use BNYM's electronic platform to message instructions, or to instruct manually through the GUI. Only in cases where contingency measures are applied should email be an acceptable medium for instructions. The data points in the template below continue to be required for email instructions.

Attn:	The Bank of New York	Mellon, Jmibmi @mollaf	
	Ajc] p] I ac ] I b Collate	ral Management	
Tel no:	To be provided during	onboarding	
[Loan / Other	Transaction] details:		
Collateral Rec	eivers name:	[COLLATERAL RECEIVER]	
Collateral Prov	viders name:	[COLLATERAL PROVIDER]	
[Loan Amou	int/Other Transaction details]	[Loan Currency/ Other Transaction Details]	Profile (if applicable)
On behalf of	·		
Signature of Authorised Person:			
Name of Authorised Person:			
Date:			

# **PART C**

(Collateral Management Master Agreement Service Module for the Collateral Provider)

ACCOUNT INFORMATION FOR CASH TRANSFERS

- 11 -

# Part C(i) Transfers of Cash from Collateral Manager to Collateral Provider

Rn8 YZ

Amkn]lw8 Rfc@[limdLcuWhpiKcjjml)Jmlbml@pjlaf

Ric Anjj] replj Npmtopopoj grpsarq ric eptogac Npmtopoprmk] ie ] jj rpl I qob pol mod A] qir mm\_c k] be mog nspops] Ir mmric eptogac epeck el rompric npmtogogni modanjj] replj k] I] eck el ropeptogacq rmanjj] replj npmtopopop si bep qmai jel boj e] I b penmrpl I q] aromi q rmmric] nnpmpopi re] aams I roper msr \_ejmu, Rifoga I mrogac qif] jj qsnepopebe] I w npetognsq qsaf I mrogac opmk rife Amjj] replj Npmtopep rm rife Amjj] replj K] I] ecp

CUR	AGENT	
CSP	Amppcqnmlbclr@tliL]kc8	YZ
	uogt bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq	YZ
	@clcd-aLm8&K'	
E@N	Ampocqnmlbclr@gliL]kc8	YZ
	uogt bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq8	YZ
	@clcd -a.Lm8&K'	
HNW	Amppcqnmlbclr@gliL]kc8	YZ
	uogt bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq8	YZ
	@clcd -aLm8&K'	
SB	Ampocqnmlbclr@gliL]kc8	YZ
	ugar bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq	YZ
	@clcd -aLm8&K'	

_			
	YMrf cpZ	Ampocqnmlbclr@gliL]kc8	YZ
		ugor bbpcqq8&K'	YZ
		L]rAjpAmbc8	YZ
		@clcd-al]kc]lb ugor	YZ
		] bbpcqq8	YZ
		@clcd -aLm8&K'	

(M) = Mandatory f	iel	¢
-------------------	-----	---

Ml_cf]jdmo8	YAMJJ RO	CP JNPMT0BCPZ	
Signature of Authorised Person: <sup>2</sup>		Signature of Authorised Person:	
Name of Authorised Person:		Name of Authorised Person:	
Dated:		Dated:	

 $<sup>^{\</sup>rm 2}$  Changes to account details for cash transfers must be signed by two Authorised Persons.

# Part C(ii) Transfers of Cash from Collateral Provider to Collateral Manager

Rfc Anjj] rcplj Nomtoboco qf] jj k] ic] jj rpl I qob pa mol A] qf mm\_c k] bc mm rfc Anjj] rcplj K] I] ecp nspqs] Ir mm rfc cpt gac epock cIr] Ib ami bogomi q dno rfc nomtogomi mol anjj] rcplj k] I] eck cIr qcpt gac q manjj] rcplj nomtoboc pa sI bcp qmai jcI bog e] Ib pcnm rpl [q] aromi q rm rfc] nn pm pogi rc] aams Ir qcr msr\_cjmu, Rfog I mrogac qf] jj qsncpqcbc] I wn pct gmsq qsaf I mrogac dpmk rfc Amjj] rcplj K] I] ecp mr rfc Amjj] rcplj Npmtobocp

CUR	AGENT	
AFD	Apobog sogqo E)Xspogaf	&ugdt8APC AFXX6. '
	-ARfc@glimdLcuWmpiKcjjml -LT	&ugdr8067TR@C@@
	-A L m	
	DmpDspfcpapcbg/rm]-alm,	
	@ L8	
АВ	A]l]bojl Osnocojj@tlimdAmkkopac)Rmpmlm	&uger8AQ@AARR
	GodjtmspmdA@@AKcjjml	&ugd+8KCJLS 1NE '
	DDARfc@glimdLcuWmpiKcjjml -LT	& ugort8067TR@C@@
	-A L m	
	Dmpospofcpapcbogrm]-alm,	
		A
CSP	Rfc@ilimdLcuWmpiKcjjml -LT	& ugot80PTR@C@@
	-A@LWKcjjmlEjm_]jRpogN]prwAjoplrq	
	-A	
E@N	Rfc@glimdLcuWmpiKcjjml)Jmlbml	2 u at 200TDE @N/'
	mp: Ambc85003	& ugerl8067TRE@00V'
	-ARfc@glimdLcuWmpiKcjjmt -LT	& ugorl 80PTR@C@@
	-A	
	DmpDspf cpapcbg rm] -a l m	
	@ L8	
HNW	@gllimdRmiwm Kogqs_ognfgJoktogcbRmiwm	&ugdr8@MRIHNHR'
	-ARfc@glimdLcuWmpiKcjjmt -LT	& ugorl 8067TR@C@@
	-A L m,	
	DmpDspfcpapcbgrm]-alm,	
	@ L8	
CI	i] bo[j]togni]Clqiojb]@[licl)mnaifmjk	&ugdt8C CC '
	-ARfc@glimdLcuWmpiKcjjml -LT	&ugdr8067TR@C@@
	-A L m	
	DmpDspfcpapcbgrm]-alm,	
	@ L8	
SB	Rfc@[limdLcuWmpiKcjjml)LcuWmpi	&ugdr8067TRS 1L'

L]rAjpAmbc8DU.0//6
-A L n8
@ L8

Other currencies available upon request

# **SCHEDULE 2 TO THE OPERATIONAL TERMS**

# **Forms of Notice**

for

Collateral Management Master Agreement (Collateral Receiver) Service Module

#### PART A(i) - NOTICES

# (Collateral Management Master Agreement Service Module for the Collateral Receiver under [English/Belgian] law)

#### ADDRESSES FOR SERVICE OF NOTICE AND INSTRUCTIONS

Iw Imrogac ] srfmoon, consonable wife cpt gaze epock cir chop rfc nomtogromi modanji] replj k ] i] eck cir qept gazeq rmanji] replj peacotopeq sibep qrmai jel bot e ] ib penm rpli q] argmi q qf ] jj \_c qsotopago i rjweoto i gd] bbpcqqeb mmfc peacototope n] prw] ib f] ib bejot epeb mpqcir\_wnmqr mpc k ] gj rm rfc objotopago jq ] r rfc ] bbpcqqeq qneagotope \_cjmu,

#### TO THE COLLATERAL RECEIVER:

#### TO THE SERVICE PROVIDER:

#### Notices relating to an Event of Default:

Rfc@[limdLcu Wmp Kcjjml)Jmlbml@p]laf
One Canada Square
Jmlbml)C/23 J
Slogcblojebmk
rrl8Ajc]p]lac]lbAnjj]rcp]jK]l]eckclr
Ck]oj8Triparty\_default@bnymellon.com

#### All other Notices:

Rfc@[limdLcuWmpiKcjjml)Jmlbml@p]laf

MlcA]l]b] os]pc

Jmlbml)C/23 J

Slogcblojebmk

ml8Ajc]p]lac]lbAmjj]rcp]jK]l]eckclr
Ck]oj8Amjj]rcp]jK]l]eckclr;\_lwkcjjml,amk

# Part A(ii) - Instructions

#### TO THE SERVICE PROVIDER:

# PART B(i) - CONFIRMATION INSTRUCTIONS (TRANSACTIONS)

To: The Bank of New York Mellon, London Branch

# (Collateral Management Master Agreement Service Module for the Collateral Receiver)

#### **Global Collateral Management**

Transaction (including Reverse Stock Loan)

(Cash processed within the Custodian)

(Cash processed outside of the Custodian)

BNY Mellon expects clients to use BNYM's electronic platform to message instructions, or to instruct manually through the GUI. Only in cases where contingency measures are applied should email be an acceptable medium for instructions. The data points in the template below continue to be required for email instructions.

Attn:	The Bank of New Yor	k Mellon, Lond	don Branch	
	Amjj] rcpj jK] l] eck cl	r; _lwkcjjml	,amk	
Tel no:	To be provided during	g onboarding		
Transaction d	etails:			
Collateral Rece	eivers name:	[COLLATER	AL RECEIVER]	
Collateral Prov	iders name:	[COLLATER	AL PROVIDER]	
Cash amount:			currency:	_
Rate on cash a	amount:			
Cash amount o	on termination:			
Transaction co	mmencement date:			
Transaction ter	mination date:			
Cash Instruction	on (if different from def	ault option)		
On behalf of:	·			
Signature of Authorised Person:				
Name of Authorised Person:				
Date:				

FMA1

September 2021

# PART B(ii) - CONFIRMATION INSTRUCTIONS (LOANS)

#### To: The Bank of New York Mellon, London Branch

#### (Collateral Management Master Agreement Service Module for the Collateral Receiver)

#### **Global Collateral Management**

# [Loan (including Reverse Stock Loan)] [Other Transactions]

(Cash processed outside of the Custodian)

BNY Mellon expects clients to use BNYM's electronic platform to message instructions, or to instruct manually through the GUI. Only in cases where contingency measures are applied should email be an acceptable medium for instructions. The data points in the template below continue to be required for email instructions.

The Bank of New York Mellon, London Branch

rcp]jK] ]eck.cl	r; _l wk cjjml ,amk		
e provided during	onboarding		
actions] details	:		
name:	[COLLATERAL RECE	IVER]	
name:	[COLLATERAL PROV	/IDER]	
ther  s]			able)
	e provided during actions] details: name: name:	name: [COLLATERAL PROV	e provided during onboarding  actions] details:  name: [COLLATERAL RECEIVER]  name: [COLLATERAL PROVIDER]  ther [Loan Currency /Oter Profile (if applic

Attn:

#### **PART C**

(Collateral Management Master Agreement Service Module for the Collateral Receiver)

ACCOUNT INFORMATION FOR CASH TRANSFERS

- 21 -

# Part CI Transfers of Cash from Collateral Manager to Collateral Receiver

Rm8 YZ

Amkn]lw8 Rfc@[limdLcuWhpiKcjjml)Jmlbml@pjlaf

RfcAnji] replj Peacologopologropsarq rfc epit gac Npmtologopm k ] ic ] jj rpl I qob pq mod A] qf rm\_ck] bc rmog nspqs] I rmm fc epit gac epic k c I robn prfc npmtogopm mod anji] replj j peacologop s I bep qmai je I boj e ] I boj penmrpl I q] arogni q rmm fc] nnpmpologop ] aams I r qer msr \_ejmu, Rfog I mogac qf] jj qsnepqebe ] I w npet gmsq qsaf I mogac opmk in fc Anji] replj Peacologop rm rfc Anji] replj K] I] eep

CUR	AGENT	
CSP	Amppcqnmlbclr@gliL]kc8	YZ
	uger bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq	YZ
	@clcd-aLm8&K'	
E@N	Ampocqnmlbclr@gliL]kc8	YZ
	uger bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq8	YZ
	@clcd-aLm8&K'	
HNW	Ampocqnmlbclr@gliL]kc8	YZ
	uogt bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq8	YZ
	@clcd -aLm8&K'	
SB	Ampocqnmlbclr@gliL]kc8	YZ
	uogt bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq	YZ
	@clcd -a.Lm8&K'	

_			
	YMrf cpZ	Ampocqnmlbclr@gliL]kc8	YZ
		ugor bbpcqq8&K'	YZ
		L]rAjpAmbc8	YZ
		@clcd-al]kc]lb ugor	YZ
		] bbpcqq8	YZ
		@clcd -aLm8&K'	

(M) = Mandatory field

Ml_cf]jdmc8	YAMJJ RCP J PCACG	<b>T</b> CPZ	
Signature of Authorised Person: <sup>3</sup>		Signature of Authorised Person:	
Name of Authorised Person:		Name of Authorised Person:	
Dated:		Dated:	

 $<sup>^{3}</sup>$  Changes to account details for cash transfers must be signed by two Authorised Persons.

# Part C(ii) Transfers of Cash from Collateral Receiver to Collateral Manager

Rfc Amjj] rcplj Pcacotocp qf] jj k] ic] jj rpl I qob po mod A] qf mm\_c k] bc mm rfc Amjj] rcplj K] I] ecp nspops] Ir mm rfc cpt gac epock cIr] Ib aml bogomi q dmp rfc npmt gagomi mod amjj] rcplj k] I] eck cIr qcpt gac q mannjj] rcplj pcacotocpq sI bcpqmai jcl botocp = II b pcnmmpl Iq] argmi q mm rfc] nnpmmpot rc] aamsIr qcr msr\_cjmu, Rf ga I mrogac qf] jj qsncpqcbc] I wnpct gmsq qsaf I mrogac dpmk rfc Amjj] rcplj K] I] ecp mr rfc Amjj] rcplj Pcacotocp

CUR	AGENT	
AFD	Apobog sogqo E)Xspogaf	&ugdt8APC AFXX6. '
	-ARfc@glimdLcuWmpiKcjjml -LT	&ugdr8067TR@C@@
	-A L m	
	DmpDspfcpapcbgrm]-alm,	
	@ L8	
ΑВ	A]l]bojl Osnocpojj@glimdAmkkopac)Rmpmlmm	&ugdr8A@@AARR'
	GodjtmspmdA@@AKcjjml	&ugd+8KCJLS 1NE '
	DDARfc@glimdLcuWmpiKcjjml -LT	& ugdr80PTR@C@@
	-A L m	
	Dmpdspfcpapcbgrm]-alm,	
	@ L8	
CSP	Rfc@llimdLcuWmpiKcjjml -LT	& ugdr8067TR@C@@
	-A@LWKcjjmlEjm_]jRpgN]prwAjoplrq	
	-A	
	@ L8	
E@N	Rfc@[limdLcu WmpiKcjjmt)Jmlbml	&.ugdr8067TRE@00V'
	mpr Ambc85 003	
	-ARfc@glimdLcuWmpiKcjjml -LT	& ugdr8067TR@C@@
	-A	
	DmpDspfcpapcbgrm]-alm,	
	@ L8	
HNW	@glimdRmiwmKgnqs_gqfgJokgncbRmiwm	&ugdr8@MRIHNHR'
	-ARfc@tlimdLcuWmpiKcjjmt -LT	&ugdr8067TR@C@@
	-A L m	
	DmpDspfcpapcbgrm]-alm,	
	@ L8	
CI	i]lbo[]toqii]Clqiojb]@[licl)mmaifmjk	&ugdt8C CC'
	-ARfc@tlimdLcuWmpiKcjjmt -LT	& ugdr80PTR@C@@
	-A L m	
	DmpDspfcpapcbgrm]-alm	
	@ L8	
SB	Rfc@[limdLcuWmpiKcjjmt]LcuWmpi	&ugdr8067TRS 1L'

L]rAjpAmbc8DU.0//6
-A L n8
@ L8

Other currencies available upon request

#### **SCHEDULE 3 TO THE OPERATIONAL TERMS**

#### Notices, Fee Schedule and Conditions Precedent documents

for

Dealer Custody (English / Belgian law) Service Module and Receiver-only Custody (English / Belgian law) Service Module

# 1 PART A x Fee Schedule

# THE BANK OF NEW YORK MELLON, LONDON BRANCH SERVICE AGREEMENT ([English]/[Belgian] law) WITH

[Insert Name of the customer]
FEE SCHEDULE
\* \* \* \* \*
20\_\_\_\_

AS AGREED SEPARATELY WITH YOU

# 2 PART B - Notices

# THE BANK OF NEW YORK MELLON, LONDON BRANCH SERVICE AGREEMENT ([English]/[Belgian] law) WITH

Inser	t Nan	ne o	f the	cu:	stomer]
	ľ	TOP	ICE	S	
	*	*	*	*	
					20
	*	*	*	*	

TO THE BANK OF NEW YORK MELLON, LONDON BRANCH

Address: One Canada Square London E14 5AL

England

Attn: Clearance and Collateral Management C k ] ġ8 Anjj] rcḍ jK] l ] eck cl r; \_l wk cjjml ,amk

Attn: Telephone: Facsimile:

#### 3 PART C - Conditions Precedent Documents

# THE BANK OF NEW YORK MELLON, LONDON BRANCH SERVICE AGREEMENT ([English]/[Belgian] law) WITH

# [Insert Name of the customer] CONDITIONS PRECEDENT DOCUMENTS

- 1. In relation to you:
  - (i) a copy of a resolution of your board that constitutes you authorising the execution and delivery of the Service Agreement and the performance of its terms and authorising a named person or persons to sign the Framework Agreement and any documents to be delivered by you pursuant thereto; the resolution (a sample copy of which is attached hereto) should be certified as a true copy by the Company Secretary or other duly authorised officer of yourself; and
  - (ii) a certificate of a duly authorised officer of yourself (a sample of which is attached to the Service Agreement) setting out the names and signatures of the persons authorised to sign the Service Agreement, and any documents to be delivered by such authorised persons pursuant thereto, on behalf of yourself.
- Such other documents and legal opinions (such legal opinions to be obtained and furnished at the cost of yourself) as the Service Provider may reasonably require in relation to the Service Agreement.

#### SIGNATURE CERTIFICATE

I hereby certify to of	&fc y thorised to execut llowing persons, a pany, holding the o af qsaf ncpmi]	Company-) [ [ [ e this Certificate s of the date her office of the Com	[[[[[[[ e on behalf reof, is a du npany set c	[[[[[[]] of the Cor uly elected opposite hi	[[[[[]]] npany, and l, qualified a s name belo	b rf]r)]q further certify and acting ow and that
<u>Name</u>	Office Directory Secretary	<u>Signat</u>	<u>ture</u>			
Signed this		20	_			
Secretary						

# **CORPORATE RESOLUTION**

I hereby certify to The Bank of New Directors of, duly adopted 20 and that such resolu	, a company the following resolution	registered under the	e laws of
RESOLVED that any of this company, acting (alone)(jointly), into a Framework Agreement with T substantially in the form attached here same shall deem necessary or desiral person to be conclusive evidence of substantial person to be conclusive evidence.	The Bank of New York eto with such changes to the execution of suble, the execution of su	authorised and empower Mellon[, London Brand nereto as the person ex	ered to enter ch][ SA/NV] kecuting the
Name	<u>Title</u>		
and RESOLVED that any	anch][SA/NV] those offic tructions under such Se withdrawal of cash and t thout limitation as to pric	ers, employees and othervice Agreement, including the deposit, withdrawal, e, items or conditions ar	er agents of ling, without purchase or
Name	<u>Title</u>		
and RESOLVED that notice of any charantee of New York Mellon[, London Electron of London Branch][ SA/NV] has actually resolutions.	Branch][ SA/NV] and, ur	til The Bank of New Y	ork Mellon[,
IN WITNESS WHEREOF, I have here the corporate seal this			and affixed
(CORPORATE SEAL)			
Secretary			

4 PART D - Global Custody Network Countries and Sub-Custodians

THE BANK OF NEW YORK MELLON, LONDON BRANCH
SERVICE AGREEMENT
([English][Belgian] law)
GLOBAL CUSTODY NETWORK
COUNTRIES AND SUB-CUSTODIANS
FOR

COUNTRIES AND SUB-CUSTODIANS FOR	
[Insert Name of the customer]	
20	

# **SCHEDULE 4 TO THE OPERATIONAL TERMS**

Forms where collateral is Korean Collateral

for

Security (English law) Service Module

#### FORM OF NOTICE OF PLEDGE WHERE COLLATERAL IS KOREAN COLLATERAL

To: [name of Korean Custodian]

Cc: The Bank of New York Mellon, London Branch as Pledgee

Date: [ ]

Dear Sirs,

Security Agreement dated [ ] between [ ] (the "Pledgor") and The Bank of New York Mellon, London Branch (the "Pledgee"), as amended from time to time (the "Security Deed")

1. By the Security Deed, the Pledgor (a) has granted to the Pledgee a continuing first priority perfected security interest (*jilkwon*) in all its rights, title, interest and benefit (present and future, actual or contingent) in, to, under and in respect of certain accounts (details of which are set out below, the yCash Accounts—) u gf Y] k c mdl mpc] I Asqmbg I Z&f c yKorean Custodian—and all amounts from time to time standing to the credit of the Cash Accounts, together with any interest accruing from time to time thereon and (b) has pledged, and agrees to pledge, to the Pledgee all its rights, title, interest and benefit (present and future, actual or contingent) in, to, under and in respect of all the securities (the yPledged Securities—) credited or to be credited from time to time to the securities accounts (details of which are set out below, the ySecurities Account—) established in the name of the Pledgor or in the name of the Pledgee with the investment registration certificate of the Pledgor with the Korean Custodian.

#### **Details of Cash Accounts**

[insert account details (account name and account number)]

#### **Details of Securities Accounts**

#### [insert account details (account name and account number)]

- We hereby inform you that we have obtained consent from the Pledgee that, without prejudice to the validity and effectiveness of the pledge established by the Security Deed, the Pledgor may receive, withdraw or otherwise transfer any amounts standing to the credit of the Cash Accounts or any Pledged Securities credited to the Securities Accounts until you receive a written notice signed by the authorised signatory of the Pledgee (the yControl Notice) specifying that, with effect from the service of the Control Notice, you shall act solely on the directions of the Pledgee provided to you from time to time.
- **3.** We irrevocably authorise and instruct you:
  - (a) to do all things that are reasonably necessary or desirable to perfect the security interest in the Pledged Securities credited or to be credited from time to time to each of the Securities Accounts including without limitation adding the words "All securities credited to this securities account pledged to The Bank of New York Mellon, London Branch with its address at One Canada Square, London, E14 5AL, United Kingdom" in the account book of such Securities Account to signify that the

FMA1

September 2021

Pledgor has granted to the Pledgee a continuing first priority, perfected security interest (*jilkwon*' & ] jj mdrf c Njcbempq rajc) paper rq ] I b & rcpcqr in, to and under the Pledged Securities that have been credited or will be credited to such Securities Account (the **Pledge Statement**-);

- (b) to deem the Pledge Statement to be repeated each time any Pledged Security is credited to the relevant Securities Account without any further notice or action by the Pledger or the Pledgee;
- (c) that the Securities Accounts are under the control of the Pledgee in accordance with the terms of the Security Deed and this notice;
- (d) to comply with the terms of any notice, instruction or communication in any way relating to, or purporting to relate to, the Pledged Securities, the Securities Accounts and/or the Cash Accounts which the Korean Custodian receives at any time from an authorised signatory of the Pledgee, who are duly authorised to act on behalf of the Pledgee, without any reference to or further authority from the Pledgor and without any inquiry by the Korean Custodian as to the justification for or validity of such notice or instruction, including the instruction to realise the Pledged Securities, to deliver the Pledged Securities to the Pledgee or any person the Pledgee nominates and to such account designated by the Pledgee or such nominee;
- if requested by the Pledgee, to send all statements and other notices relating to any
  of the Cash Accounts, the Securities Account and the Pledged Securities to the
  Pledgee;
- (f) to disclose to the Pledgee without any reference to or further authority from the Pledgor and without any enquiry by you as to the justification of such disclosure, such information relating to the Cash Accounts, the Securities Accounts and/or the Pledged Securities as the Pledgee may at any time and from time to time request;
- (g) with effect from the receipt by you of the Control Notice, the Korean Custodian shall act solely on the directions of the Pledgee (whether such directions are contained in the Control Notice or provided to you at a later time pursuant to the provisions of paragraph (d) above), provided always that the Pledgor may make deposits into, and withdrawals from, a Securities Account or a Cash Account until you receive a Control Notice from the Pledgee;
- (h) on or following delivery of a Control Notice, to not act upon any instruction received from the Pledgor (including, but not limited, to any instruction in respect of any withdrawal from a Securities Account or a Cash Account) without the prior written instructions of the Pledgee;
- (i) on or following delivery of a Control Notice, to allow the Pledgee to exercise (in the name of the Pledgor and without any further consent or authority on the part of the Pledgor) any voting rights and any powers or rights which may be exercised by the legal or beneficial owner of any Pledged Securities, any person who is the holder of any Pledged Securities or otherwise;

(k) (following delivery of a Control Notice, subject to any written instructions to the contrary given by the Pledgee) to deposit directly into the relevant Securities Account or a Cash Account held by the Korean Custodian which is itself the subject of a security interest under the Security Deed:

> any dividend, interest or other distribution paid or payable in relation to any Pledged Securities and any right, money or property accruing or offered at any time in relation to any Pledged Securities by way of redemption, substitution, exchange, bonus or preference, under option rights or otherwise; and

> (ii) all proceeds of any disposal paid or payable in relation to any Pledged Securities; and

(l) to comply with the provisions of this notice.

4. None of the above instructions or the terms of this notice may be revoked, amended, varied or waived unless the prior written consent of the Pledgee has been obtained and notified to you by the Pledgee.

5. This notice shall be governed by and construed in accordance with Korean law.

Additionally, please acknowledge receipt of this notice and your agreement to the terms of this notice and the attached acknowledgment and consent by (a) signing a copy of this notice, (b) signing the acknowledgment and consent and (c) returning them to the Pledgee.

Yours faithfully

[PLEDGOR]: [\*]

By:
Name:
Title:

Acknowledged by:

[Korean Custodian]
By:

Dated:
[Fixed Date Stamp]

#### FORM OF ACKNOWLEDGMENT AND CONSENT

To: The Bank of New York Mellon, London Branch as Pledgee (as defined in and construed in accordance with the Notice referred to below)

We acknowledge receipt of a notice from [ ] (the yPledgor-) to us in the terms set out in the attached notice (the yNotice-) and consent to the creation of pledge on the Cash Accounts and the Pledged Securities credited or to be credited to the Securities Accounts according to the terms of the Notice. Unless otherwise defined herein, terms and expressions defined in or construed for the purposes of the Notice shall have the same meaning herein.

#### We confirm that:

- (a) we have not received notice of any previous pledges or assignments of, charges over, trusts in respect of or any other third party interest in, any of the Cash Accounts, the Securities Account and the Pledged Securities and we will not, without the Pledgee's prior written consent, (i) consent to any further pledge or assignment of, charges over, trust in respect of or any other third party interest in, any of the Cash Accounts, the Securities Account and the Pledged Securities and (ii) amend or vary any rights attaching to any of the Cash Accounts and the Securities Accounts; and
- (b) we agree and undertake to be bound by the terms of the Notice and, with effect from the receipt by us of a Control Notice signed by an authorised signatory of the Pledgee, we will act only in accordance with the instructions given by an authorised signatory of the Pledgee in relation to the Cash Accounts, the Securities Account and the Pledged Securities.

#### Further, we confirm that:

- (a) we will hold all Pledged Securities in the Securities Accounts subject to the provisions of the Notice;
- (b) we have neither claimed nor exercised, nor will claim or exercise, any security interest, setoff, counter-claim or other right in respect of any of the Cash Account, the Securities Account and the Pledged Securities;
- (c) we will not permit any amount of Pledged Securities to be withdrawn from a Securities Account in breach of any terms of the Notice or any authorised written instructions of the Pledgee; and
- (d) if we receive any instruction from the Pledgor which conflicts with any instruction previously received from the Pledgee or the terms of the Notice, we shall (i) promptly notify the Pledgee and (ii) not give effect to the Pledgor's instructions without the Pledgee's consent.

This acknowledgment shall be governed and construed in accordance with Korean law.

For and on behalf of

#### [Korean Custodian]

By:

FMA1

September 2021

Dated: [Fixed Date Stamp]

# **SCHEDULE 5 TO THE OPERATIONAL TERMS**

# Relevant notices to be inserted

for

Security [English law] / [Belgium Law] Service Module

# YNSERT RELEVANT NOTICESZ

# SCHEDULE 1A TO THE OPERATIONAL TERMS

# **Forms of Notice**

for

Collateral Management Master Agreement (Collateral Provider) Service Module

#### PART A(i) - NOTICES

# (Collateral Management Master Agreement Service Module for the Collateral Provider under [English] / [Belgian] law)

#### ADDRESSES FOR SERVICE OF NOTICE AND INSTRUCTIONS

Iw Imrogac ] srfmoon, consonable wife contogace epockel rechnoming for normal grown med an ji] replj i k ] I] eckel repet gace man ji] replj normal go pocket sekel repet gace man ji] replj normal go pocket gacel rjwedjel gd] bept go pocket gacel gd gacel rjwedjel gd] rife ] bept go pocket gacel gd gacel rjwedjel gd gacel gd gacel gacel gd g

#### TO THE COLLATERAL PROVIDER:

#### TO THE SERVICE PROVIDER:

#### Notices relating to an Event of Default:

Rfc@]limdLcuWmpiKcjjml -LT
24 PscKmlmwcpqrp]]r
@/...@psqqcjq
@cjegsk
rrl8Ajc]p]lac]lbAmjj]rcp]jK]l]eckclr
Ck]gi8Triparty\_default@bnymellon.com

#### All other Notices:

Rfc@]limdLcuWmpiKcjjml -LT
24 PscKmlrmwcpqrpj]r
@/...@psqqcjq
@ojeosk
rrl8Ajc]pjlac]lbAmjj]rcpjjK]l]eckclr
Ck]oj8Amjj]rcpjjK]l]eckclr;\_lwkcjjml,amk

FMA1 September 2021

- 41 -

### PART A(ii) - INSTRUCTIONS

### TO THE SERVICE PROVIDER:

Rfc@tlimdLcu WmpiKcjjml -LT
24 Psc Kmlmwcpqrp]]r
@/...@psqqcjq
@cjegsk
rrl8Ajc]p]lac]lbAmjj]rcp]jK]l]eckclr
Ck]gi8Amjj]rcp]jK]l]eckclr;\_lwkcjjml,amk

# PART B(i) - CONFIRMATION INSTRUCTIONS (TRANSACTIONS)

To: The Bank of New York Mellon SA/NV

# (Collateral Management Master Agreement for the Collateral Provider)

#### **Global Collateral Management**

### Transaction (including Reverse Stock Loan)

(Cash processed within the Custodian)

BNY Mellon expects clients to use BNYM's electronic platform to message instructions, or to instruct manually through the GUI. Only in cases where contingency measures are applied should email be an acceptable medium for instructions. The data points in the template below continue to be required for email instructions.

Attn:	The Bank of New York	k Mellon SA/N	1/		
	CollateralManagemen	it@bnymellon	.com		
Tel no:	To be provided during	onboarding			
Transaction d	letails:				
Collateral Rec	eivers name:	[COLLATER	AL RECEIVER]		
Collateral Prov	riders name:	[COLLATER	AL PROVIDER]		
Cash amount:			currency:	_	
Rate on cash a	amount:				
Cash amount o	on termination:				
Transaction co	mmencement date:			<u></u>	
Transaction te	rmination date:				
Cash Instructio	on (if different from defa	ault option)			
On behalf of	:				
Signature of Authorised Person:					
Name of Authorised Person:					
Date:					

# PART B(ii) - CONFIRMATION INSTRUCTIONS (LOANS)

To: The Bank of New York Mellon SA/NV

# (Collateral Management Master Agreement Service Module for the Collateral Provider)

#### **Global Collateral Management**

### Loan (including Reverse Stock Loan)

(Cash processed outside of the Custodian)

BNY Mellon expects clients to use BNYM's electronic platform to message instructions, or to instruct manually through the GUI. Only in cases where contingency measures are applied should email be an acceptable medium for instructions. The data points in the template below continue to be required for email instructions,

Attn:	The Bank of New	York Mellon -L T	
	CollateralManage	ment@bnymellon.com	
Tel no:	To be provided de	uring onboarding	
Loan detai	ils:		
Collateral F	Receivers name:	[COLLATERAL RECEIVER]	
Collateral F	Provider's name:	[COLLATERAL PROVIDER]	
	Loan Amount	Loan Currency	Profile (if applicable)
On behal	f of:		
Signature Authorise Person:			
Name of Authorise Person:	ed 		
Date:			

### **PART C**

(Collateral Management Master Agreement Service Module for the Collateral Provider)

ACCOUNT INFORMATION FOR CASH TRANSFERS

- 45 -

# Part CI Transfers of Cash from Collateral Manager to Collateral Provider

Rm8 YZ

Amkn]lw8 Rfc@llimdLcuWmpiKcjjml -LT

Ric Anjj] replj Npmt opepoj grpsarg ric ept gac Npmt opep mmk ] ic ] jj mpl l qob pg mdA] gr mm\_c k ] be mog nspqs] Ir mmric ept gac epeck clir chop ric npmt opgorn indanjj] replj k ] l] eck clir qept gac qimanjj] replj npmt opep slibep qimai jel boj e ] l bipen mmril l qliangmi qimiric ] nn pmmpoj re ] aamslir qer msr \_cjmu, Rif og limigac qif ] jj qsnepqebe ] liwinpet omsq qsafi limigac qbmk informatji] replj Npmt opep mmiric Amjj] replj K] l] ecp

CUR	AGENT	
CSP	Amppcqnmlbclr@tliL]kc8	YZ
	uogt bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq	YZ
	@clcd-aLm8&K'	
E@N	Ampocqnmlbclr@gliL]kc8	YZ
	uogt bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq8	YZ
	@clcd -a.Lm8&K'	
HNW	Ampocqnmlbclr@gliL]kc8	YZ
	uogt bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq8	YZ
	@clcd -aLm8&K'	
SB	Ampocqnmlbclr@gliL]kc8	YZ
	ugar bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq	YZ
	@clcd -aLm8&K'	

_			
	YMrf cpZ	Ampocqnmlbclr@gliL]kc8	YZ
		ugor bbpcqq8&K'	YZ
		L]rAjpAmbc8	YZ
		@clcd-al]kc]lb ugor	YZ
		] bbpcqq8	YZ
		@clcd -aLm8&K'	

(M) = Mandatory field

Ml_cf]jdmo8	YAMJJ RCP	J NPMT 0BCPZ		
Signature of Authorised Person: <sup>4</sup>			Signature of Authorised Person:	
Name of Authorised Person:			Name of Authorised Person:	
Dated:			Dated:	

 $<sup>^{\</sup>rm 4}$  Changes to account details for cash transfers must be signed by two Authorised Persons.

# Part CII Transfers of Cash from Collateral Provider to Collateral Manager

Rfc Anjj] rcplj Nontoboco qf]jj k]ic]jj rpl| qocboq mdA] qf rm\_c k] bc rm rfc Anjj] rcplj K]l] eco nspqs] l r rm rfc cptosac epock clr] l b aml bogomiq dno rfc nontogomi mdanjj] rcplj k]l] eck clr qcptosacq manjj] rcplj nontobocoq sl bcpqmaijcl bogle] l b pcnmrpl|q] aromiq rmrfc] nnpmrpoglrc] aamsl r qcr msr\_cjmu, Rfog I mrosac qf]jj qsncpqcbc] l wnpctognsq qsaf I mrosac domk rfc Anjj] rcplj K]l] ecp m rfc Anjj] rcplj Npmtoboco

CUR	AGENT	
AFD	Apobog sogqo E)Xspogaf	&ugt8APC AFXX6. '
	-ARfc@glimdLcuWmpiKcjjml -LT	& ugdr8067TR@0C@@0
	-A L m	
	DmpDspfcpapcbgrm]-alm,	
АВ	A] ]boj  Osnocpojj@glimodAmkk.cpac)Rmpmlm	&uger8A@@AARR'
	GiditmspmdA@AKcjjml	& ugt8KCJLS 1NE '
	DDARfc@glimdLcuWmpiKcjjml -LT -ALm,∎	& ugdr80PTR@C@@
	Dmpdsprf cpapcbgrm] -a I m	
CSP	Rfc@glimdLcuWmpiKcjjm1 -LT	& ugdr8067TR@0C@@0
	-A@LWKcjjml Ejm_]jRpgN]prwAjoplrq	
	-A	
	@ L8	
E@N	Rfc@ilimdLcu Wmpi Kcjjmt Jmtbmt	&ugdr8067TRE@00V'
	L]rAjpAmbc8 A5003	
	-ARfc@glimdLcuWmpiKcjjml -LT	& ugdr8067TR@C@@
	-A	
	DmpDspfcpapcbgrm]-alm,	
	IBAN:	0 10 G IDI IN II DI
HNW	KSDE @li) Jrb	& uget8@MRIHNHR
	-ARfc@tlimdLcuWmpiKcjjmt -LT -ALm,∎	& ugot80PTR@C@@
	Dmpdspf cpapcbgrm] -a I m	
CI	i] bolj]togai] Clqiojob] @]licl) maaifmjk	&ugat8CCC'
	-ARfc@llimdLcuWmpiKcjjm1 -LT	& ugdr80PTR@C@@
	-A L m	
	DmpDspfcpapcbgrm] -a Im,	
	@ L8	
S B	Rfc@llimdLcuWmpiKcjjml)LcuWmpi	&ugdr8067TRS 1L'
	L]rAjpAmbc8DU.0//6	
	-A L n8	
	PcoBSOBE C-Ejm_]jAjc]p]lac&ksqr_cosmcb'	

Other currencies available upon request

# **SCHEDULE 2A TO THE OPERATIONAL TERMS**

### **Forms of Notice**

for

Collateral Management Master Agreement (Collateral Receiver) Service Module

#### **PART AI - NOTICES**

# (Collateral Management Master Agreement Service Module for the Collateral Receiver under [English/Belgian] law)

#### ADDRESSES FOR SERVICE OF NOTICE AND INSTRUCTIONS

Iw Imrogac ] srfmoon, consonable wife cpt gaze epock cir chop rfc nomtogromi mod anjij replj k ] i] eck cir qept gazeq rm anjij replj peacotopeq sibep qrmai jel bot e ] ib penm rpliq] argmiq qf ] jj \_c qsotopago i rjweoto i gd] bbpcqqcb mmfc peacototope n] prw] ib f] ib bejot epeb mpqcir\_wnmqr mpc k ] gj rm rfc objotopago jq ] r rfc ] bbpcqqcq qncaogopb \_cjmu,

#### TO THE COLLATERAL RECEIVER:

#### TO THE SERVICE PROVIDER:

#### Notices relating to an Event of Default:

Rfc@]|imdLcuWmpiKcjjml -LT
24 PscKmlrmwcpqrpj]|r
@/...@psqqcjq
@cjegsk
rrl8Ajc]pj|ac]|bAmjj]rcpjjK]|]eckclr
Ck]gi8Triparty\_default@bnymellon.com

#### All other Notices:

Rfc@[limdLcuWmpiKcjjmlSA/NV
24 PscKmlmwcpqrp]]r
@/...@osqqcjq
@ojegsk
rrl8Ajc]p[lac]lbAmjj]rcp[jK]l]eckclr
Ck]oj8Amjj]rcp[jK]l]eckclr;\_\_lwkcjjml,amk

FMA1 September 2021

- 51 -

### Part A(ii) - Instructions

# TO THE SERVICE PROVIDER:

Rfc@[limdLcuWmpiKcjjmlSA/NV 24 PscKmlmwcpqrpj]r @/...@osqqcjq @ojeosk rrl8Ajc]pjlac]lbAmjj]rcpjjK]l]eckclr Ck]gi8Amjj]rcpjjK]l]eckclr;\_lwkcjjml,amk

# PART BI - CONFIRMATION INSTRUCTIONS (TRANSACTIONS)

To: The Bank of New York Mellon SA/NV

# (Collateral Management Master Agreement for the Collateral Receiver)

#### **Global Collateral Management**

### Transaction (including Reverse Stock Loan)

(Cash processed within the Custodian)

BNY Mellon expects clients to use BNYM's electronic platform to message instructions, or to instruct manually through the GUI. Only in cases where contingency measures are applied should email be an acceptable medium for instructions. The data points in the template below continue to be required for email instructions.

Attn:	The Bank of New York Mellon SA/NV				
	Ajc] p] I ac ] I b Collate	eral Managem	nent		
Tel no:	To be provided during	onboarding			
Transaction d	etails:				
Collateral Rece	eivers name:	[COLLATEF	RAL RECEIVER]		
Collateral Prov	iders name:	[COLLATEF	RAL PROVIDER]		
Cash amount:			currency:		
Rate on cash a	ımount:				
Cash amount c	on termination:				
Transaction co	mmencement date:				
Transaction ter	mination date:				
Cash Instructio	n (if different from defa	ault option)			
On behalf of:					
Signature of Authorised Person:					
Name of Authorised Person:					
Date:					

FMA1

September 2021

# PART B(ii) - CONFIRMATION INSTRUCTIONS (LOANS)

To: The Bank of New York Mellon SA/NV

# (Collateral Management Master Agreement Service Module for the Collateral Receiver)

#### **Global Collateral Management**

Loan (including Reverse Stock Loan)

(Cash processed outside of the Custodian)

BNY Mellon expects clients to use BNYM's electronic platform to message instructions, or to instruct manually through the GUI. Only in cases where contingency measures are applied should email be an acceptable medium for instructions. The data points in the template below continue to be required for email instructions.

Attn:	: The Bank of New York Mellon SA/NV				
	Ajc] pl l ac ] l b Colla	ateral Management			
Tel no:	To be provided duri	ing onboarding			
Loan details:					
Collateral Rec	eivers name:	[COLLATERAL RECEIVER	R]		
Collateral Pro	viders name:	[COLLATERAL PROVIDER	R]		
Loan Amou	nt	Loan Currency	Profile (if applicable)		
On behalf of	f:				
Signature of Authorised Person:	·				
Name of Authorised Person:					
Date:					

### **PART C**

(Collateral Management Master Agreement Service Module for the Collateral Receiver)

ACCOUNT INFORMATION FOR CASH TRANSFERS

# Part CI Transfers of Cash from Collateral Manager to Collateral Receiver

Rm8 YZ

Amkn]lw8 Rfc@llimdLcuWmpiKcjjml -LT

RfcAnjij replj Peacologopologropsarq nfc epitogae Nportopopormik ] ic ] jj rpl I qdepq nodA] qf rm\_ck] be rmog nspqs] Ir mrnfc epitogae epeek e Ir chop nfc nportognogon nodanjij replj jk] I ] eek e Ir qeptogaeq rmanjij replj peacologopogo slibep qmaai jel bolg e] I bipenmrpl I qlianogon qirmife ] nnpormpogore ] aanos Ir qerimsri\_cjmu, Rfog Imrogae qf] jj qsnepqebe ] I winpetogorsq qsaf Imrogae chork infe Anjij] replj Peacologop mrinfe Anjij] replj Ecp

CUR	AGENT	
CSP	Amppcqnmlbclr@gliL]kc8	YZ
	uger bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq	YZ
	@clcd-aLm8&K'	
E@N	Ampocqnmlbclr@gliL]kc8	YZ
	uger bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq8	YZ
	@clcd-aLm8&K'	
HNW	Ampocqnmlbclr@gliL]kc8	YZ
	uogt bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq8	YZ
	@clcd -aLm8&K'	
SB	Ampocqnmlbclr@gliL]kc8	YZ
	uogt bbpcqq8&K'	YZ
	L]rAjpAmbc8	YZ
	@clcd-al]kc]lb ugor	YZ
	] bbpcqq	YZ
	@clcd -a.Lm8&K'	

_			
	YMrf cpZ	Ampocqnmlbclr@gliL]kc8	YZ
		ugor bbpcqq8&K'	YZ
		L]rAjpAmbc8	YZ
		@clcd-al]kc]lb ugor	YZ
		] bbpcqq8	YZ
		@clcd -aLm8&K'	

(M) = Mandatory field

Ml_cf]jdmo8	YAMJJ RCP JPO	CACGICPZ	
Signature of Authorised Person: <sup>5</sup>		Signature of Authorised Person:	
Name of Authorised Person:		Name of Authorised Person:	
Dated:		Dated:	

<sup>&</sup>lt;sup>5</sup> Changes to account details for cash transfers must be signed by two Authorised Persons.

# Part C(ii) Transfers of Cash from Collateral Receiver to Collateral Manager

Rfc Amjj] rcplj Pcacotocp qf] jj k] ic] jj rpl I qob po mod A] qf mm\_c k] bc mm rfc Amjj] rcplj K] I] ecp nspops] Ir mm rfc cpt gac epock cIr] Ib aml bogomi q dmp rfc npmt gagomi mod amjj] rcplj k] I] eck cIr qcpt gac q mannjj] rcplj pcacotocpq sI bcpqmai jcl botocp = II b pcnmmpl Iq] argmi q mm rfc] nnpmmpot rc] aamsIr qcr msr\_cjmu, Rf ga I mrogac qf] jj qsncpqcbc] I wnpct gmsq qsaf I mrogac dpmk rfc Amjj] rcplj K] I] ecp mr rfc Amjj] rcplj Pcacotocp

CUR	AGENT	
AFD	Apcbog sogqoc E)Xspogaf -ARfc@glimodLcuWmpiKcjjm1 -LT -ALm, DmpDspnfcpapcbogrm]-alm,	&ugort8APC AFXX6. ' &ugort8OPTR@C@@O
АВ	A] ]boj  Ok ncpojj@jlimdAmkkcpac)RmpmImm OsojtmspmdA@GAKcjjml  DDARfc@glimdLcuWmpiKcjjml -LT  -ALm,  Dmpolspnfcpapcbogrm]-alm,	&ugdr8A@@AARR' &ugdr8KCJLS 1NE ' &ugdr8OPTR@C@@@
CSP	Rfc@glimdLcuWmpiKcjjml -LT -A@gLWKcjjmlEjm_]jRppgN]pwAjoplrq -A	& ugdr 80РТ R@C@@
E@N	Rfc@limdLcuWmpiKcjjmlJmlbml L]rAjpAmbc8 A5003 -ARfc@limdLcuWmpiKcjjml -LT -ADmpDspfcpapcbgrm]-alm, IBAN:	& ugort80PTRE@00V' & ugort80PTR@12@1@0
HNW	KSDE@[li)Jrb -ARfc@[limdLcuWmpiKcjjmt -LT -ALm, Dmpospofcpapcbogrm]-alm,	&ugorle@MRIHNHR' &ugorle@PTR@©@©
CI	i] bolj togi Clqiojb]@[licl)maifmjk -ARfc@[limdLcuWmpiKcjjm!-LT -ALm, DmpDsprfcpapcbogrm]-alm, @@[L8	&ugdr8CPTR@C@@
S B	Rfc@]limdLcuWmpiKcjjml)LcuWmpi L]rAjpAmbc8DU.0//6 -ALm8 Pco8SOBEC-Ejm_]jAjc]pjlac&ksqr_cosmrcb'	&ugdr8OPTRS 1L'

Other currencies available upon request

### SCHEDULE 3A TO THE OPERATIONAL TERMS

#### Notices, Fee Schedule and Conditions Precedent documents

for

Dealer Custody (English / Belgian law) Service Module and Receiver-only Custody (English / Belgian law) Service Module

### 1 PART A x Fee Schedule

THE BANK OF NEW YORK MELLON SA/NV SERVICE AGREEMENT ([English]/[Belgian] law) WITH

[Insert Name of the customer]
FEE SCHEDULE
\* \* \* \* \* 20\_\_\_\_\_

AS AGREED SEPARATELY WITH YOU

#### THE BANK OF NEW YORK MELLON SA/NV SERVICE AGREEMENT ([English]/[Belgian] law) WITH

[Insert Name of the customer] NOTICES					
	r	IOI	ICE:	>	
	*	*	*	*	
					20
	*	*	*	*	

TO THE BANK OF NEW YORK MELLON SA/NV

Address:

46 Rue Montoyerstraat B-1000 Brussels Belgium

Attn: Ajc] pl I ac ] I b Collateral Management C k ] gi8 Anjj] rcpl jK] I ] eck cl r; \_I wk cjjml ,amk

TO ::
Address::

Attn:
Telephone:
Facsimile:

#### **PART C - Conditions Precedent Documents**

#### THE BANK OF NEW YORK MELLON SA/NV SERVICE AGREEMENT ([English]/[Belgian] law) WITH

# [Insert Name of the customer] CONDITIONS PRECEDENT DOCUMENTS

- 1. In relation to you:
  - (i) a copy of a resolution of your board that constitutes you authorising the execution and delivery of the Service Agreement and the performance of its terms and authorising a named person or persons to sign the Framework Agreement and any documents to be delivered by you pursuant thereto; the resolution (a sample copy of which is attached hereto) should be certified as a true copy by the Company Secretary or other duly authorised officer of yourself; and
  - (ii) a certificate of a duly authorised officer of yourself (a sample of which is attached to the Service Agreement) setting out the names and signatures of the persons authorised to sign the Service Agreement, and any documents to be delivered by such authorised persons pursuant thereto, on behalf of yourself.
- Such other documents and legal opinions (such legal opinions to be obtained and furnished at the cost of yourself) as the Service Provider may reasonably require in relation to the Service Agreement.

	<u>SI</u>	<u>GNATU</u>	RE CERTIF	<u>ICATE</u>			
I hereby certify to The nd [[[[[[[[[[[[[[[[[[[[[[[[[[[[[[[[[[[	[[[[[[ &f c rised to execute ing persons, as ding the office	y <b>Compa</b> e this Ce of the da of the C	iny-) [[[[[ ertificate on late hereof, is Company se	[[[[[[]]]]]]]]]]]]]]]]]]]]]]]]]]]]]]]]	[[[[[[]]] he Compa ected, qua e his nam	[[[[]] any, and lified and le below	]  b   f ] r) ] q further certify acting officer and that the
<u>Name</u>	Office Directory Secretary		<u>Signature</u>				
Signed this		20					
Secretary				-			

# **CORPORATE RESOLUTION**

Directors of, duly adopted	York Mellon[, London Branch][ SA/NV] that the Board of, a company registered under the laws of the following resolutions on the day of utions are in full force and effect:
of this company, acting (alone)(jointly), into a Framework Agreement with T substantially in the form attached here	of the following officers, employees or agents be and hereby (is)(are) authorised and empowered to enter the Bank of New York Mellon[, London Branch][ SA/NV] eto with such changes thereto as the person executing the able, the execution of such Framework Agreement by such approval.
<u>Name</u>	<u>Title</u>
(alone)(jointly), be and (is)(are) hereby Bank of New York Mellon[, London Brathis company authorised to issue Inst limitation, in relation to the deposit or v	of the following persons, acting authorised from time to time to designate in writing to The anch][ SA/NV] those officers, employees and other agents of tructions under such Service Agreement, including, without withdrawal of cash and the deposit, withdrawal, purchase or thout limitation as to price, items or conditions and otherwise ovisions of such Service Agreement:
<u>Name</u>	<u>Title</u>
Bank of New York Mellon[, London E	ange in these resolutions be communicated in writing to The Branch][ SA/NV] and, until The Bank of New York Mellon[, received such notice, it is authorised to act pursuant to these
IN WITNESS WHEREOF, I have hereuthe corporate seal this	unto set my hand as secretary of said corporation and affixed day of 20
(CORPORATE SEAL)	
Secretary	

# PART D - Global Custody Network Countries and Sub-Custodians

THE BANK OF NEW YORK MELLON SA/NV SERVICE AGREEMENT ([English][Belgian] law) GLOBAL CUSTODY NETWORK COUNTRIES AND SUB-CUSTODIANS

`	FOR
	[Insert Name of the customer]
	20

### **SCHEDULE 4A TO THE OPERATIONAL TERMS**

Forms where collateral is Korean Collateral

for

Security (English law) Service Module

### FORM OF NOTICE OF PLEDGE WHERE COLLATERAL IS KOREAN COLLATERAL

[name of Korean Custodian]

To:

Cc:	The Bank of New York Mellon SA/NV as Pledgee
Date: [	1
Dear S	irs,
	ty Agreement dated [ ] between [ ] (the "Pledgor") and The Bank of New York SA/NV (the "Pledgee"), as amended from time to time (the "Security Deed")
1.	By the Security Deed, the Pledgor (a) has granted to the Pledgee a continuing first priority perfected security interest ( <i>jilkwon</i> ) in all its rights, title, interest and benefit (present and future, actual or contingent) in, to, under and in respect of certain accounts (details of which are set out below, if c yCash Accounts—'u gif Y] k c mdl mmc] I Asqmbgl Z8f c yKorean Custodian—and all amounts from time to time standing to the credit of the Cash Accounts, together with any interest accruing from time to time thereon and (b) has pledged, and agrees to pledge, to the Pledgee all its rights, title, interest and benefit (present and future, actual or contingent) gl m) sl bcp] I b gl pcqncar md] jj if c qcaspagcq 8f c yPledged Securities—'apcbgcb mpm_c credited from time to time to the securities accounts (details of which are set out below, the ySecurities Accounts—I I b c] af )] ySecurities Account—'established in the name of the Pledgor or in the name of the Pledgee with the investment registration certificate of the Pledgor with the Korean Custodian.
	Details of Cash Accounts
	[insert account details (account name and account number)]
	Details of Securities Accounts

FMA1 September 2021 [insert account details (account name and account number)]

- 2. We hereby inform you that we have obtained consent from the Pledgee that, without prejudice to the validity and effectiveness of the pledge established by the Security Deed, the Pledgor may receive, withdraw or otherwise transfer any amounts standing to the credit of the Cash Accounts or any Pledged Securities credited to the Securities Accounts until you receive a upgral I magac qoel ab \_w rf c ] srf mogac qoel ] mpw md rf c Njabecc &f c yControl Notice\_' specifying that, with effect from the service of the Control Notice, you shall act solely on the directions of the Pledgee provided to you from time to time.
- **3.** We irrevocably authorise and instruct you:
  - to do all things that are reasonably necessary or desirable to perfect the security interest in the Pledged Securities credited or to be credited from time to time to each of the Securities Accounts including without limitation adding the words "All securities credited to this securities account pledged to The Bank of New York Mellon SA/NV with its address at 46 Rue Montoyerstraat, B-1000 Brussels, Belgium" in the account book of such Securities Account to signify that the Pledgor has granted to the Pledgee a continuing first priority, perfected security interest (jilkwon' d ] jj mdrf c Njcbemp'q rajc) paper rq ] I b d rcpcqr d ) mm] I b sl bcprf c Pledged Securities that have been credited or will be credited to such Securities Account &f c yPledge Statement-'9
  - (b) to deem the Pledge Statement to be repeated each time any Pledged Security is credited to the relevant Securities Account without any further notice or action by the Pledger or the Pledge;
  - (c) that the Securities Accounts are under the control of the Pledgee in accordance with the terms of the Security Deed and this notice;
  - (d) to comply with the terms of any notice, instruction or communication in any way relating to, or purporting to relate to, the Pledged Securities, the Securities Accounts and/or the Cash Accounts which the Korean Custodian receives at any time from an authorised signatory of the Pledgee, who are duly authorised to act on behalf of the Pledgee, without any reference to or further authority from the Pledger and without any inquiry by the Korean Custodian as to the justification for or validity of such notice or instruction, including the instruction to realise the Pledged Securities, to deliver the Pledged Securities to the Pledgee or any person the Pledgee nominates and to such account designated by the Pledgee or such nominee;
  - if requested by the Pledgee, to send all statements and other notices relating to any of the Cash Accounts, the Securities Account and the Pledged Securities to the Pledgee;
  - (f) to disclose to the Pledgee without any reference to or further authority from the Pledgor and without any enquiry by you as to the justification of such disclosure, such information relating to the Cash Accounts, the Securities Accounts and/or the Pledged Securities as the Pledgee may at any time and from time to time request;

- (g) with effect from the receipt by you of the Control Notice, the Korean Custodian shall act solely on the directions of the Pledgee (whether such directions are contained in the Control Notice or provided to you at a later time pursuant to the provisions of paragraph (d) above), provided always that the Pledgor may make deposits into, and withdrawals from, a Securities Account or a Cash Account until you receive a Control Notice from the Pledgee;
- (h) on or following delivery of a Control Notice, to not act upon any instruction received from the Pledgor (including, but not limited, to any instruction in respect of any withdrawal from a Securities Account or a Cash Account) without the prior written instructions of the Pledgee;
- (i) on or following delivery of a Control Notice, to allow the Pledgee to exercise (in the name of the Pledgor and without any further consent or authority on the part of the Pledgor) any voting rights and any powers or rights which may be exercised by the legal or beneficial owner of any Pledged Securities, any person who is the holder of any Pledged Securities or otherwise;
- (k) (following delivery of a Control Notice, subject to any written instructions to the contrary given by the Pledgee) to deposit directly into the relevant Securities Account or a Cash Account held by the Korean Custodian which is itself the subject of a security interest under the Security Deed:
  - any dividend, interest or other distribution paid or payable in relation to any Pledged Securities and any right, money or property accruing or offered at any time in relation to any Pledged Securities by way of redemption, substitution, exchange, bonus or preference, under option rights or otherwise;
  - (ii) all proceeds of any disposal paid or payable in relation to any Pledged Securities; and
- (l) to comply with the provisions of this notice.
- 4. None of the above instructions or the terms of this notice may be revoked, amended, varied or waived unless the prior written consent of the Pledgee has been obtained and notified to you by the Pledgee.
- 5. This notice shall be governed by and construed in accordance with Korean law.

Additionally, please acknowledge receipt of this notice and your agreement to the terms of this notice and the attached acknowledgment and consent by (a) signing a copy of this notice, (b) signing the acknowledgment and consent and (c) returning them to the Pledgee.

Yours faithfully

[PLEDGOR]: [\*]

Ву:	
Name:	
Title:	
d by:	
todian]	
tamp]	
t	Name: Title:  Dy:  Odian]

#### FORM OF ACKNOWLEDGMENT AND CONSENT

To: The Bank of New York Mellon SA/NV as Pledgee (as defined in and construed in accordance with the Notice referred to below)

We acknowledge receipt of a notice from [ ] (the yPledgor-) to us in the terms set out in the attached I migac &f c yNotice-'] I b and qcl r minf c apc] right indicates minf c A] qf aansl rq and the Pledged Securities credited or to be credited to the Securities Accounts according to the terms of the Notice. Unless otherwise defined herein, terms and expressions defined in or construed for the purposes of the Notice shall have the same meaning herein.

#### We confirm that:

- (a) we have not received notice of any previous pledges or assignments of, charges over, trusts in respect of or any other third party interest in, any of the Cash Accounts, the Securities Account and the Pledged Securities and we will not, without the Pledgee's prior written consent, (i) consent to any further pledge or assignment of, charges over, trust in respect of or any other third party interest in, any of the Cash Accounts, the Securities Account and the Pledged Securities and (ii) amend or vary any rights attaching to any of the Cash Accounts and the Securities Accounts; and
- (b) we agree and undertake to be bound by the terms of the Notice and, with effect from the receipt by us of a Control Notice signed by an authorised signatory of the Pledgee, we will act only in accordance with the instructions given by an authorised signatory of the Pledgee in relation to the Cash Accounts, the Securities Account and the Pledged Securities.

#### Further, we confirm that:

- (a) we will hold all Pledged Securities in the Securities Accounts subject to the provisions of the Notice;
- (b) we have neither claimed nor exercised, nor will claim or exercise, any security interest, setoff, counter-claim or other right in respect of any of the Cash Account, the Securities Account and the Pledged Securities;
- (c) we will not permit any amount of Pledged Securities to be withdrawn from a Securities Account in breach of any terms of the Notice or any authorised written instructions of the Pledgee; and

FMA1

September 2021

if we receive any instruction from the Pledgor which conflicts with any instruction previously received from the Pledgee or the terms of the Notice, we shall (i) promptly notify the Pledgee and (ii) not give effect to the Pledgor's instructions without the Pledgee's consent.

This acknowledgment shall be governed and construed in accordance with Korean law.

For and on behalf of

[Korean Custodian]

By:

Dated:

[Fixed Date Stamp]

### **SCHEDULE 5A TO THE OPERATIONAL TERMS**

### Relevant notices to be inserted

for

Security [English law] / [Belgium Law] Service Module

# [INSERT RELEVANT NOTICES]

# The Bank of New York Mellon Regulatory Terms



# **Table of Contents**

Cont	rents P	age
1	Introduction	2
2	Definitions	2
3	Regulatory Information Document	2
4	Market Transparency Requirements	2
5	Concerning the Service Provider	3
6	Confidentiality, Information Sharing and Data Privacy	3
7	Provision of Information	6
8	Computer Services and electronic access	7
9	No Reliance - No Advice	7
10	Governing Law and Jurisdiction	7
Sche	dule 1 to the Regulatory Terms - BNYM Legal Entity Specific Terms	9
Sche	dule 2 to the Regulatory Terms - Personal Data Transfer	12

### 1 Introduction

Pursuant to the terms of a framework agreement executed and (to the extent applicable) delivered (a "Framework Agreement"), a Service Agreement shall take effect between you and the Service Provider in respect of each of the Service Modules elected to apply to you as set out in the Service Form.

#### 2 Definitions

The following terms shall have the meaning given to them in the Framework Agreement: "Applicable Regulations", "Central Terms", "Operational Terms", "Regulatory Terms", "Services", "Service Agreement", "Service Form", "Service Module" and "Service Provider".

In this Regulatory Terms the following terms shall have the below meaning:

"MiFID II" together the Markets in Financial Instruments Directive (Directive 2014/65/EU), and the associated EU regulatory and implementing technical standards and implementing law and regulation in the EEA states.

"MiFIR" Markets in Financial Instruments Regulation (Regulation 600/2014)

"Order" any instructions received by the Service Provider from you or on your behalf or generated by the Service Provider on your behalf in relation to a Transaction.

"Transaction" a transaction resulting in the acquisition or disposal of a Financial Instrument (as defined in MiFID II).

### 3 Regulatory Information Document

The definitions and provisions contained in the "Regulatory Information Document", as amended, restated and/or supplemented from time to time (the "Regulatory Information Document"), are incorporated into these Regulatory Terms.

By accepting Services from the Service Provider you are deemed to have read and agreed the information contained within the Regulatory Information Document. A copy of the Regulatory Information Document may be viewed using the following link: https://www.bnymellon.com/RID (the "Information Website").

In the event of any inconsistency between the Regulatory Information Document and the Regulatory Terms, the Regulatory Terms will govern.

## 4 Market Transparency Requirements

MiFIR contains an enhanced market transparency regime, which requires the reporting and/or publication of Transactions to regulatory authorities and the public. Please refer to the Information Website for more detailed information on market transparency requirements.

In order for the Service Provider to comply with the Service Provider's Transaction Reporting Requirements or Market Transparency Requirements (as applicable) you agree to provide such information (and updates to such information as may have already been provided) relating to you and your agents, employees, underlying principals or others as the Service Provider may reasonably require in order (and in time) for the Service Provider to comply with any such requirements in respect of the Transactions or the services provided or expected to be provided to you ("Counterparty Data").

FMA2 September 2021 You hereby: (A) represent to the Service Provider that such Counterparty Data as you deliver is, at the time of delivery, true, accurate and complete in every material respect; (B) acknowledge and agree that the Service Provider may use the Counterparty Data to comply with its Transaction Reporting Requirements or Market Transparency Requirements, and may rely on the Counterparty Data without investigation, unless and until you inform the Service Provider otherwise; and (iii) undertake to provide the Service Provider, on reasonable notice, with any material changes or updates to the Counterparty Data.

You acknowledge and agree that the Service Provider may refuse to enter into, execute, transmit, deal in or otherwise arrange any Transaction where you have not provided such Counterparty Data (and waiver or procured the waiver of any confidentiality or data protection / privacy obligations in respect of such information) as the Service Provider may reasonably require:

- in order for the Service Provider to comply with any Transaction Reporting Requirements or Market Transparency Requirements in respect of such Transaction; or
- (b) where the Service Provider's non-receipt of such Counterparty Data (including, without limitation, an applicable legal entity identifier code) would mean that the Service Provider is prohibited by Applicable Regulation to enter into, execute, transmit, deal in or otherwise arrange (as the case may be) such Transaction (including, without limitation, pursuant to Article 13(2) of RTS 22)<sup>1</sup>.

## 5 Concerning the Service Provider

Permitted Transactions: The Service Provider or any of its affiliated companies or associates may act as agent for, provide banking, investment advisory, investment management and other services to, and generally engage in any kind of business with, others (including without limitation issuers of securities, money market instruments or other assets held for and on behalf of you) to the same extent as if the Service Provider was not a service provider under the Service Agreement. Nothing in the Service Agreement shall be deemed to restrict the right of the Service Provider or its affiliated companies or associates to perform such services for any other person or entity, and the performance of such services for others will not be deemed to violate or give rise to any duty or obligation to you not specifically undertaken by the Service Provider under the Service Agreement. If the Service Provider, or any of its affiliated companies or associates, provide services which are ancillary to the Service Provider's functions of custodian and banker or carry out other businesses and activities including but not limited to the matters contemplated above, the Service Provider or any of its affiliated companies or associates, as appropriate, may receive and retain any fee, commissions, spreads or other compensation in relation to such service, business or activity.

## 6 Confidentiality, Information Sharing and Data Privacy

6.1 Use of your Information: You agree and consent to the BNYM entities using your Information, in connection with acting either as the contracting entity under the Service Agreement, or under a transaction or as service provider or intermediary to you, or otherwise in connection with the performance of the Services and any other obligations under the Service Agreement or transactions and in connection with certain other activities related

<sup>&</sup>lt;sup>1</sup> Note: Art 20, 21, 26 MiFIR; RTS 1, 2 and 22; SUP 17A

thereto, including, without limitation, audit, accounting, tax, administration, risk management, credit, legal, compliance, operations, sales and marketing, relationship management, information technology, records and data storage, performance measurement, data aggregation and compilation and analysis of your Information (collectively, the "Activities").

- 6.2 Use of your Information: Notwithstanding anything to the contrary in the Service Agreement, each BNYM entity may, in connection with the Activities or for any other purpose permitted under the Service Agreement, collect, use, store and disclose, within and outside of the European Economic Area (including but not limited to the United States of America, the United Kingdom, Australia, Canada, Cayman Islands, Hong Kong, India, Japan, Republic of Korea and Singapore) your Information to: (a) other BNYM entities; and (b) third party service providers who are required to maintain the confidentiality of your Information. In addition, the Service Provider may aggregate your Information (other than your Personal Data) with other data collected and/or calculated by the Service Provider, and the Service Provider will own all such aggregated data, provided that the Service Provider shall not distribute the aggregated data in a format that identifies you or any particular individual after such aggregation.
- **6.3 Disclosure of your Information**: You acknowledge the Service Provider may disclose your Information to governmental, tax, regulatory, law enforcement and other authorities in relevant jurisdictions where the Service Provider operates and/or otherwise as required by law, rule or guideline (including tax reporting regulations) or requested by such authorities.

Your information (including identity) may be disclosed to any trade repositories, approved reporting mechanisms, approved publication arrangements and any other infrastructure as may be required or permitted by law or regulation in the relevant jurisdictions by the Service Provider directly or through a third-party service provider.

# 6.4 Notice of collection, use, storage and disclosure of your Information and Personal Data:

- (a) To the extent applicable, both parties acknowledge and agree that they are separate and independent Data Controllers in relation to the Personal Data processed pursuant to this Service Agreement. You represent you have lawful grounds (including consent where required), and that BNYM relies on your representation, for BNYM's collection, use, storage and disclosure of (a) your Information as set out in this clause 6, and (b) Personal Data, as set out in this clause 6 and BNYM's personal data privacy notice ("Privacy Notice").
- (b) BNYM's Privacy Notice is contained at https://www.bnymellon.com/us/en/data-privacy.jsp, receipt of which is hereby acknowledged by you, and shall apply to BNYM's use, storage and disclosure of your Information and Personal Data pursuant to this Agreement. You shall provide a copy of the Privacy Notice to the relevant Data Subjects, before or (at the latest) at the time when you or a person acting on your behalf provides BNYM with Personal Data.

You further acknowledge that BNYM may make amendments and additions to its Privacy Notice by posting a revised version at the abovementioned website link (or such other link as BNYM may advise from time to time).

## 6.5 International Transfer

(a) Where BNYM is required to transfer Personal Data to you, and such transfer is an international transfer which is not subject to an appropriate transfer mechanism

providing an adequate level of protection in compliance with applicable data protection laws, then the parties shall comply with UK Model Clauses incorporated into this Agreement by reference or such other data transfer clauses that are reasonably necessary to satisfy applicable data protection laws. For the UK Model Clauses BNYM, being located in the UK and disclosing the Personal Data, will be the data exporter, and you, being the party receiving the Personal Data, will be the data importer. The subject matter and duration of the Processing, the nature and purpose of the Processing, the type of Personal Data and categories of Data Subject relating to the international transfer are as set out in Schedule 2 herein.<sup>2</sup>

- (b) If, at any time, Data Protection Laws require that transfers from BNYM to you must be subject to specific additional safeguards (including but not limited to specific technical and organisational measures), the parties shall work together in good faith to implement such safeguards and ensure that any onward transfers of Personal Data in relation to BNYM is conducted with the benefit of such additional safeguards.
- **6.6 Confidentiality**: The Service Provider and you will at all times respect the confidentiality of the Service Agreement and any arrangements or agreements made or entered into in connection with the Service Agreement and each agrees not disclose to any other person any information acquired as a result of or pursuant to the Service Agreement, except:
  - (a) to its officers, employees, agents, delegates, legal and other advisers and auditors;
  - (b) in the case of the Service Provider, to any BNYM entities and their officers, employees, agents, delegates, legal and other advisers and auditors;
  - (c) with the consent of the other party; or
  - (d) if the disclosure is necessary in connection with any registration of any security or to comply with any applicable law, regulation, the rules of any relevant governmental body, regulatory authority, revenue authority, the rules of any securities or stock exchange or an order of a court, tribunal, regulatory authority or other party, or as otherwise agreed, and where you are making such a disclosure, the Service Provider is given prior notice of the disclosure unless such notification is prohibited by applicable law, governmental or regulatory authority, exchange, court or tribunal.
- **6.7** Definitions: For the purposes of this Clause 6, the following terms shall have the following meanings:
  - (a) "BNYM" and "BNYM entity" means The Bank of New York Mellon Corporation and/or each of its affiliated companies and associates (including each of their respective branches and representative offices, individually and/or collectively), acting either as the contracting entity under this Service Agreement or as BNYM or intermediary to The Bank of New York Mellon, London Branch, or otherwise in a relationship with you;
  - (b) "Data Controller" means an entity which, alone or jointly with others, determines the purposes and means of the Processing of Personal Data;
  - (c) "Data Protection Laws" means the GDPR (together with laws implementing or supplementing the GDPR, in each case as amended and superseded from time to time), and/or any laws relating to the processing of Personal Data enacted in the United Kingdom that substantially amend, replace or supersede the GDPR as a result of the United Kingdom leaving the European Union and/or any laws relating

٠

<sup>&</sup>lt;sup>2</sup> Where BNYM LB is the service provider, refer to the UK Model Clauses. Where BNYM, SA/NV is the service provider, refer to the EU Model Clauses.

to privacy and the protection and processing of Personal Data enacted in the relevant jurisdiction;

- (d) "Data Subject" means the individual to whom Personal Data relates;
- (e) "GDPR" means The General Data Protection Regulation (EU) 2016/679;
- (f) "your Information" means data regarding you and your affiliated companies and associates and either the Collateral Receiver's Account or the Collateral Provider's Account, as the context may require, excluding Personal Data;
- (g) "Personal Data" means a) in relation to BNYM, the type of personal data as set out in Schedule 2 hereto and b) in relation to you, personal data of your employees and representatives, your affiliated companies and associates, and where you are the trustee of a trust, individual beneficiaries and employees and representatives of beneficiaries of that trust;
- (h) "Processing" means any operation or set of operations which is performed on your Information, Personal Data or on sets of Personal Data, whether or not by automated means, such as collection, recording, organisation, structuring, storage, adaptation or alteration, retrieval, consultation, use, disclosure by transmission, dissemination or otherwise making available, alignment or combination, restriction, erasure or destruction; and
- (i) "UK Model Clauses" means the standard contractual clauses established pursuant to the UK Data Protection Act 2018 (or any successor standard contractual clauses that may be adopted from time to time).
- **6.8** This Clause 6 shall survive termination of the Service Agreement.

## 7 Provision of Information

You consent to the provision by the Service Provider of the following information, where not personally addressed to you, by means of a website (which may or may not be in addition to other means of communication):

- (a) general information about the Service Provider and its Services;
- (b) information about the nature and risks of certain financial instruments;
- (c) information concerning the safeguarding of financial instruments and holding of client money;
- (d) information on costs and associated charges;
- (e) information about the Service Provider's order handling and execution policies, conflicts of interest policies, complaints policies and other policies of the Service Provider; and
- (f) any other information required to be provided to you under applicable law or regulation.

The websites where such information is available can be viewed using these links:

https://www.bnymellon.com/RID

FMA2

https://connect.bnymellon.com/public\_content/marketingContent/GM/content/documents/gmw/GM\_FX\_disclosures.html

http://www.bnym.com/regulatoryinformation

or any such other website or websites as the Service Provider may notify to you from time to time.

## 8 Computer Services and electronic access

The Service Provider may, at your request, provide certain Computer Services to you and other persons on terms that are agreed by the Service Provider and you (and, in some cases, with the other person) under a separate written agreement. Where Computer Services are provided, you accept that Instructions may be validly transmitted by electronic means and expressly waive the right to bring any action or assert any defence on the sole ground that an Instruction was given by electronic means, via the internet or via non-encrypted electronic mail.

For the purposes of this Clause 8:

"Computer Services" means those electronic access services and electronic communication facilities (if any), which may also allow an Authorised Person to transmit Instructions through an electronic access system offered by the Service Provider as may be agreed from time to time pursuant to the Service Agreement.

#### 9 No Reliance - No Advice

Neither the Service Provider nor any of its affiliates shall owe you any duty to advise you upon the merits, suitability and appropriateness of any Services entered into or contemplated by you unless specifically agreed otherwise in writing.

The provision of any Service by the Service Provider pursuant to the Service Agreement will not, unless specifically agreed between you and the Service Provider in writing, give rise to any advisory, fiduciary or equitable duties on the part of the Service Provider or its affiliates other than those expressly provided for in the Service Agreement.

You agree that nothing contained in the Service Agreement shall create any fiduciary, trustee, agency, joint venture or partnership relationship between the Service Provider or any affiliate of the Service Provider, on the one hand, and you or any affiliate of yours on the other.

Without limitation to the generalities of the foregoing, neither the Service Provider nor any of its affiliates shall give you legal, regulatory, accounting, taxation, financial or any other advice in relation to any Services, trading decisions or products and you are solely responsible for seeking and obtaining your own advice and taking your own trading decisions.

## 10 Governing Law and Jurisdiction

10.1 Governing Law: The Service Agreement, these Regulatory Terms and any non-contractual obligations arising out of or in connection with them are governed by, and shall be construed in accordance with, the governing law specified in the applicable Service Module.

FMA2 September 2021 Jurisdiction: The courts specified in the applicable Service Module are to have exclusive jurisdiction to settle any disputes that may arise out of or in connection with the Service Agreement or these Regulatory Terms (including non-contractual disputes or claims) and accordingly any legal action or proceedings arising out of or in connection with the Service Agreement or these Regulatory Terms ("Proceedings") shall be brought in such courts. The parties irrevocably submit to the exclusive jurisdiction of such courts and waive any objection to Proceedings in such courts whether on the ground of venue or on the ground that the Proceedings have been brought in an inconvenient forum.

# Schedule 1 to the Regulatory Terms - BNYM Legal Entity Specific Terms

#### PART 1

Where the Service Provider is The Bank of New York Mellon, London Branch ("BNYM LB"), the following provisions shall apply:

- 1. The Bank of New York Mellon is authorised and regulated by the New York State Department of Financial Services and the Federal Reserve and authorised by the Prudential Regulation Authority. The Bank of New York Mellon, London Branch is subject to regulation by the Financial Conduct Authority and limited regulation by the Prudential Regulation Authority. Details about the extent of Bank of New York Mellon's regulation by the Prudential Regulation Authority are available from Bank of New York Mellon on request.
- 2. The Financial Services Compensation Scheme

The Service Provider is a participant in the UK Financial Services Compensation Scheme (the "Scheme"). The Scheme can pay compensation to claimants if an institution is unable to meet its financial obligations. The Scheme is only available to certain types of claimants and claims. Payments to eligible claimants under the Scheme will vary depending on the type of protected claim (e.g. deposits or investments) the claimants hold with respect to the relevant institution. Payments under the Scheme in respect of protected deposits are subject to a limit on the maximum compensation that any eligible depositor is entitled to claim. Payments under the Scheme in respect of Designated Investment Business (as defined under the Scheme) are subject to a maximum payment per eligible investor. Eligible investors with protected claims of different types can make claims up to the relevant compensation limit for each type of protected claim, but in respect of a particular type of protected claim the compensation limit applies to the combined amount of protected claims of that type across all the eligible claimant's dealings and accounts with the Service Provider. Further details of the Scheme, including deposit and investment limits, are available at the Scheme's official website at http://www.fscs.org.uk.

3. Clause 3 (*Regulatory Information Document*) shall be supplemented by the following provision where the Dealer Custody (English law) Service Module or Receiver-only Custody (English law) Service Module applies:

Terms used in this supplementary clause, shall have the meaning set out in Schedule 1 (*Definitions*) to the applicable Service Module.

There are limited circumstances in which the Service Provider may hold certain sums as client money for the benefit of you in accordance with the Client Money Rules. These circumstances are limited to the requirements under the Client Assets Rules pursuant to which the Service Provider may be required to segregate certain sums from the Service Provider's own funds as client money in certain cases where the Service Provider has identified a shortfall in the number of client securities held by or for it. Such segregation will continue until such time as the relevant shortfall has been resolved at which point the Service Provider will re-appropriate such money. Such client money amount will be held in accordance with the Client Money Rules on behalf of you, to the extent that you are affected by the relevant shortfall. In the absence of a Service Provider failure, such segregation does not create a cash entitlement of you against the Service Provider. In the event of the failure of the Service Provider, the Client Money Distribution and Transfer Rules will apply to any such money held as client money by the Service Provider. Client money will be held with a

third party bank or banks. The Service Provider does not accept any liability for any default or delay in the distribution of client money in the event of the failure of a bank holding client money on its behalf. Should the failure of a bank with which the Service Provider holds any client money occur at the same time as the failure of the Service Provider, you may share in any shortfall of client money on a pro rata basis. The Service Provider may from time to time notify you of other circumstances in which it may hold client money in accordance with the Client Money Rules. The Service Provider shall not pay any interest earned on any client money to you. In the limited circumstances described in this paragraph in which the Service Provider holds certain sums as client money for the benefit of you in accordance with the Client Money Rules, our standard practice would be for the Service Provider to open accounts with third party banks within the United Kingdom but there may be reasons (including, but not limited to diversification requirements) where the Service Provider may arrange for such money to be held outside the United Kingdom. Such money may be held in accounts with a third party bank or banks in a state which is not an EEA Member State and, in such case, the relevant accounts will be subject to the laws of that state and as a result such money may be treated in a different manner from that which would apply if such money were held by a third party bank(s) located in the EEA.

In addition, you acknowledge that any Cash Account the Service Provider opens for you in accordance with the terms of the Service Agreement are not payment accounts as defined by the Payments Services Regulations 2017 and must not be used to make or receive payments which are not directly related to the servicing of your assets and investment services. If you request the Service Provider to make or receive payments of this kind from or to your Cash Account, or the Service Provider reasonably believes that you are using your Cash Accounts in this way, the Service Provider has the right to refuse to execute your Instructions in relation to those Cash Accounts and may require you to open a payment account with the Service Provider, or a third party, for these purposes. The Service Provider is under no obligation to monitor the use of your Cash Accounts or the purposes of any Instructions the Service Provider receives in connection with the Cash Accounts, although the Service Provider may do so from time to time.

Unless the context otherwise requires, any reference to EU legislation, regulatory requirement, or guidance (including any concepts as defined therein) should be read as a reference to that EU legislation, regulatory requirement or guidance (including any concepts as defined therein) as it forms part of UK domestic law pursuant to the European Union (Withdrawal) Act 2018 (as amended) (the EUWA) or as otherwise adopted under, or given effect to in, UK legislation or the UK regulatory regime (UK Onshored Legislation, Regulatory Requirement, or Guidance) and any references to EU competent authorities should be read as references to the relevant UK competent authority. All references to legislation, regulatory requirements or guidance in this clause refer to the relevant legislation, regulatory requirements or guidance as amended from time to time.

#### PART 2

Where the Service Provider is The Bank of New York Mellon, S.A./N.V., the following provisions shall apply:

1. The Bank of New York Mellon SA/NV is regulated and supervised by the European Central Bank together with the National Bank of Belgium, as a significant credit institution under the

European Single Supervisory Mechanism. Details about the extent of BNYM's regulation by such authorities are available from BNYM on request.

- 2. The Bank of New York Mellon SA/NV is a member of the Belgian deposit guarantee scheme. However, as it is licensed as an assimilated settlement institution in accordance with the Belgian Law of 22 February 1998, deposits held at The Bank of New York Mellon SA/NV are not eligible for protection. More information is available at http://fondsdegarantie.belgium.be.
- 3. Pursuant to the Belgian rules implementing Directive (EU) 2018/843 of 30 May 2018 (5th Anti-Money Laundering Directive), the Service Provider is legally required to disclose certain information regarding you to a central registry in Belgium. Further information on the disclosure obligations under such regime are made available on the following website <a href="https://www.bnymellon.com/emea/en/regulatory-resources.jsp#5AMLD">https://www.bnymellon.com/emea/en/regulatory-resources.jsp#5AMLD</a> and you agree to review such information (and any future amendments thereto) when published on such website and consent to the disclosure of the information as set out in therein.
- 4. Clause 3 (*Regulatory Information Document*) shall be supplemented by the following provision where the Dealer Custody (Belgian law) Service Module or Receiver-only Custody (Belgian law) Service Module applies:

Terms used in this supplementary clause, shall have the meaning set out in Schedule 1 (*Definitions*) to the applicable Service Module.

You acknowledge that any Cash Account the Service Provider opens for you in accordance with the terms of the Service Agreement are not payment accounts as defined by the European Payments Services Directive 2015/2366 of 25 November 2015 and its Belgian implementing measures and must not be used to make or receive payments which are not directly related to the servicing of your assets and investment services. If you request the Service Provider to make or receive payments of this kind from or to your Cash Account, or the Service Provider reasonably believes that you are using your Cash Accounts in this way, the Service Provider has the right to refuse to execute your Instructions in relation to those Cash Accounts and may require you to open a payment account with the Service Provider, or a third party, for these purposes. The Service Provider is under no obligation to monitor the use of your Cash Accounts or the purposes of any Instructions the Service Provider receives in connection with the Cash Accounts, although the Service Provider may do so from time to time.

## Schedule 2 to the Regulatory Terms - Personal Data Transfer

## Data subjects

BNYM employees

### Purposes of the transfer(s)

The transfer is made for the following purposes:

On-boarding clients

### Categories of data

The personal data transferred concern the following categories of data (please specify): Certificates of incumbency (BNYM-employees)

## Recipients

The personal data transferred may be disclosed only to the following recipients or categories of recipients:

BNYM Clients (BNYM employees' Personal Data)

#### Sensitive data (if appropriate)

The personal data transferred concern the following categories of sensitive data: None

## Additional useful information (storage limits and other relevant information):

None

## Contact points for data protection enquiries:

Relationship Manager

# The Bank of New York Mellon Security (English Law) Service Module



# **Table of Contents**

Cont	tents P	age
1	Introduction	2
2	Definitions and interpretation	2
3	The Secured Obligations	2
4	Security Interests	2
5	Advances and right of retention	3
6	Restrictions on other Security	4
7	Perfection and Release of the Service Provider's Security	4
8	Undertakings by the Company	5
9	Enforcement – General Provisions	6
10	Power of Sale	6
11	Appropriation	7
12	Appointment of Receiver	7
13	Power of Attorney	8
14	Protection of Purchasers	8
15	Consolidation of Accounts and Set-Off	9
16	Currency	9
17	Suspense Account	9
18	The Service Provider's Rights	9
19	Constitution	10
20	Governing Law and Jurisdiction	10
Sche	dule 1 to the Security (English law) Service Module - Definitions	11
	dule 2 to the Security (English law) Service Module - Additional terms where registration is red in Australia	
	dule 3 to the Security (English law) Service Module - Additional terms where the Collatera	

#### 1 Introduction

Pursuant to the terms of a framework agreement executed and (to the extent applicable) delivered (a "Framework Agreement"), a Service Agreement shall take effect between you and the relevant Service Provider in respect of each of the Service Modules elected to apply to you as set out in the Service Form.

## 2 Definitions and interpretation

The following terms shall have the meaning given to them in the Framework Agreement: "Applicable Regulations", "Operational Terms", "Regulatory Terms", "Service Agreement", "Service Form", "Service Module" and "Service Provider".

Wherever used in the Service Agreement, the terms set out in Schedule 1 to this Service Module (*Definitions*) shall, unless the context requires otherwise, have the meanings set out in that Schedule.

## 3 The Secured Obligations

- 3.1 The Company shall:
  - 3.1.1 repay to the Service Provider any Advance on demand; and
  - 3.1.2 pay or discharge each of the other Collateral Secured Obligations at the time and in the manner provided for in the relevant document.
- 3.2 If any amount demanded under Clause 3.1.1 or payable under Clause 3.1.2 is not paid immediately upon demand or on its due date, as applicable, interest shall accrue on that amount at the rate agreed between the Company and the Service Provider or, in the event of no such rate having been agreed, at a rate determined in accordance with the Service Provider's usual practice (the rate so agreed or determined to apply after as well as before any judgment), such interest to be paid by the Company to the Service Provider upon interest payment dates selected by the Service Provider in accordance with its usual practice and to be compounded with rests on such payment dates in the event of its not being duly and punctually paid.

## 4 Security Interests

- **4.1** Each of the security interests constituted by this Clause 4 is made with full title guarantee.
- **4.2** The Company charges by way of fixed charge, and to the extent applicable pledges, in favour of the Service Provider as security for the payment and discharge of the Collateral Secured Obligations all of its rights in:
  - 4.2.1 any Cash Account, any Deposit and any indebtedness represented by any Deposit; and
  - 4.2.2 any Securities Account and any Relevant Securities.
- 4.3 As security for the payment and discharge of the Collateral Secured Obligations, the Company assigns absolutely to the Service Provider, subject to a proviso for re-assignment on redemption in accordance with Clause 7.4, all of its rights in the Collateral Management Agreements.

- 4.4 The Company charges by way of fixed charge, and to the extent applicable pledges, in favour of the Service Provider as security for the payment and discharge of the Secured Obligations all of its rights in:
  - 4.4.1 any Cash Account, any Deposit and any indebtedness represented by any Deposit; and
  - 4.4.2 any Securities Account and any Relevant Securities.
- 4.5 As security for the payment and discharge of the Secured Obligations, the Company assigns absolutely to the Service Provider, subject to a proviso for re-assignment on redemption in accordance with Clause 7.4, all of its rights in the Collateral Management Agreements.
- 4.6 The security interests constituted by Clauses 4.2 and 4.3 respectively as security for the payment and discharge of the Collateral Secured Obligations shall rank in priority to those constituted by Clauses 4.4 and 4.5 respectively as security for the payment and discharge of the Secured Obligations.
- 4.7 This Security shall:
  - 4.7.1 constitute continuing security interests in favour of the Service Provider and shall be in addition to and independent of every bill, note, guarantee, mortgage or other security interest which the Service Provider may at any time hold for any of the Secured Obligations and it is hereby declared that no prior security interest held by the Service Provider over any Charged Assets shall merge in this Security; and
  - 4.7.2 remain in full force and effect as a continuing security until discharged by the Service Provider.
- **4.8** The Service Agreement shall constitute notice to the Service Provider of the security interests constituted by this Clause 4.

# 5 Advances and right of retention

- 5.1 The Service Provider may make Advances available to the Company pursuant to or in connection with the Collateral Management Agreements and/or the provision of the Collateral Management Services. Notwithstanding any other provision of the Collateral Management Agreements:
  - 5.1.1 the Service Provider is under no obligation to make any Advance available to the Company, so that the Service Provider may in its discretion decide whether or not to make any Advance requested by or on behalf of the Company and, if so, in what amount; and
  - 5.1.2 any Advance shall be repayable by the Company to the Service Provider on demand.
- 5.2 The Service Provider intends to monitor the extent to which the Charged Assets constitute what it considers from time to time to be a sufficient level of security for the Service Provider in respect of the Collateral Secured Obligations. The Service Provider may at any time refuse to effect or permit a transfer of any Charged Assets to or at the order of the Company or otherwise in connection with the provision of the Collateral Management Services, including any transfer which it would otherwise have been minded or under an obligation to effect or permit, to the extent that the Service Provider determines in its discretion that retention of those Charged Assets in the Accounts or otherwise under the control of the Service Provider

is necessary or desirable to maintain that sufficient level of security for the Collateral Secured Obligations. This right of retention:

- 5.2.1 shall apply notwithstanding any other provision of the Collateral Management Agreements; and
- 5.2.2 shall be without prejudice to any other lien or right of retention which the Service Provider may have by law or contract in respect of some or all of the Collateral Secured Obligations or some or all of any other Secured Obligations.
- 5.3 The Service Provider may from time to time notify the Company of its requirements regarding the nature and value of Charged Assets which the Service Provider might in connection with possible Advances accept as sufficient security in respect of the Collateral Secured Obligations, including specifying margin or haircut requirements by way of over-collateralisation. Any such notification shall be indicative only and shall neither create an obligation on the Service Provider to make any Advance available nor limit the rights of the Service Provider under the Service Agreement, including without limitation under Clauses 5.2 and 7.2.
- 5.4 The Company undertakes to the Service Provider and BNYM S.A./N.V. that it shall not at any time exercise or purport to exercise any rights which it might have to give any instructions in relation to Relevant Securities to any person (including BNYM S.A./N.V. or any other Subcustodian, Depository or delegate) other than the Service Provider.

## 6 Restrictions on other Security

- 6.1 The Company shall not at any time without the prior written consent or agreement of the Service Provider create, extend or permit to subsist any mortgage or other fixed security, floating charge, pledge, hypothecation or lien or other security interest of any kind over the Charged Assets, whether in any such case ranking in priority to or pari passu with or after this Security, other than:
  - 6.1.1 any lien arising by operation of applicable law; and
  - 6.1.2 any security interest in favour of the Service Provider or any Sub-custodian, Depository or delegate constituted by or expressly envisaged in the Collateral Management Agreements.

# 7 Perfection and Release of the Service Provider's Security

- 7.1 The Company shall promptly upon notice from the Service Provider execute all documents and do all things (including the delivery, assignment or other transfer or payment of the Charged Assets to the Service Provider) that the Service Provider may at any time reasonably specify for the purpose of: (a) exercising any of its rights under or in connection with the Service Agreement; or (b) securing and perfecting its security over or title to the Charged Assets; or (c) enabling the Service Provider to vest the Charged Assets in its name or in the name(s) of its nominee(s), agent or any purchaser.
- 7.2 Without prejudice to Clause 7.1, the Company shall, at any time the Service Provider requests and at the Company's cost, execute in favour of the Service Provider, or as it may direct, such further security interests as in each such case the Service Provider shall stipulate over the Company's rights in any property or other assets (such assets to become Charged Assets) of whatever nature or tenure and wherever situate for the purpose of more effectively providing sufficient security to the Service Provider for the payment or discharge

of the Collateral Secured Obligations and, if the Service Provider so stipulates, the other Secured Obligations. Without prejudice to the generality of the above, such other security interests shall be of such nature and in such form as shall be prepared on behalf of the Service Provider and may contain provisions such as are herein contained or provisions to the same effect and/or such other provisions of whatever kind as the Service Provider shall consider requisite for the improvement or perfection of this Security. The Company agrees to deliver to the Service Provider when called for by it such additional property and other assets of a kind and of a market value satisfactory to the Service Provider, so that there will, at all times, be with the Service Provider a margin of security for the payment of all Collateral Secured Obligations and, if the Service Provider so stipulates, other Secured Obligations which shall be satisfactory to it.

- 7.3 The Service Provider may register, and give any notice in connection with, this Security at the Company's expense. The Company consents to any such registration or notification. The Company must provide the Service Provider with any information it requires for the purposes of completing such registration or notification and do all other things, and enable and facilitate the Service Provider to do all things, as are necessary or desirable to effect such registration or notification including giving consent to such registration or notification where required.
- 7.4 This Clause 7.4 applies at any time following the date (the "Discharge Date") on which:
  - 7.4.1 all of the Secured Obligations have been unconditionally and irrevocably paid or discharged in full to the satisfaction of the Service Provider. For this purpose, if the Service Provider considers that an amount paid to it is capable of being avoided or otherwise set aside on an insolvency of the payer or otherwise, then it will not be considered to have been irrevocably paid; and
  - 7.4.2 the Collateral Management Agreements have been terminated and the Service Provider is satisfied that it has ceased to have any commitment, obligation or other liability (whether actual or contingent) under or in respect of them.

Following the Discharge Date, if requested by the Company, the Service Provider shall as soon as reasonably practicable release and discharge this Security and re-assign the assets assigned to the Service Provider under the Service Agreement to the Company without recourse, representation or warranty and subject to the rights of any person having prior rights over those assets.

### 8 Undertakings by the Company

- **8.1** The Company hereby undertakes with the Service Provider that the Company will at all times while this Security subsists:
  - 8.1.1 provide the Service Provider, its employees, professional advisers and agents with all such information regarding the Company's business and affairs as the Service Provider may from time to time require; and
  - 8.1.2 indemnify the Service Provider (and as a separate covenant any Receiver or Receivers appointed by it) against all existing and future rents, taxes, duties, fees, renewals fees, charges, assessments, impositions and outgoings whatsoever (whether imposed by deed or statute or otherwise and whether in the nature of capital or revenue and even though of a wholly novel character) which now or at any

time during the continuance of this Security are payable in respect of the Charged Assets or by the owner or occupier thereof.

- 8.2 If any such sums as are referred to in Clause 8.1.2 shall be paid by the Service Provider (or any such Receiver or Receivers), the same shall be repaid by the Company on demand with interest as provided in Clause 3.2 from the time or respective times of the same having been paid.
- 8.3 The Company hereby undertakes to and agrees with the Service Provider and BNYM S.A./N.V. that the Company will not take any action in relation to the Relevant Securities or any other Charged Assets which is inconsistent with this Security and the other rights granted to the Service Provider under Clause 4 (Security Interests) and the other provisions of the Service Agreement and the Company will take all appropriate action which may be required to assure the priority of this Security and those other rights granted in favour of the Service Provider under the Service Agreement.

### 9 Enforcement – General Provisions

- 9.1 This Security shall become enforceable if any of the following events shall occur:
  - 9.1.1 the Company fails to repay any Advance immediately on demand by the Service Provider:
  - 9.1.2 the Company fails to pay or discharge any other Secured Obligation on the due date for its payment or discharge; or
  - 9.1.3 an Insolvency Event occurs in relation to the Company.
- 9.2 At any time after this Security has become enforceable, the Service Provider may enforce this Security, and its rights under the Service Agreement, in the manner and on the terms it thinks fit. In particular, it may without further notice exercise in relation to the Charged Assets:
  - 9.2.1 the power of sale and all other powers conferred on mortgagees by the LPA or otherwise by law, in each case as extended or otherwise amended by the Service Agreement;
  - **9.2.2** to the extent that Clause 11 (*Appropriation*) applies, the power to appropriate the Charged Assets in accordance with Clause 11.2;
  - 9.2.3 the power to exercise any rights which the Company might have to give any instructions in relation to Relevant Securities to any person (including any Subcustodian, Depository or delegate) other than the Service Provider;
  - 9.2.4 the right to exercise and do in relation to the Charged Assets all the rights and things which the Service Provider would be capable of exercising or doing if it were the absolute beneficial owner of the Charged Assets; and
  - 9.2.5 (whether or not it has appointed a Receiver) any or all of the rights which are conferred by the Service Agreement (whether expressly or by implication) on a Receiver.

#### 10 Power of Sale

10.1 At any time after this Security has become enforceable, the Service Provider shall be entitled, without prior notice to the Company or prior authorisation from any court, to sell, transfer or otherwise dispose of the Charged Assets on any terms and for any consideration

(which may include cash, securities or obligations and may be payable in a lump sum or instalments) as the Service Provider may think fit. The Service Provider shall be entitled to apply the proceeds of that sale or other disposal in paying the costs of that sale or disposal and in or towards the discharge of, first, the Collateral Secured Obligations and, second, the other Secured Obligations.

10.2 The power of sale or other disposal in Clause 10.1 shall operate as a variation and extension of the statutory power of sale under Section 101 of the LPA. The restrictions contained in Section 93 and 103 of the LPA shall not apply to the Service Agreement or to any exercise by the Service Provider of its right to consolidate mortgages or its power of sale.

## 11 Appropriation

- 11.1 This Clause 11 applies to the extent the Charged Assets constitute "financial collateral" and the Service Agreement constitutes or forms part of a "financial collateral arrangement" (within the meaning of the Financial Collateral Arrangements (No. 2) Regulations 2003).
- 11.2 If and to the extent that this Clause 11 applies by virtue of the security interests constituted by Clauses 4.2 and 4.3, the Service Provider may appropriate the Charged Assets in or towards discharge of the Collateral Secured Obligations. The Company agrees and acknowledges that by virtue of the provisions of the Service Agreement, including Clauses 5.2 and 5.3, it does so apply and undertakes not to argue to the contrary in any action or proceedings.
- 11.3 If and to the extent that this Clause 11 applies in relation to the security interests constituted by Clauses 4.4 and 4.5, the Service Provider may appropriate the Charged Assets in or towards discharge of, first, the Collateral Secured Obligations and, second, the other Secured Obligations.
- 11.4 If the Service Provider does appropriate the Charged Assets in or towards discharge of any Secured Obligations, then the Service Provider shall for these purposes value:
  - 11.4.1 any relevant Cash Account or other bank account at the amount standing to the credit of that account, together with any accrued interest not credited to that account, at the time of the appropriation; and
  - 11.4.2 any other relevant Charged Asset by reference to an independent valuation or other procedure determined by the Service Provider, acting reasonably, at the time of the appropriation.

## 12 Appointment of Receiver

- 12.1 At any time after this Security has become enforceable, the Service Provider may by writing (acting through an authorised officer of the Service Provider) without notice to the Company appoint one or more persons to be a receiver, receiver and manager or administrative receiver (any person so appointed a "Receiver") of the Charged Assets. Each such person shall be (a) entitled to act individually as well as jointly and (b) for all purposes deemed to be the agent of the Company, which shall be solely responsible for the Receiver's acts and defaults and for the payment of his remuneration.
- 12.2 In addition to the powers of the Service Provider conferred by Clause 10 (*Power of Sale*), each Receiver shall have, in relation to the Charged Assets in respect of which he was appointed, all the powers (a) conferred by the LPA on a Receiver appointed under the LPA,

- (b) of an administrative receiver as set out in Schedule 1 to the Insolvency Act 1986 (whether or not the Receiver is an administrative receiver) and (c) (if the Receiver is an administrative receiver) all the other powers exercisable by an administrative receiver in relation to the Company by virtue of the Insolvency Act 1986.
- 12.3 Only monies actually paid by the Receiver to the Service Provider in satisfaction or discharge of the Secured Obligations shall be capable of being applied by the Service Provider in satisfaction thereof.

## 13 Power of Attorney

- **13.1** The Company irrevocably appoints the following, namely:
  - 13.1.1 the Service Provider;
  - 13.1.2 each and every person to whom the Service Provider shall from time to time have delegated the exercise of the power of attorney conferred by this Clause 13.1; and
  - 13.1.3 any Receiver appointed hereunder and for the time being holding office as such,

jointly and also severally to be the attorney or attorneys of the Company and in its name and otherwise on its behalf and as its act and deed to sign, seal, execute, deliver, perfect and do all deeds, instruments, acts and things which may be required (or which the Service Provider or any Receiver appointed hereunder shall consider requisite) for carrying out any obligation imposed on the Company by or pursuant to the Service Agreement (including but not limited to the obligations of the Company under Clause 7.2), for carrying any sale, lease or other dealing by the Service Provider or such Receiver into effect, for conveying or transferring any legal estate or other interest in land or other assets or otherwise howsoever, for getting in the Charged Assets, and generally for enabling the Service Provider and the Receiver to exercise the respective powers conferred on them by or pursuant to the Service Agreement or by applicable law. The Service Provider shall have full power to delegate the power conferred on it by this Clause 13.1, but no such delegation shall preclude the subsequent exercise of such power by the Service Provider itself or preclude the Service Provider from making a subsequent delegation thereof to some other person. Any such delegation may be revoked by the Service Provider at any time.

- 13.2 The Company shall ratify and confirm all transactions entered into by the Service Provider or such Receiver or delegate of the Service Provider in the exercise or purported exercise of the Service Provider's or such Receiver's respective powers and all transactions entered into, documents executed and things done by the Service Provider or such Receiver or delegate by virtue of the power of attorney given by Clause 13.1.
- 13.3 The power of attorney hereby granted is as regards the Service Provider, its delegates and any such Receiver (and as the Company hereby acknowledges) granted irrevocably and for value as part of the security constituted by the Service Agreement to secure proprietary interests of and the performance of obligations owed to the respective donees within the meaning of the Powers of Attorney Act 1971.

## 14 Protection of Purchasers

No purchaser or other person dealing with the Service Provider or its delegate or any Receiver appointed hereunder shall be bound to see or inquire whether the right of the Service Provider or such Receiver to exercise any of its or his powers has arisen or become exercisable or be concerned with notice to the contrary, or be concerned to see whether any such delegation by the Service Provider shall have lapsed for any reason or been revoked.

#### 15 Consolidation of Accounts and Set-Off

In addition to its other rights (including security interests) under the Service Agreement and by operation of applicable law, the Service Provider shall have the right at any time and without notice to the Company (as well before as after making any demand hereunder) to combine or consolidate all or any of the Deposits and set-off or transfer any sum or sums standing to the credit of any one or more Cash Accounts in or towards satisfaction of any of the Secured Obligations. This Clause 15 applies despite any other agreement between the Company and the Service Provider.

## 16 Currency

For the purpose of or pending the discharge of any of the Secured Obligations the Service Provider may convert any monies received recovered or realised or subject to application by the Service Provider under the Service Agreement (including the proceeds of any previous conversion under this Clause 16) from their existing currency of denomination into such other currency of denomination as the Service Provider may think fit and any such conversion shall be effected at the Service Provider's then prevailing spot selling rate of exchange for such other currency against the existing currency. Each previous reference in this Clause 16 to a currency extends to any funds of that currency and for the avoidance of doubt funds of one currency may be converted into funds of a different currency.

## 17 Suspense Account

All monies received, recovered or realised by the Service Provider under the Service Agreement (including the proceeds of any conversion of currency) may in the discretion of the Service Provider be credited to any suspense or impersonal account and may be held in such account for so long as the Service Provider may think fit (with interest accruing thereon at such rate, if any, as the Service Provider may deem fit) pending their application from time to time (as the Service Provider shall be entitled to do in its discretion) in or towards the discharge of any of the Secured Obligations.

## 18 The Service Provider's Rights

- 18.1 The Service Provider may at any time or times without discharging or in any way affecting this Security or any right of the Service Provider in respect of this Security grant to the Company time or indulgence or abstain from asserting, calling, exercising or enforcing any remedies, securities, guarantees or other rights which it may now or in the future have from or against the Company.
- 18.2 Any receipt release or discharge of this Security or of any liability arising under the Service Agreement shall not release or discharge the Company from any liability to the Service Provider or any BNYM Affiliate for the same or any other monies which may exist independently of the Service Agreement.
- 18.3 The Service Provider may in its discretion grant time or other indulgence, or make any other arrangement, variation or release with, any person or persons not party hereto (whether or not such person or persons are jointly liable with the Company) in respect of any of the Secured Obligations or of any other security interest therefor or guarantee in respect thereof

without prejudice either to this Security or to the liability of the Company for the Secured Obligations or the exercise by the Service Provider of any rights, remedies and privileges conferred upon it by the Service Agreement.

- 18.4 The rights, powers and remedies provided in the Service Agreement are cumulative and are not, nor are they to be construed as, exclusive of any rights, powers or remedies provided by applicable law.
- 18.5 No failure on the part of the Service Provider or BNYM S.A./N.V. to exercise, or delay on its part in exercising, any of the rights, powers and remedies provided by the Service Agreement or by applicable law (collectively, "Service Provider's Rights") shall operate as a waiver thereof, nor shall any single or partial waiver of any of the Service Provider's Rights preclude any further or other exercise of that one of the Service Provider's Rights concerned or the exercise of any other of the Service Provider's Rights.
- All the costs, charges and expenses incurred by the Service Provider or any Receiver or delegate in relation to the Service Agreement or the Secured Obligations (including the costs, charges and expenses incurred in the carrying of the Service Agreement into effect or in the exercise of any of the rights, remedies and powers conferred on the Service Provider hereby or in the perfection or enforcement of this Security or in the perfection or enforcement of any other security interest for or guarantee in respect of the Secured Obligations) shall be reimbursed by the Company to the Service Provider on demand on a full indemnity basis. Until so reimbursed the same shall carry interest as mentioned in Clause 3.2 accruing from the date of the same being incurred by the Service Provider.

### 19 Constitution

The Company hereby certifies that its creation by the Service Agreement of security interests in favour of the Service Provider does not contravene any of the provisions of its constitution, its other constitutive documents or its other governing instrument.

## 20 Governing Law and Jurisdiction

- **20.1 Governing Law**: The Service Agreement and this Service Module and any non-contractual obligations arising out of or in connection with them are governed by, and shall be construed in accordance with, English law.
- 20.2 Jurisdiction: The courts of England are to have exclusive jurisdiction to settle any disputes that may arise out of or in connection with the Service Agreement and this Service Module (including non-contractual disputes or claims) and accordingly any legal action or proceedings arising out of or in connection with the Service Agreement and this Service Module ("Proceedings") shall be brought in such courts. The parties irrevocably submit to the exclusive jurisdiction of such courts and waive any objection to Proceedings in such courts whether on the ground of venue or on the ground that the Proceedings have been brought in an inconvenient forum.

# Schedule 1 to the Security (English law) Service Module - Definitions

## (1) In this Security agreement:

"Account" means any Cash Account or Securities Account;

"Advance" means any advance, overdraft or other form of credit accommodation made available from time to time by the Service Provider to the Company pursuant to or in connection with the Collateral Management Agreements and/or the Collateral Management Services:

**"BNYM Affiliate"** means any entity in which The Bank of New York Mellon Corporation (a Delaware corporation controls (directly or indirectly) an interest of no less than 30% in the voting stock or interests of such entity;

"BNYM S.A./N.V." means The Bank of New York Mellon S.A./N.V., a company organised under the laws of Belgium;

"Cash Account" means any cash account opened or maintained by the Service Provider pursuant to or in connection with the Collateral Management Agreements and/or the Collateral Management Services;

"Charged Assets" means the assets from time to time the subject of this Security, "Charged Asset" means any of the Charged Assets and any reference to one or more of the Charged Assets includes all or any part of it or of each of them;

## "Collateral Management Agreements" means:

- (a) the Custody Agreement;
- (b) the Service Agreement; and
- (c) any other agreement or document from time to time in force between the Company and the Service Provider relating to the provision of the Collateral Management Services,

in each case as supplemented, otherwise varied, novated or replaced from time to time (however fundamental the variation, novation or replacement and whether or not more onerous from the Company's perspective);

"Collateral Management Services" means the collateral management services from time to time provided to the Company by the Service Provider;

"Collateral Secured Obligations" means all Advances and other monies from time to time due or owing, and all obligations and other actual or contingent liabilities from time to time incurred, by the Company to the Service Provider under or in connection with the Collateral Management Agreements and/or the Collateral Management Services;

"Company" means you, as defined in the Framework Agreement;

"Custody Agreement" means the Service Agreement that incorporates the Dealer Custody (English law) Service Module dated on or about the date of the Service Agreement between the Company and the Service Provider;

"Data Controller" shall have the same meaning as in the Data Protection Laws;

"Data Protection Laws" means all applicable laws relating to privacy and the protection and processing of Personal Data in the relevant jurisdiction;

"Data Subject" shall have the same meaning as in the Data Protection Laws;

"Deposit" means any credit balance from time to time on any Cash Account;

"Depository" means BNY Mellon CSD S.A./N.V., Euroclear, Clearstream (Luxembourg), the Federal Reserve/Treasury Book-Entry System, the Depository Trust and Clearing Corporation and any other securities depository, clearing agency, book-entry system or other entity that provides handling, clearing, or safekeeping services in which the Service Provider or any of its Sub-custodians participates as a customer or member;

"Disclosing Party" means the party disclosing Personal Data;

"Insolvency Event" means any of the following in relation to the Company:

- (a) it becomes insolvent, is unable or fails or admits its inability to pay its debts as they fall due or ceases to carry on all or a material part of its business;
- (b) any step is taken with a view to a moratorium or a composition, assignment or similar arrangement with any of its creditors;
- (c) a meeting of its shareholders, directors or other officers is convened for the purpose of considering any resolution for, to petition for or to file documents with a court or any registrar for, its winding-up, administration or dissolution or any such resolution is passed;
- (d) any person presents a petition, or files documents with a court or any registrar, for its winding-up, administration, dissolution or reorganisation (by way of voluntary arrangement, scheme of arrangement or otherwise);
- (e) any security interest is enforced over all or substantially all of its assets;
- (f) an order for its winding-up, administration or dissolution is made;
- (g) any liquidator, trustee in bankruptcy, judicial custodian, compulsory manager, receiver, administrative receiver, administrator or similar officer is appointed in respect of it or any of its assets;
- (h) its shareholders, directors or other officers or the Company itself request(s) or apply/ies to court for the appointment of, or give(s) notice of their intention to appoint, a liquidator, trustee in bankruptcy, judicial custodian, compulsory manager, receiver, administrative receiver, administrator or similar officer; or
- (i) any analogous step or procedure is taken in any applicable jurisdiction;

"LPA" means the Law of Property Act 1925;

"Permitted Purpose" means the purposes set out in (i) this Agreement and (ii) the Receiving Party's privacy notice (as notified by the Receiving Party);

"Personal Data" shall have the same meaning as in the Data Protection Laws;

"Process" shall have the same meaning as in the Data Protection Laws;

"Receiver" has the meaning given to it in Clause 12.1;

"Receiving Party" means the party receiving the Personal Data;

"Relevant Securities" means any securities from time to time held or recorded in a Securities Account, including any such securities held by the Service Provider or to its order, on its behalf, for its account or otherwise under its control or direction;

"Secured Obligations" means all monies from time to time due or owing, and all obligations and other actual or contingent liabilities from time to time incurred, by the Company to the Service Provider or any BNYM Affiliate on any account, including the Collateral Secured Obligations;

"Securities Account" means any securities account opened or maintained by the Service Provider pursuant to or in connection with the Collateral Management Agreements and/or the Collateral Management Services;

"security" unless (as in the case of the expressions "security interest" and "this Security") the context requires otherwise means any bond, debenture, note, stock, share, warrant, unit or other debt or equity security of any kind (including instruments representing the right to receive, purchase or subscribe for any such security) and any reference to any security shall include:

- (j) any dividend, interest or other payment or distribution paid or payable in respect of it;
- (k) any right, money or property accruing or offered at any time in respect of it by way of redemption, exchange, bonus or otherwise; and
- (I) any right against any nominee or other trustee, fiduciary, custodian or Depository with respect to it;

"security interest" means an assignment by way of security (including an absolute assignment subject to a proviso for re-assignment on redemption) or other mortgage, charge, pledge, lien or other security interest securing the obligation of any person or any other agreement or arrangement having a similar effect;

"Sub-custodian" has the meaning given to it in the Custody Agreement; and

"this Security" means the security interests constituted by or pursuant to the Service Agreement.

- (2) Section 61 of the LPA shall govern the construction hereof and references to:
  - (a) a "law" include common or customary law and any constitution, decree, judgment, legislation, order, ordinance, regulation, statute, treaty or other legislative measure, in each case of any jurisdiction whatever (and "lawful" and "unlawful" shall be construed accordingly);
  - (b) the expression "applicable law" shall include English law and any other laws applicable to the Service Provider, the Company and the Charged Assets in jurisdictions outside of England;
  - (c) the word "assets" includes present and future properties, revenues, rights and other assets of every description;
  - (d) the word "including" means "including without limitation" (and related words shall be construed accordingly);
  - (e) "rights" in any security, document or other asset shall include any title, estate, interest, claim, remedy, power, authority, discretion or other right of any kind, both

- present and future, in, to, under, in respect of or derived from that security, document or other asset or the proceeds of any disposal of that security, document or other asset;
- (f) any statute or any provision of any statute shall include reference to any statutory modification or re-enactment thereof and to any regulations or orders made thereunder and from to time in force;
- (g) the word "**subsidiary**" shall have the same meaning as in section 1159 of the Companies Act 2006;
- (h) a "transfer" of Charged Assets includes any account transfer of some or all of them on the Service Provider's collateral management platform and any other delivery, disposition or other transfer of some or all of them; and
- (i) "the Service Agreement" shall include the whole or part of any Security under this Service Agreement.

# Schedule 2 to the Security (English law) Service Module - Additional terms where registration is required in Australia

- **Application**: The terms set out below shall apply to the Company and the Service Provider and in the event of any inconsistency between this Schedule 2 to this Service Module (*Additional terms where registration is required in Australia*) and the balance of the Service Agreement, this Schedule 2 shall govern.
- 2 Interpretation: Terms not defined within this Schedule 2 to this Service Module (*Additional terms where registration is required in Australia*) have the same meaning as set out in the main body of the Service Agreement.
- **Perfection and Release of the Service Provider's Security**: The following additional Clauses are inserted into Clause 5 (*Advances and Right of Retention*) of the Service Agreement after Clause 5.4:
  - "5.5 The Service Provider may register, and give any notice in connection with, any security interest constituted by or pursuant to the Service Agreement at the Company's expense. This includes registration under the PPSA for whatever collateral class the Service Provider thinks fit. The Company consents to any such registration or notification and agrees not to make an amendment demand. The Company must provide the Service Provider with any information it requires for the purposes of effecting such registration or notification and do all other things as are necessary to effect such registration or notification including giving consent to such registration or notification where required.
    - 5.6 To the maximum extent permitted, for the purposes of sections 115(1) and 115(7) of the PPSA, the Service Provider need not comply with the following sections of the PPSA:-
      - (a) section 95 (notice by secured party of removal of accession);
      - (b) section 118 (enforcing security interests in accordance with land law decisions);
      - (c) section 121(4) (notice by secured party of enforcement of security interest in liquid assets);
      - (d) section 125 (obligation of secured party to dispose of or retain collateral after seizure);
      - section 130, to the extent that it requires the Service Provider to give any notice to the Company (notice by secured party of disposal of collateral);
      - (f) section 132(3)(d) (obligation of secured party to show amounts paid to other secured parties in statement of account); and
      - (g) section 132(4) (statement of account by secured party if it does not dispose of collateral within prescribed period).
    - 5.7 Without limitation to any other provision of the Service Agreement, it is a default of the Company under the Service Agreement for the purposes of section 123(1) of the PPSA if any person with a security interest in a Charged Asset seizes or

becomes entitled to seize that Charged Asset without the consent of the Service Provider.

- 5.8 To the maximum extent permitted:-
  - (a) for the purposes of section 115(7) of the PPSA, the Service Provider need not comply with sections 132 and 137(3) of the PPSA;
  - (b) sections 142 and 143 of the PPSA are excluded; and
  - (c) the Company agrees not to exercise its rights to make any request of the Service Provider under section 275 of the PPSA, to authorise the disclosure of any information under that section or to waive any duty of confidence that would otherwise permit non-disclosure under that section.
- 5.9 If the PPSA is amended after the date of the Service Agreement to permit the Company and the Service Provider to agree to not comply with or to exclude other provisions of the PPSA, the Service Provider may notify the Company that any of these provisions are excluded, or that the Service Provider need not comply with any of these provisions, as notified to the Company by the Service Provider.
- 5.10 To the maximum extent permitted, the Company waives:-
  - (d) its rights to receive any notice that is required by any provision of the PPSA (including a notice of a verification statement) or any other law (regardless of the jurisdiction of such law) before a secured party or Receiver (as defined in Clause 12.1) exercises a right, power or remedy; and
  - (e) any time period that must otherwise lapse under any law (regardless of the jurisdiction of such law) before a secured party or Receiver exercises a right, power or remedy.
- 5.11 If the law (regardless of the jurisdiction of such law) which requires a period of notice or a lapse of time cannot be excluded, but the law (regardless of the jurisdiction of such law) provides that the period of notice or lapse of time may be agreed, that period or lapse is one day or the minimum period the law allows to be agreed (whichever is the longer). However, nothing in this Clause 5.11 prohibits the Service Provider or any Receiver from giving a notice under the PPSA or any other law (regardless of the jurisdiction of such law)."
- **4 Undertakings by the Company**: The following additional clause is inserted into Clause 8 (*Undertakings of the Company*) of the Service Agreement after Clause 8.3:
  - "8.4 The Company agrees to notify the Service Provider:
    - (a) at least 14 days before the Company does any of the following:
      - (i) if the Company does not have an Australian Company Number or Australian Registered Body Number, the Company changes its name; and

the Company becomes trustee of a trust with an Australian Business Number or Australian Registered Scheme Number, or a partner in a partnership with an Australian Business Number; and

- (ii) if the Company has an Australian Company Number or Australian Registered Body Number, as soon as possible after the Company becomes aware that the number will change or cease to apply."
- **Confidentiality**: Clause 6.7 of the Regulatory Terms shall not permit the Service Provider to disclose any information under section 275(4) of the PPSA unless section 275(7) of the PPSA applies."
- **Interpretation**: The following additional definition is inserted into Schedule 1 (*Definitions*) following the definition of "LPA":

""PPSA" means the Personal Property Securities Act 2009 (Cth);"

# Schedule 3 to the Security (English law) Service Module - Additional terms where the Collateral is Korean Collateral

#### **PART A**

- Application: Where an Account, Relevant Securities and/or Deposits are held or located in the Republic of Korea, in addition to the terms of the Service Agreement, the terms set out below shall apply to the Company and the Service Provider. Notwithstanding any provision to the contrary in the Service Agreement, this Security created in favour of the Service Provider over the Relevant Securities and the Deposits credited or to be credited into an Account that are held or located in the Republic of Korea (collectively, the "Korean Collateral"), the terms and conditions set out in this Schedule 3 to this Service Module (Additional terms where the Collateral is Korean Collateral) shall apply. With respect to the Korean Collateral, in the event of any inconsistency between this Schedule 3 to this Service Module (Additional terms where the Collateral is Korean Collateral) and the balance of the Service Agreement, this Schedule 3 shall govern.
- 8 Interpretation: Terms not defined within this Schedule 3 to this Service Module (Additional terms where the Collateral is Korean Collateral) have the same meaning as set out in the Service Agreement.
- **Governing Law and Jurisdiction:** In relation to Korean Collateral only, Clause 20 (*Governing Law and Jurisdiction*) is deleted in its entirety and replaced with the following provision:

## "20. Law and Jurisdiction

- 20.1 The Service Agreement and all matters arising from or related to it (whether contractual or non-contractual in nature) shall be governed by and construed in accordance with English law; provided that the laws of the Republic of Korea shall be applied to the extent necessary in order to interpret and give effect to the provisions in the Service Agreement in so far as such provisions relate to any Korean Collateral.
- 20.2 The English courts shall have exclusive jurisdiction over any actions or proceedings arising directly or indirectly from any one or more of the Service Agreement and the other Collateral Management Agreements, including any actions or proceedings regarding the creation and validity of a security interest under the Service Agreement or the giving of instructions or the taking of any other actions in relation to Relevant Securities or any other Charged Assets (a "Dispute"), and the Company hereby submits to the exclusive jurisdiction of such courts. The parties agree that the English courts are the most appropriate and convenient courts to deal with any such actions or proceedings and, accordingly, they shall not argue to the contrary; provided that without prejudice to the jurisdiction of the English courts, the Seoul Central District Court has jurisdiction to settle any Dispute arising in connection with the Korean Collateral."

#### **PART B**

## 10 In this Part B:

"Korea" means the Republic of Korea;

"Korean Cash Account" means an Account located in Korea that is a Cash Account established in the name of the Company;

"Korean Custodian" means a Custodian or Sub-Custodian with which a Korean Secured Account is opened;

"Korean Secured Account" means a Korean Cash Account or a Korean Securities Account;

"Korean Securities" means any and all securities now credited or hereafter to be credited to the Korean Securities Accounts;

"Korean Securities Account" means an Account located in Korea that is a securities account established in the name of the Company or in the name of the Service Provider with the investment registration certificate of the Company;

"Pledge" means the pledge granted hereunder in favour of the Service Provider; and

"Pledged Property" means all of the Company's right, title and interest, now owned or hereafter acquired in and to (a) all balances, credits, deposits, monies or other sums now or hereafter in any or all of the Korean Cash Accounts or on deposit in any or all of the Korean Cash Accounts and any interest accrued or payable thereon and the proceeds thereof and (b) the Korean Securities.

## 11 Security Interest

- 11.1 The Company hereby pledges and grants, and agrees to pledge and grant, to the Service Provider a first priority, perfected security interest (*Jil Kwon* in Korean) in the Pledged Property which security interest is and shall be continuing security for the purpose of securing the Secured Obligations. The Pledge over the Pledged Property in relation to any Korean Security to be credited to a Korean Securities Account after the date of the Service Agreement shall be deemed to be created at the time when such Korean Security is credited to such Korean Securities Account.
- 11.2 Upon the establishment by the Company of a Korean Secured Account under or pursuant to the Custody Agreement after the date of the Service Agreement, the Company shall cause the relevant Korean Custodian to notify the details thereof to the Service Provider within five days of establishment of each Korean Secured Account. By delivery of any such notice to the Service Provider, the Pledged Property in relation to such Korean Secured Account so notified shall be subject to the Pledge hereunder.

## 12 Perfection of Pledge for Korean Cash Accounts

- 12.1 In respect of each Korean Cash Account, the Company shall, forthwith following the execution of the Service Agreement and, as the case may be, the establishment thereof, (a) deliver a duly completed and executed notice to the relevant Korean Custodian of the Pledge of such Korean Cash Account effected pursuant to paragraph 2.1 of this Schedule 3 to this Service Module (Additional terms where the Collateral is Korean Collateral), in the form set out in Schedule 4 to the Operational Terms, (b) procure delivery to the Service Provider of a copy of such notice and (c) within three days of the date of the Service Agreement, procure an executed acknowledgment and consent thereof, with fixed date stamp affixed thereon, from that Korean Custodian in substantially the form set out in the form of acknowledgment and consent attached to such notice (or in such form as may be acceptable to the Service Provider).
- 12.2 The Company shall from time to time deliver to the Service Provider or to the Service Provider's order, all deeds, certificates and other documents constituting or evidencing title to the Pledged Property in relation to the Korean Cash Accounts or any part thereof, provided that such documents may be released by the Service Provider to the Company as custodian

for and on behalf of the Service Provider or any part thereof including, without limitation, the deposit certificate or passbook in relation to the Korean Cash Accounts and upon the request of the Service Provider, the Company shall promptly deliver all of such documents (so released to the Company) to the Service Provider.

## 13 Perfection of Pledge for Korean Securities

The Company shall deliver or cause to be delivered the Korean Securities to the Service Provider in accordance with the Service Agreement. To effect delivery of the Korean Securities, the Company shall (a) deliver a duly completed and executed notice to the relevant Korean Custodian of the pledge of the Korean Securities effected pursuant to paragraph 2.1 of this Schedule 3 to this Service Module (Additional terms where the Collateral is Korean Collateral), in the form set out in Schedule 4 to the Operational Terms, (b) procure delivery to the Service Provider a copy of such notice, (c) procure an executed acknowledgment and consent thereof, with fixed date stamp affixed thereon, from that Korean Custodian in substantially the form set out in the form of acknowledgment and consent attached to such notice (or in such form as may be acceptable to the Service Provider) and (d) procure that such Korean Custodian enters into each securities Account a statement (the "Pledge Statement") to the effect that the Korean Securities credited to such securities Account are pledged in favour of the Service Provider together with the name and address of the Service Provider, (e) procure that the relevant Korean Custodian agrees that the Pledge Statement shall be deemed to be repeated each time any Korean Security is credited to the relevant securities Account and (f) take all such other action as the Pledgee may reasonably request. Delivery of the Korean Securities by the Company to the Service Provider in the manner described in this paragraph 4 of this Schedule 3 constitutes conclusive evidence of the Pledge of the Korean Securities. The Company agrees that all Korean Securities hereafter delivered pursuant to this paragraph 4 of this Schedule 3 shall for all purposes hereunder be considered to constitute part of the Korean Securities and be pledged hereunder.

# The Bank of New York Mellon Collateral Management Master Agreement (Collateral Provider) Service Module

BNY MELLON

# **Table of Contents**

Contents		Page	
1	Introduction	2	
2	Definitions and Interpretation:	2	
3	Appointment and Authorisation	2	
4	Delivery of Eligible Collateral	3	
5	Valuation and Substitution of Securities	7	
6	Redelivery of Collateral	9	
7	Events of Default	9	
8	Concerning the Service Provider	10	
9	Additional Terms	12	
10	Governing Law and Jurisdiction	12	
	edule 1 to the Collateral Management Master Agreement (Collateral Provider) Service Mo		
	edule 2 to the Collateral Management Master Agreement (Collateral Provider) Service Mo		
- Leg	gal Entity Amendments	18	

## 1 Introduction

Pursuant to the terms of a framework agreement executed and (to the extent applicable) delivered (a "Framework Agreement"), a Service Agreement shall take effect between you and the relevant Service Provider in respect of each of the Service Modules elected to apply to you as set out in the Service Form.

## 2 Definitions and Interpretation:

The following terms shall have the meaning given to them in the Framework Agreement: "Applicable Regulations", "Operational Terms" "Regulatory Terms", "Service Agreement", "Service Form", "Service Module" and "Service Provider".

Wherever used in the Service Agreement, the terms set out in Schedule 1 to this Service Module (*Definitions*) shall, unless the context requires otherwise, have the meanings set out in that Schedule.

In addition, unless the contrary intention appears:

- 2.1.1 except as may otherwise apply for Income payable on particular Securities or as otherwise agreed in writing by the parties, all provisions for the custody, transfer, payment or receipt of Cash shall mean custody or transfer of, payment in, or receipt of, same day freely transferable funds of the relevant currency; and
- 2.1.2 the Service Agreement shall apply separately to each Deal that the Collateral Provider enters into with a Collateral Receiver to which the Service Provider and the Collateral Receiver's Collateral Manager have also agreed to provide collateral management services.

## 3 Appointment and Authorisation

- 3.1 Appointment: In consideration of the agreement of the Service Provider to provide the services in accordance with the Service Agreement, the Collateral Provider appoints the Service Provider with effect from the date of the Service Agreement as its agent to provide services in respect of Deals.
- **3.2** Agents: The Collateral Provider authorises the Service Provider to use (including for the avoidance of doubt, to provide information relating to the Collateral Provider and the Service Agreement) third parties (each an "Agent") (including its affiliated companies and associates and branches of the Service Provider) to perform its duties under the Service Agreement.
- 3.3 Delivery of Securities and Cash: The Collateral Provider authorises the Service Provider to procure delivery to the Collateral Provider's Account and procure receipt from the Collateral Provider's Account of all Securities and Cash transferred in respect of Deals in accordance with the Service Agreement and Instructions and such deliveries shall be deemed to be transferred pursuant to the Collateral Provider's and the Collateral Receiver's respective obligations under the relevant Principal Agreements. The Service Provider will reflect all transfers of Securities and Cash to and from the Collateral Provider's Account on its books and records.

## 4 Delivery of Eligible Collateral

#### 4.1 Prior to Deal Date:

- 4.1.1 On each Business Day that the Collateral Receiver and the Collateral Provider agree to effect a Deal, and in relation to a Loan, on each Business Day on which a Loan is outstanding, and in relation to any Other Transactions on each Business Day on which there exists outstanding obligations to be performed by the Collateral Provider in respect of any such Other Transaction the Collateral Provider shall deliver to the Service Provider, prior to a time specified by the Service Provider, a Confirmation Instruction containing all information in the Confirmation Instruction for the Deal, including in relation to Loans, the aggregate value of Securities (and, in relation to a Reverse Stock Loan, Cash) borrowed or to be borrowed on such Business Day in relation to all such Loans ("Notified Loan Value") and in relation to Transactions, the aggregate value of Cash to be paid by the Collateral Receiver to the Collateral Provider on the Deal Date in relation to all such Transactions ("Notified Purchase Price") and in relation to any Other Transaction, the aggregate value of Eligible Collateral to be maintained in the Collateral Receiver's Account in relation to each such Other Transaction (the "Required Collateral Amount"). If the Service Provider does not receive matching Confirmation Instructions from the Collateral Receiver, or the Collateral Receiver's Collateral Manager on the Collateral Receiver's behalf, and the Collateral Provider, it shall take such further action as it deems necessary or desirable (in its sole discretion) in order to obtain matching Confirmation Instructions.
- 4.1.2 Loans: With respect to each Loan, the Collateral Provider agrees by close of business on the relevant Business Day to deliver or cause to be delivered to the Collateral Provider's Account, Eligible Collateral having a Margin Value (together with any Eligible Collateral already held in the Collateral Provider's Account) that is equal to or greater than the Notified Loan Value attributable to such Loan, specified in the Confirmation Instructions in the relevant Base Currency.
- 4.1.3 Transactions: With respect to each Transaction, the Collateral Provider agrees, at or before the latest time on the Deal Date for delivery of Securities in the location where Securities are to be delivered in respect of each Transaction, to procure delivery in respect of each Transaction to the Collateral Provider's Account, sufficient Eligible Collateral to complete such Transaction on such Deal Date.
- 4.1.4 Other Transactions: With respect to each Other Transaction, the Collateral Provider agrees by close of business on the relevant Business Day to deliver or cause to be delivered to the Collateral Provider's Account, Eligible Collateral having a Margin Value (together with any Eligible Collateral already held in the Collateral Provider's Account) that is equal to or greater than the Required Collateral Amount attributable to such Other Transaction, as set in the Confirmation Instructions in the relevant Base Currency.

## 4.2 Deal Date:

4.2.1 Deal Date for Loans: On the Deal Date for each Loan, the Service Provider shall procure the transfer of Eligible Collateral from the Collateral Provider's Account to the Collateral Receiver's Account in accordance with Confirmation Instructions for such Loan, subject to the following provisions:

- (i) the Service Provider shall ascertain that all Securities and Cash to be transferred to the Collateral Receiver's Account are Eligible Collateral. Any Securities and Cash which are not Eligible Collateral shall not be included in the calculations referred to in Clause 4.3 (*The Service Provider's Inability to Complete a Deal and Partial Deliveries*) or transferred to the Collateral Receiver's Account; and
- (ii) the Service Provider shall determine the Margin Value in relation to the Loan for the purpose of carrying out its duty under Clause 5 (*Valuation and Substitution of Securities*).
- 4.2.2 Deal Date for Transactions: On the Deal Date for each Transaction, the Service Provider shall procure the transfer of Eligible Collateral from the Collateral Provider's Account to the Collateral Receiver's Account in accordance with the Confirmation Instructions for such Transaction, subject to the following provisions:
  - (i) the Service Provider shall ascertain that all Securities and Cash to be transferred to the Collateral Receiver's Account are Eligible Collateral. Any Securities and Cash which are not Eligible Collateral shall not be included in the calculations referred to in Clause 4.3 (*The Service Provider's Inability to Complete a Deal and Partial Deliveries*) or transferred to the Collateral Receiver's Account;
  - (ii) the Service Provider shall determine the Margin Value in relation to the Transaction for the purpose of carrying out its duties under Clause 5 (Valuation and Substitution of Securities); and
  - (iii) the Service Provider shall procure the disbursement from the Collateral Receiver's Account to the Collateral Provider's Account or to such other account as the Collateral Provider may direct, an amount of Cash in the Base Currency equal to the Notified Purchase Price.
- 4.2.3 Deal Date for Other Transactions: On the Deal Date for each Other Transaction, the Service Provider shall procure the transfer of Eligible Collateral from the Collateral Provider's Account to the Collateral Receiver's Account in accordance with Confirmation Instructions for such Other Transaction, subject to the following provisions:
  - (i) the Service Provider shall ascertain that all Securities and Cash to be transferred to the Collateral Receiver's Account are Eligible Collateral. Any Securities and Cash which are not Eligible Collateral shall not be included in the calculations referred to in Clause 4.3 (*The Service Provider's Inability to Complete a Deal and Partial Deliveries*) or transferred to the Collateral Receiver's Account; and
  - (ii) the Service Provider shall determine the Margin Value in relation to the Other Transaction for the purpose of carrying out its duty under Clause 5 (*Valuation and Substitution of Securities*).

#### 4.3 The Service Provider's Inability to Complete a Deal and Partial Deliveries:

4.3.1 Transactions: If the Service Provider is unable to instruct a Transaction in accordance with the Confirmation Instructions because either the Collateral Receiver or the Collateral Provider has failed to transfer, or procure the transfer of, sufficient Cash to the Collateral Receiver's Account or Eligible Collateral to the Collateral Provider's Account, respectively, the Service Provider shall promptly notify the Collateral Receiver, the Collateral Receiver's Collateral Manager and the Collateral Provider. If there is not (a) sufficient Eligible Collateral from the Collateral Provider and/or (b) sufficient Cash from the Collateral Receiver, in each case and as applicable, by the deadline for the receipt of Securities at the appropriate Clearing System or Sub-custodian or, with respect to Physical Securities delivered to the Custodian, by such time as the Custodian may designate with respect to particular types of Physical Securities, or by the latest time on the Deal Date for receipt of same day funds in the location in which such Cash is to be deposited, the Collateral Provider irrevocably agrees and instructs the Service Provider to effect the Transaction as follows, provided that the Collateral Receiver and the Collateral Provider shall remain obligated pursuant to the original terms of each Transaction:

- (i) subject to paragraph (iii) below, if the Market Value of Cash in the Collateral Receiver's Account (such amount to be notified by the Collateral Receiver's Collateral Manager to the Service Provider on its request) is less than the Notified Purchase Price set out in the Confirmation Instructions,
  - (a) the Market Value of Cash in the Collateral Receiver's Account shall be deemed to be the Notified Purchase Price,
  - (b) the aggregate Margin Value of Eligible Collateral to be transferred to the Collateral Receiver's Account shall be reduced accordingly.
  - (c) the remaining terms of the Transaction shall be determined in accordance with the Confirmation Instructions, and
  - (d) the Collateral Receiver and the Collateral Provider shall provide the Collateral Receiver's Collateral Manager and the Service Provider, respectively, with further matching Instructions with respect to a recalculated Notified Purchase Price for such Transaction;
- (ii) in relation to such Transactions, if the Cash in the Collateral Receiver's Account (such amount to be notified by the Collateral Receiver's Collateral Manager to the Service Provider on its request) equal to the Notified Purchase Price (including the deemed Notified Purchase Price pursuant to paragraph (i) above) exceeds the aggregate Margin Value of the Eligible Collateral in the Collateral Provider's Account, the Service Provider shall procure the transfer from the Collateral Receiver's Account to the Collateral Provider's Account Cash in an amount equal to the aggregate Margin Value of such Eligible Collateral, and the difference between the amount credited to the Collateral Provider's Account and the Notified Purchase Price shall be held by the Collateral Receiver's Custodian in the Collateral Receiver's Account; and
- (iii) notwithstanding any other provision of the Service Agreement, if on the Deal Date for any Transaction, the Service Provider has procured the transfer of Eligible Collateral to the Collateral Receiver's Account and has instructed the credit of the Collateral Provider's Account with an amount of Cash in the Base Currency equal to the Notified Purchase Price for such Transaction in accordance with the Confirmation Instructions under this Clause 4 but thereafter the Collateral Receiver's Collateral Manager determines that the

Collateral Receiver's Account has not been finally credited with the amount of Cash in the Base Currency due as payment of the Notified Purchase Price for such Transaction and notifies the same to the Service Provider, the Service Provider shall procure the transfer back to the Collateral Provider's Account of all Securities and Cash as shall have been transferred to the Collateral Receiver's Account in respect of such Transaction and such Securities and Cash shall again be for all purposes the property of the Collateral Provider and the Collateral Provider shall be free to deal with such Securities and Cash on such terms as the Collateral Provider shall decide.

**4.3.2 Simultaneous transfers for all Deals**: The Collateral Provider agrees that in effecting Deals, transfers between the Collateral Receiver's Account and the Collateral Provider's Account shall be deemed to be simultaneous.

#### 4.4 Ownership of Eligible Collateral:

- 4.4.1 Loans: In respect of each Loan, upon the transfer of Eligible Collateral to the Collateral Receiver's Account, subject to the Collateral Provider receiving Securities or other obligations under the Loan, the Eligible Collateral shall be for all purposes the property of the Collateral Receiver during the term of the Loan to which it relates.
- 4.4.2 Transactions: In respect of each Transaction, upon the transfer of Cash to the Collateral Provider's Account and the transfer of Eligible Collateral to the Collateral Receiver's Account, it is agreed by the Collateral Provider that, subject to the Collateral Receiver's Account being finally credited with Cash not less than the Notified Purchase Price for the Transaction in order to facilitate the transfer of Cash to the Collateral Provider's Account, Eligible Collateral in the Collateral Receiver's Account shall be for all purposes the property of the Collateral Receiver during the term of the Transaction to which it relates.
- 4.4.3 Other Transactions: In respect of each Other Transaction, upon the transfer of Eligible Collateral to the Collateral Receiver's Account, the Eligible Collateral shall be for all purposes the property of the Collateral Receiver during the term of the Other Transaction to which it relates.
- 4.4.4 Notwithstanding 4.4.1, 4.4.2 and 4.4.3 above, the Service Provider will refuse to act upon any Instruction of the Collateral Receiver or the Collateral Provider to deliver Eligible Collateral other than as expressly provided in the Service Agreement unless otherwise agreed in writing between the Collateral Receiver, the Collateral Receiver's Collateral Manager and the Collateral Provider from time to time.

#### 4.5 Distributions, etc.

- 4.5.1 If the Service Provider is made aware of an impending Income record date or other distribution date on any Eligible Collateral held by the Collateral Receiver's Custodian in the Collateral Receiver's Account, the Service Provider shall not be required to notify the Collateral Provider but shall act in accordance with the prior Written Instructions of the Collateral Provider specifying the required treatment in respect of such Eligible Collateral unless the Service Provider receives contrary Written Instructions from the Collateral Provider no later than two Business Days prior to the Income record date or other distribution date.
- 4.5.2 If Income is paid by the issuer on Eligible Collateral, the Service Provider shall promptly procure that the Collateral Receiver's Custodian transfers to the Collateral

Provider's Account an amount of Cash equal to, and in the same currency as, such Income.

- 4.5.3 The Service Provider's obligation under the Service Agreement is with respect to amounts actually received after any payment or withholding. If the Collateral Receiver is required to pay such amount on a gross basis under the Principal Agreement, the Collateral Receiver shall be required to pay the amount of any shortfall directly to the Collateral Provider or to put the Collateral Receiver's Custodian in funds to pay the amount of such shortfall to the Collateral Provider.
- 4.6 Corporate Actions, Income, Voting Rights, etc: Except as set out in the Service Agreement, the Service Provider shall have no obligation to the Collateral Provider in respect of Securities the Collateral Receiver's Custodian holds in the Collateral Receiver's Account to:
  - **4.6.1** procure any claim or the receipt of any dividends, interest payments or other entitlements accruing to such Securities;
  - 4.6.2 procure that any voting rights are exercised in relation to such Securities; or
  - 4.6.3 procure any act in relation to conversions, subdivisions, consolidations, takeovers, pre-emption options, subscription rights, other offers or capital reorganisations or other rights relating to such Securities,

unless, only with respect to sub-Clauses 4.6.2 and 4.6.3, the Collateral Receiver's Collateral Manager has received Written Instructions from the Collateral Provider, or the Service Provider on the Collateral Provider's behalf, at least two Business Days prior to the date on which such action is to be taken and in the case of a call on partly paid Securities or a rights issue, all and any sum due has been paid by the Collateral Provider into the Collateral Receiver's Account or the Collateral Provider's Account, as appropriate.

#### 5 Valuation and Substitution of Securities

**5.1 Valuation of Securities**: At the opening of each Business Day during which a Deal is outstanding, the Service Provider shall determine the Margin Value of all Eligible Collateral in the Collateral Receiver's Account.

#### 5.2 Margin Maintenance:

Margin Deficit: If the aggregate value of the Termination Values of each outstanding Relevant Deal on any Business Day is greater than the aggregate Margin Value of Eligible Collateral in the Collateral Receiver's Account with respect to such Relevant Deals, the Service Provider shall promptly notify the Collateral Provider. On the date of such notice, the Collateral Provider shall transfer to the Collateral Provider's Account (for onward transfer to the Collateral Receiver's Account) sufficient additional Eligible Collateral ("Additional Eligible Collateral") so that, after transfer of such Additional Eligible Collateral to the Collateral Receiver's Account, the aggregate Margin Value of Eligible Collateral (including such Additional Eligible Collateral) equals or exceeds the aggregate value of the Termination Values of all such outstanding Relevant Deals. If the Collateral Provider fails to transfer an appropriate amount of Additional Eligible Collateral on the date of any such notice, the Service Provider shall notify the Collateral Receiver, the Collateral Receiver's Collateral Manager and the Collateral Provider. All Additional Eligible Collateral

transferred to the Collateral Receiver's Account shall be deemed to be Eligible Collateral for the purposes of the Service Agreement.

- 5.2.2 Margin Excess: If the aggregate Margin Value of the Eligible Collateral held in the Collateral Receiver's Account on any Business Day exceeds the aggregate value of the Termination Values of all outstanding Relevant Deals (the amount of such excess being the "Margin Excess"), the Service Provider or the Collateral Receiver's Collateral Manager shall procure the transfer of Eligible Collateral from the Collateral Receiver's Account to the Collateral Provider's Account having an aggregate Margin Value equal to the Margin Excess. The Collateral Receiver has authorised the Service Provider and the Collateral Receiver's Collateral Manager to instruct the transfer of Equivalent Collateral from the Collateral Receiver's Account pursuant to this Clause 5.2.25.2.2 and the Collateral Provider authorises the Service Provider and the Collateral Receiver's Collateral Manager to act on its behalf in selecting the Equivalent Collateral to be transferred.
- 5.3 Substitution: The Service Provider is authorised by the Collateral Receiver, upon Instructions from the Collateral Provider received by the Service Provider by the relevant deadline for the applicable Clearing System or Sub-custodian (as notified by the Service Provider to the Collateral Provider from time to time) on any Business Day, to procure the transfer of Equivalent Collateral from the Collateral Receiver's Account to the Collateral Provider's Account in accordance with the relevant deadline set out above against the transfer, by way of substitution, to the Collateral Receiver's Account of Alternative Collateral which is Eligible Collateral determined by the Service Provider to have a Margin Value (taken together with any Eligible Collateral not substituted under this Clause 5.3) not less than the aggregate value of the Termination Values of all outstanding Relevant Deals (or to the extent that the Margin Value of Eligible Collateral in the Collateral Receiver's Account was less than the aggregate value of the Termination Values of all outstanding Relevant Deals immediately prior to such substitution, such deficit has not been increased by such substitution).

The Collateral Provider understands and acknowledges that such substitution may not occur simultaneously if the operating hours of any Clearing Systems or Sub-custodians involved do not coincide (whether as a result of time differences or any other factor).

If there is insufficient Alternative Collateral which is Eligible Collateral in the Collateral Provider's Account (the Service Provider shall notify the Collateral Receiver's Collateral Manager of the same), the Collateral Provider authorises the Service Provider to procure the transfer of Cash having the same Margin Value as the substituted Eligible Collateral (and, for the purposes of this Clause 5.3 only, Cash shall be deemed to be Eligible Collateral notwithstanding the provisions of any agreement to the contrary) from the Collateral Provider's Account to the Collateral Receiver's Account against the transfer of Equivalent Collateral from the Collateral Receiver's Account to the Collateral Provider's Account.

If there is insufficient Cash available in the Collateral Provider's Account (the Service Provider shall notify the Collateral Receiver's Collateral Manager of the same), the Service Provider shall reduce the amount of the Equivalent Collateral that it instructs to be transferred from the Collateral Receiver's Account to the Collateral Provider's Account so that their Margin Value is equal to or less than the Margin Value of such Cash.

If no Cash is available in the Collateral Provider's Account (the Service Provider shall notify the Collateral Receiver's Collateral Manager of the same), the Service Provider shall delay the transfer of the Equivalent Collateral until the transfer of the Alternative Collateral to the Collateral Receiver's Account.

#### 6 Redelivery of Collateral

#### 6.1 Transfer of Securities and Cash:

- 6.1.1 Loans: In relation to each Loan, the Service Provider shall promptly procure the redelivery of Equivalent Collateral to the Collateral Provider on termination of the relevant Loan on receipt of Instructions to do so by the Collateral Receiver, or the Collateral Receiver's Collateral Manager on the Collateral Receiver's behalf. The Service Provider shall have no duty to enquire at the time of the redelivery as to whether the Collateral Provider has performed any obligations it may have to the Collateral Receiver or as to any other matter.
- 6.1.2 Transactions: In relation to each Transaction, the Collateral Provider agrees to transfer Cash in the amount of the Termination Value in the Base Currency, to the Collateral Provider's Account before the Termination Date and, subject to Clause 6.2 (Shortfalls of Cash in relation to Transactions), irrevocably instructs the Service Provider to procure the payment to the Collateral Receiver of Cash in the amount of the Termination Value in the Base Currency by crediting Cash to the Collateral Receiver's Account.
- 6.1.3 Other Transactions: In relation to each Other Transaction, the Service Provider shall promptly procure the redelivery of Equivalent Collateral to the Collateral Provider on termination of the relevant Other Transaction on receipt of Instructions to do so by the Collateral Receiver, or the Collateral Receiver's Collateral Manager on the Collateral Receiver's behalf. The Service Provider shall have no duty to enquire at the time of the redelivery as to whether the Collateral Provider has performed any obligations it may have to the Collateral Receiver or as to any other matter.
- 6.2 Shortfalls of Cash in relation to Transactions: In relation to each Transaction, if the Collateral Provider's Account does not contain sufficient Cash available to repurchase all Eligible Collateral with respect to a Transaction on the Termination Date, the Service Provider shall notify the Collateral Provider, the Collateral Receiver's Collateral Manager and the Collateral Receiver, and the Collateral Provider shall give the Service Provider Instructions identifying which Eligible Collateral is to be repurchased, and the adjusted Termination Value, provided that the Collateral Receiver and the Collateral Provider shall remain obligated pursuant to the original terms of such Transaction.
- **6.3** Amendment of Termination Date: The Termination Date for a Transaction shall not be changed subsequent to the Deal Date unless the Service Provider receives matching Instructions to that effect from the Collateral Receiver, or the Collateral Receiver's Collateral Manager on the Collateral Receiver's behalf, and the Collateral Provider.
- **6.4 On Demand Transactions**: In the case of on demand Transactions, if the Collateral Provider makes the demand, the Collateral Provider shall notify the Service Provider of such demand immediately upon notifying the Collateral Receiver to the Deal of the demand.

#### 7 Events of Default

Where an Event of Default occurs in relation to a Deal, and the Collateral Provider is the Non-Defaulting Party, it shall promptly notify the Service Provider in writing of the occurrence

of such Event of Default. The Service Provider may rely upon such notice without further enquiry. The Service Provider shall promptly notify the Defaulting Party and the Collateral Receiver's Collateral Manager of its receipt of notification of an Event of Default from the Non-Defaulting Party. Irrespective of whether the Collateral Provider is the Defaulting Party or Non-Defaulting Party, the Service Provider's obligations in respect of Clauses 4 (*Delivery of Eligible Collateral*), 5 (*Valuation and Substitution of Securities*) and 6 (*Redelivery of Collateral*) for that Deal in which an Event of Default has occurred shall cease upon receipt of such notice from the Non-Defaulting Party. The Service Provider shall have no obligation to act pursuant to Written Instructions if it believes in good faith that such action will violate any applicable statute, regulation, rule, order or judgment.

#### 8 Concerning the Service Provider

- **8.1** Indemnity: The Collateral Provider agrees to indemnify the Service Provider and any of its affiliated companies and associates on demand (on an after-tax basis) and to release and hold it harmless against any and all:
  - 8.1.1 taxes, assessments, duties and other governmental charges or withholding of a similar nature (including but not limited to interest and penalties), for which the Service Provider and any of its affiliated companies and associates is or is sought to be made liable or accountable in respect of any Deal or in connection with the Service Agreement; and
  - 8.1.2 Losses (including claims by the Collateral Receiver or the Collateral Provider) which are sustained by the Service Provider and any of its affiliated companies and associates in respect of any Deal or in connection with the Service Agreement or which are the result of any action or inaction by the Collateral Receiver, the Collateral Provider, a third party or the Service Provider and any of its affiliated companies, except for those Losses directly arising out of the negligence, fraud, or wilful default of the Service Provider or an affiliated company of the Service Provider or an associate of the Service Provider. It is expressly understood and agreed that the Service Provider's or any of its affiliated companies' or associates' rights to these indemnities shall be enforceable against the Collateral Provider directly, without any obligation to first proceed against any third party for whom it may act, and irrespective of any rights or recourse that the Collateral Provider may have against any such third party. This indemnity shall be a continuing obligation of the Collateral Provider notwithstanding the termination of any Deals or of the Service Agreement.
- **8.2 No Guarantee**: Notwithstanding any other provision of the Service Agreement, it is expressly agreed and acknowledged by the Collateral Provider that the Service Provider is not:
  - 8.2.1 guaranteeing performance of, or assuming any liability for, the obligations of the Collateral Receiver or the Collateral Provider, or the effectiveness of the collateral arrangements, under the Service Agreement or the Principal Agreement;
  - **8.2.2** assuming any credit risk associated with Deals, which liabilities and risks are the responsibility of the Collateral Receiver and the Collateral Provider;
  - 8.2.3 undertaking to make credit available to the Collateral Provider to enable it to complete Deals; or
  - **8.2.4** liable for any Losses which result from the general risks of investing or of investing or holding assets in a particular holding or jurisdiction.

- **8.3 No Duty of Enquiry**: Notwithstanding any other provision of the Service Agreement, the Service Provider shall be under no obligation to inquire into, and shall not be liable for:
  - 8.3.1 the validity of the issue of any Securities purchased or sold by or for the Collateral Receiver or the Collateral Provider, the legality of the purchase or sale of such Securities or the propriety of the amount paid or received for such Securities;
  - **8.3.2** the due authority of any Authorised Person to act on behalf of the Collateral Provider with respect to Cash or Securities held in the Collateral Provider's Account;
  - 8.3.3 the due authority of the Collateral Receiver, the Collateral Provider or any entities for which the Collateral Receiver or the Collateral Provider acts to purchase, sell or hold any particular Security under the Service Agreement; or
  - 8.3.4 the terms of any Principal Agreement.

#### 8.4 Instructions:

- 8.4.1 In the Service Agreement, "Instructions" shall mean Written Instructions issued by the Collateral Provider, provided that where it is not reasonably practicable for the Collateral Provider to do otherwise, such Instructions may take the form of Oral Instructions instead (subject always to paragraph 8.4.2 below). The Service Provider shall be under no obligation to enquire whether it was reasonably practicable for the party giving such Oral Instructions to do otherwise. All Instructions from the Collateral Provider shall be communicated to the Service Provider at the address, fax number or telephone number (as applicable) specified in Part All (Instructions) of Schedule 1 to the Operational Terms (Collateral Management Master Agreement (Collateral Provider) Service Module Forms of Notice) (as amended from time to time).
- 8.4.2 Subject to the terms below, the Service Provider shall be entitled to rely upon any Instructions actually received by the Service Provider and reasonably believed by the Service Provider to be duly authorised and delivered. The Collateral Provider agrees to forward to the Service Provider Written Instructions confirming Oral Instructions given by it by the close of business of the same day that such Oral Instructions are given to the Service Provider. The Collateral Provider agrees that the fact that such confirmatory Written Instructions are not received or that contrary Written Instructions are received by the Service Provider shall in no way affect the validity or enforceability of transactions authorised by such Oral Instructions and effected by the Service Provider.
- 8.4.3 If the Service Provider receives Written Instructions that appear to have been transmitted by the Collateral Provider via (i) fax, email, the internet or other unsecure electronic method provided that the Service Provider has previously agreed with the Collateral Provider the purposes for which such unsecure electronic methods may be used, or (ii) secure electronic transmission containing applicable authorisation codes, passwords and/or authentication keys, the Collateral Provider understands and agrees that the Service Provider cannot determine the identity of the actual sender of such Written Instructions and that the Service Provider shall conclusively presume that such Written Instructions have been sent by an Authorised Person. The Collateral Provider shall be responsible for ensuring that only Authorised Persons transmit such Written Instructions to the Service Provider and that all Authorised Persons treat applicable user and authorisation codes, passwords and/or authentication keys with all reasonable care.

- 8.4.4 The Collateral Provider acknowledges and agrees that it is fully informed of the protections and risks associated with the various methods of transmitting Written Instructions to the Service Provider and that there may be more secure methods of transmitting Written Instructions than the method(s) selected by the Collateral Provider. The Collateral Provider agrees that the security procedures (if any) to be followed in connection with its transmission of Written Instructions provide to it a commercially reasonable degree of protection in light of its particular needs and circumstances.
- 8.5 Reliance on Exchange Rate/Pricing Services and Data Vendors: The Service Provider is authorised to use any generally recognised currency exchange rate quotation service, data vendor, and any generally recognised pricing information service (including brokers and dealers of Securities, Clearing Systems or Sub-custodians) in order to perform its collateral eligibility screening and valuation responsibilities under the Service Agreement, and the Collateral Provider agrees that the Service Provider shall not be liable for any Losses suffered or incurred by the Collateral Provider as a result of errors or omissions of any such exchange rate quotation service, data vendor, pricing information service, broker or dealer.
- **8.6 Principal Agreement:** Without limiting the generality of Clause 8.2 (*No Guarantee*), the Collateral Provider acknowledges and agrees that the Service Provider is not a party to and has no rights or obligations under each Principal Agreement and the Service Provider has not provided it with, and it has not relied upon the Service Provider to provide it with, any advice or confirmation concerning the legal, tax, regulatory or commercial implications of entering into each Principal Agreement and each Deal.
- **8.7 No Additional Duties**: The Service Provider shall have no duties or responsibilities save as specifically set forth in the Service Agreement, and no covenant or obligation shall be implied in the Service Agreement against the Service Provider.

#### 9 Additional Terms

- **9.1 Authorised Persons**: Subject to Clause 8.4 (*Instructions*), the Service Provider shall be fully protected in acting under the provisions of the Service Agreement upon Oral or Written Instructions from a person reasonably believed to be an Authorised Person as set forth in the last such notice delivered to the Service Provider.
- 9.2 Other agreements: The Service Agreement shall not apply to any repurchase transaction, Over the Counter transaction or loan of securities in respect of which the Collateral Provider, the Collateral Receiver, the Collateral Receiver's Collateral Manager and the Service Provider have each agreed the provisions of any other agreement shall apply.

#### 10 Governing Law and Jurisdiction

- **10.1 Governing Law**: The Service Agreement and this Service Module and any non-contractual obligations arising out of or in connection with them are governed by, and shall be construed in accordance with, English law.
- 10.2 Jurisdiction: The courts of England are to have exclusive jurisdiction to settle any disputes that may arise out of or in connection with the Service Agreement and this Service Module (including non-contractual disputes or claims) and accordingly any legal action or proceedings arising out of or in connection with the Service Agreement and this Service Module ("Proceedings") shall be brought in such courts. The parties irrevocably submit to

the exclusive jurisdiction of such courts and waive any objection to Proceedings in such courts whether on the ground of venue or on the ground that the Proceedings have been brought in an inconvenient forum.

# Schedule 1 to the Collateral Management Master Agreement (Collateral Provider) Service Module - Definitions

- "affiliated company" and "associate" shall have the same meaning as in the FCA Rules;
- "Alternative Collateral" Eligible Collateral provided by way of substitution in accordance with the provisions of Clause 5.3 (Substitution);
- "Authorised Person" any person duly authorised to give Oral Instructions and Written Instructions on behalf of the Collateral Provider, such persons and their specimen signatures to be designated in a notice to the Service Provider;
- "Base Currency" in relation to a Deal, the currency of the Deal as set out in Confirmation Instructions;
- "Business Day" in relation to a Deal, a day which is a Business Day pursuant to the relevant Principal Agreement and which is also a day on which the Service Provider and the Collateral Receiver's Collateral Manager is open for business in connection with the Service Agreement;
- "Cash" the money and currency of any jurisdiction which the Custodian accepts for deposit in a cash account;
- "Clearing System" Clearstream (Luxembourg), CREST, The Depository Trust and Clearing Corporation, Euroclear, the Federal Reserve/Treasury Book-Entry System or any other clearing agency, securities depository, book-entry system or other entity that provides handling, clearing or safekeeping services with which the Custodian or a Sub-custodian maintains accounts to hold Securities or Cash that are the subject of Deals;
- "Collateral Provider" you, as defined in the Framework Agreement;
- "Collateral Provider's Account" the clearing or custodial account established and maintained by the Custodian on behalf of the Collateral Provider, or the Collateral Provider's depositary (if any), for the deposit of Eligible Collateral, *inter alia*, with respect to Deals and any account for the deposit of Cash maintained by the Custodian in connection with such clearing or custodial account;
- "Collateral Receiver" the person that is the counterparty to the Collateral Provider in respect of a Deal, to which person the Collateral Provider is obliged (under the terms of the applicable Principal Agreement) to provide the relevant assets on the applicable Deal Date and on each subsequent Business Day on which a Deal continues;
- "Collateral Receiver's Account" the custodial account established and maintained by the Collateral Receiver's Custodian on behalf of the Collateral Receiver, or the Collateral Receiver's depositary (if any) for the deposit of Eligible Collateral with respect to Deals and any account for the deposit of Cash maintained by the Collateral Receiver's Custodian in connection with such custodial account;
- "Collateral Receiver's Collateral Manager" means either The Bank of New York Mellon, London branch or The Bank of New York Mellon, S.A./N.V. as the case may be;
- "Collateral Receiver's Custodian" means the custodian of the Collateral Receiver to which the Collateral Receiver's Collateral Manager has authority to instruct in respect of the Service Agreement;

"Confirmation Instruction" an Instruction substantially in the appropriate form annexed at Part BI (Confirmation Instructions) and BII (Confirmation Instructions) of Schedule 1 to the Operational Terms (Collateral Management Master Agreement (Collateral Provider) Service Module – Forms of Notice) (as amended from time to time);

"Custodian" The Bank of New York Mellon, London branch;

"Deal" a Transaction, Loan and/or Other Transaction, as the case may be, and "Deals" shall have the corresponding meaning;

"Deal Date" the first day that a delivery of Eligible Collateral is due in accordance with the terms of a Deal;

"Defaulting Party" either the Collateral Provider or the Collateral Receiver, as the case may be, in respect of which the relevant Event of Default has occurred under the relevant Principal Agreement in respect of a Deal;

"Eligible Collateral" Securities and Cash of a type or currency that may be agreed in writing among the Service Provider, the Collateral Provider, the Collateral Receiver's Collateral Manager and the Collateral Receiver (and as amended from time to time in accordance with Clause 17 (*Amendments to Eligible Collateral*) of the Framework Agreement) and provided, or to be provided, by the Collateral Provider to the Collateral Receiver, as collateral with respect to a Deal in return for the receipt of the relevant assets by the Collateral Provider from the Collateral Receiver;

"Equivalent Collateral" in relation to any Eligible Collateral (whether Cash or non-Cash) provided under the Service Agreement, Securities or other property of an identical type, nominal value, description and amount to the particular Eligible Collateral provided;

"Event of Default" an event that occurs in respect of a Defaulting Party and the occurrence of which has the effect under the Principal Agreement of requiring the Collateral Provider and the Collateral Receiver to perform their obligations under the Principal Agreement at the time that the event occurs or immediately after the event occurs;

"Income" any dividends, or other distributions;

"Instruction" as defined in Clause 8.4 (Instructions);

"Loan" a loan of Securities made available to the Collateral Provider by the Collateral Receiver pursuant to the terms of the Stocklending Agreement or a Reverse Stock Loan, in both cases where the Service Provider is instructed to deliver and take delivery of Securities only (and the Service Provider will not process any transfer of cash between Collateral Provider and Collateral Receiver), subject to the Service Agreement;

"Losses" any and all losses including but not limited to claims, liabilities, damages, costs, expenses, penalties, actions, demands, settlements, investigations, proceedings and judgments (including legal and other professional advisors' fees and expenses) sustained by a person;

"Margin Percentage" the percentage agreed among the Collateral Receiver, the Collateral Provider, the Collateral Receiver's Collateral Manager and the Service Provider in respect of a Deal with respect to specific types of Eligible Collateral;

"Margin Value" the amount obtained by dividing the Market Value of Securities or Cash, as the case may be, by the applicable Margin Percentage;

"Market Value" (A) in relation to any Security as of any date, the amount as calculated by the Service Provider, after conversion to the Base Currency, equal to the sum of: (i) the market value of such

Security based on the most recently available closing bid price made available to the Service Provider, such closing price usually obtained on the Business Day immediately preceding the Business Day on which the Service Provider calculates the market value from pricing information services which the Service Provider uses generally for pricing such Securities; and (ii) in the case of a fixed income debt security, accrued but unpaid Income, if any, on such Security (provided that if the Service Provider is unable to obtain the price of such Security from such pricing information services on any Business Day, the market value shall be as determined in good faith by the Service Provider in the reasonable exercise of its discretion based on information furnished to the Service Provider by one or more brokers in such Security or on the basis of a formula utilised by the Service Provider for such purpose in the ordinary course of its business); and (B) in relation to Cash or any certificate of deposit, its nominal or face amount;

"Non-Defaulting Party" either the Collateral Provider or the Collateral Receiver, as the case may be, in respect of which the relevant Event of Default has not occurred under the relevant Principal Agreement in respect of a Deal;

"Oral Instructions" verbal instructions or directions actually received by the Service Provider;

"Other Agreement" any ISDA Master Agreement or equivalent thereof, pursuant to which the Collateral Provider and Collateral Receiver have executed Other Transactions;

"Other Transaction" any Over-the-Counter derivative transaction entered into between the Collateral Provider and Collateral Receiver, pursuant to which the Service Provider is instructed to deliver and take delivery of Securities only as collateral, subject to the Service Agreement;

"Physical Securities" Securities and money market instruments in certificated form;

"Principal Agreement" in relation to a Deal, the Stocklending Agreement, Repurchase Agreement or Other Agreement under which that Deal is made;

"Relevant Deal" either (i) a Transaction or Transactions between the Collateral Receiver and the Collateral Provider; (ii) a Loan or Loans between the Collateral Receiver and the Collateral Provider or (iii) an Other Transaction or Other Transactions between the Collateral Receiver and the Collateral Provider;

#### "Repurchase Agreement" the standard form of:

- (a) the PSA/ISMA Global Master Repurchase Agreement (November 1995 version) or the TBMA/ISMA Global Master Repurchase Agreement (October 2000 version) or the Sifma/ICMA Global Master Repurchase Agreement (2011 version);
- (b) the AFTB Master Agreement for Repurchase Transactions;
- (c) the January 2001 FBE Master Agreement for Financial Transactions (European Master Agreement) (including each annex to such agreement governing the entering into of Transactions):
- (d) the 2002 ISDA Master Agreement (including each annex to such agreement governing the entering into of Transactions); or
- (e) any other agreement pursuant to which the Collateral Receiver and the Collateral Provider respectively agree to purchase and repurchase Securities for Cash on the basis of the transfer of all right, title and interest in those Securities and that Cash between the Collateral Receiver and the Collateral Provider,

entered into between the relevant Collateral Receiver and the Collateral Provider in respect of a Transaction other than a Reverse Stock Loan;

"Reverse Stock Loan" a loan of Securities made available to the Collateral Receiver by the Collateral Provider in exchange for cash pursuant to the terms of a Stocklending Agreement and notified to the Service Provider in accordance with a Confirmation Instruction either as (i) a Transaction; or (ii) a Loan;

"Securities" all debt and equity securities and other instruments and intangible assets (including instruments representing the right to receive, purchase or subscribe to the foregoing or representing other rights or interests in the foregoing) as may be agreed from time to time by the Service Provider and the Collateral Provider and Physical Securities and "Security" shall have the corresponding meaning;

#### "Stocklending Agreement" the standard form of:

- (a) the May 2000 version or the January 2010 version of the Global Master Securities Lending Agreement or the December 1995 version of the Overseas Securities Lender's Agreement;
- (b) the February 1997 AFTI Master Agreement for Loans of Securities;
- (c) the January 2001 FBE Master Agreement for Financial Transactions (European Master Agreement) (including each annex to such agreement agreeing the entering into of Loans);
- the 2002 ISDA Master Agreement (including each annex to such agreement governing the entering into of Loans); or
- (e) any other agreement pursuant to which the Collateral Receiver and the Collateral Provider agree to lend and redeliver Securities in return for the transfer of Securities and/or Cash on the basis of the transfer of all right, title and interest in those Securities and that Cash between the Collateral Receiver and the Collateral Provider,

entered into between the relevant Collateral Receiver and the Collateral Provider in respect of a Loan;

"Sub-custodian" a bank or financial institution appointed to act as a sub-custodian of the Custodian from time to time, excluding a Clearing System;

"**Termination Date**" in relation to a Deal, the date on which the Deal is terminated under the relevant Principal Agreement;

"Termination Value" (i) in relation to a Transaction on any date (including the Termination Date), the value of cash (including the relevant cash or reporate) which the Collateral Provider must, if required to do so, transfer to the Collateral Receiver to repurchase the Eligible Collateral, including Alternative Collateral, from the Collateral Receiver; (ii) in relation to a Loan, the Notified Loan Value and (iii) in relation to Other Transactions the Required Collateral Amount;

"Transaction" a transaction effected pursuant to the Repurchase Agreement or a Reverse Stock Loan, in both cases where the Service Provider is instructed to deliver and take delivery of Securities and Cash (as applicable), subject to the Service Agreement; and

"Written Instructions" written communications, other than notices, actually received by the Service Provider by S.W.I.F.T., letter, fax transmission or such other method specified by the Service Provider from time to time as available for use in connection with the services provided under the Service Agreement.

# Schedule 2 to the Collateral Management Master Agreement (Collateral Provider) Service Module - Legal Entity Amendments

Where the Service Provider is The Bank of New York Mellon, London branch, the following provisions shall apply:

N/A

Where the Service Provider is The Bank of New York Mellon, S.A./N.V., the following provisions shall apply:

- 1 References to London, The Bank of New York Mellon, London branch and the State of New York
- 1.1 The definition of "Custodian" shall be to The Bank of New York Mellon, S.A./N.V.

## The Bank of New York Mellon Dealer Custody (English Law) Service Module



#### **Table of Contents**

Contents		Page	
1	Introduction	2	
2	Definitions and Interpretation	2	
3	Appointment of the Service Provider as Custodian	2	
4	Duties and Responsibilities	5	
5	General Authorisations	12	
6	Advances, Overdrafts and Indebtedness	12	
7	Instructions	14	
8	Indemnity	15	
9	Governing Law and Jurisdiction	15	
Sche	Schedule 1 to the Dealer Custody (English law) Service Module - Definitions		

#### 1 Introduction

Pursuant to the terms of a framework agreement executed and (to the extent applicable) delivered (a "Framework Agreement"), a Service Agreement shall take effect between you and the relevant Service Provider in respect of each of the Service Modules elected to apply to you as set out in the Service Form.

#### 2 Definitions and Interpretation

The following terms shall have the meaning given to them in the Framework Agreement: "Operational Terms", "Regulatory Terms", "Service Agreement", "Service Form", "Service Module" and "Service Provider".

Wherever used in the Service Agreement, the terms set out in Schedule 1 to this Service Module (*Definitions*) shall, unless the context requires otherwise, have the meanings set out in that Schedule.

#### 3 Appointment of the Service Provider as Custodian

- 3.1 You appoint the Service Provider with effect from the date of the Service Agreement as custodian of the Securities deposited for safekeeping with the Service Provider or with any Sub-custodian appointed by the Service Provider and to hold Cash in accordance with the terms of the Service Agreement.
- The Service Provider may from time to time appoint any Sub-custodian in connection with the performance of its duties under the Service Agreement. The Service Provider shall exercise reasonable care in selecting and continuing to use Sub-custodians in each country in light of the established rules, practices and procedures then prevailing in each such country. The Service Provider may subcontract the performance of any part or all of its rights, duties, obligations or functions under the Service Agreement (and, for the avoidance of doubt, provide information relating to you and the Service Agreement) to any third party (including affiliated companies, associates or branches of the Service Provider) selected by the Service Provider or its affiliated companies, associates or branches. The Service Provider shall exercise reasonable care in selecting and continuing to use such third party service providers.
- 3.3 The Service Provider is authorised and directed to, and shall open and maintain in its books in such name as you shall reasonably direct:
  - (a) one or more Securities Accounts for the custody and safekeeping, in accordance with the terms of the Service Agreement, of any Securities and non-cash Distributions deposited by or for you with, or otherwise received for you by, the Service Provider, any Sub-custodian or any Clearing System in any country in which the Service Provider provides custody services (each a "Securities Account");
  - (b) one or more cash accounts for all Cash, Distributions and monies received for the account of you or your nominee (each a "Cash Account"). In certain circumstances Cash held for you will be credited to an internal client omnibus operating account established for more than one client (rather than established in your own name). Cash held by the Service Provider in this internal client operating account will be reconciled daily with your records such that an individual customer's entitlement is identifiable at all times; and

- (c) Accounts that contain your Securities or Cash may be subject to the law of other jurisdictions, including those of non EEA jurisdictions, and your rights may be different from those that would apply were English law to be applicable.
- Cash held for you is held by the Service Provider as banker and not as a trustee under the Client Money Rules, save as provided in the Regulatory Terms. In the event of the failure of the Service Provider, the Client Money Distribution and Transfer Rules will not apply to such Cash and so you will not be entitled to share in any Distribution under the Client Money Distribution and Transfer Rules.
- 3.5 There are limited circumstances in which Service Provider may hold certain sums as client money for Your benefit in accordance with the Client Money Rules. These circumstances are limited to the requirements under the Client Asset Rules pursuant to which Service Provider may be required to segregate certain sums from Service Provider's own funds as client money in certain cases where Service Provider has identified a shortfall in the number of client securities held by or for it. Such segregation will continue until such time as the relevant shortfall has been resolved at which point Service Provider will re-appropriate such money. Such client money amount will be held in accordance with the Client Money Rules on your behalf, to the extent that you are affected by the relevant shortfall. In the absence of a Service Provider failure, such segregation does not create a cash entitlement for You against the Service Provider. In the event of the failure of Service Provider, the Client Money Distribution and Transfer Rules will apply to any such money held as client money by Service Provider. Client money will be held with a third party bank or banks. Service Provider does not accept any liability for any default or delay in the distribution of client money in the event of the failure of a bank holding client money on its behalf. Should the failure of a bank with which Service Provider holds any client money occur at the same time as the failure of Service Provider, You may share in any shortfall of client money on a pro rata basis. Service Provider may from time to time notify You of other circumstances in which it may hold client money in accordance with the Client Money Rules. Service Provider shall not pay any interest earned on any client money to You. In the limited circumstances described in this paragraph in which Service Provider holds certain sums as client money for Your benefit in accordance with the Client Money Rules, our standard practice would be for Service Provider to open accounts with third party banks within the United Kingdom but there may be reasons (including, but not limited to diversification requirements) where Service Provider may arrange for such money to be held outside of the United Kingdom. Such money may be held in accounts with a third party bank or banks in a state which is not an EEA Member State and, in such case, the relevant accounts will be subject to the laws of that state and as a result such money may be treated in a different manner from that which would apply if such money were held by a third party bank(s) located in the EEA.
- 3.6 In the event that the Service Provider is required in a particular market to open a Cash Account (with a Sub-custodian) on behalf of you in your own name, you irrevocably authorise the Service Provider to open such account(s) and to give, on your behalf, all such Instructions to the relevant Sub-custodian in a particular market, as the Service Provider deems necessary and required to fulfil the requirements of the Service Agreement.
- 3.7 The Service Provider will identify the Client Securities in its books and records as being beneficially owned by you or, if you have advised the Service Provider that you are acting on behalf of others, by such others and in such event will identify Client Securities as being beneficially owned by your clients so that such books and records record that Client Securities are held separately from Proprietary Securities.

- 3.8 The Service Provider shall hold Property at the Service Provider, Clearing System and Subcustodians and the Service Provider may amend Schedule 3, Part D to the Operational Terms (Global Custody Network Countries and Sub-Custodians) from time to time in its sole discretion. The Service Provider shall notify you orally or in writing of such amendment as soon as reasonably practicable. Property may be held only with Sub-custodians which have entered into a Sub-custodian Agreement with the Service Provider and with any relevant Clearing System. Sub-custodians may hold Property in a Clearing System in which such Sub-custodians participate or are members.
- 3.9 Property held with Sub-custodians shall be held subject to the terms and conditions of the relevant Sub-custodian Agreement. Property held in Clearing System shall be held in accordance with, and subject to, the agreements, rules, regulations and conditions imposed by such Clearing System.
- 3.10 Your Securities may be held in an omnibus Client Securities Account with Securities of other customers of the Service Provider at a Sub-custodian or Clearing System and will be treated as fungible with all other Securities of the same issue held in such account by the Service Provider with such Sub-custodian or Clearing System. This means that your redelivery rights in respect of the Securities are not in respect of the Securities actually deposited with the Service Provider from time to time but rather in respect of Securities of the same number, class, denomination and issue as those Securities originally deposited with the Service Provider in the Securities Accounts from time to time. Such Sub-custodian or Clearing System may then hold your Securities in an omnibus account with a third party that it engages (a "delegate"). Your entitlement to your Securities may not be identifiable by separate certificates or other physical documents of title, or an equivalent electronic record at any relevant Sub-custodian, Clearing System or delegate.
- 3.11 If Property is held (directly or indirectly) with a Sub-custodian, Clearing System or delegate which defaults or becomes insolvent, or if the Service Provider becomes insolvent, the consequences for you will depend in part upon the relevant applicable law (which may not be English law), the effects of which are outside the control of the Service Provider. Where such a default or insolvency occurs, there may be (amongst other risks) delays in settling or transferring Property or, where the Service Provider, a relevant Sub-custodian, Clearing System or delegate held fewer Securities than it should have for the benefit of all of its custody clients, there may be a shortfall. Any such shortfall may have to be shared pro rata among all clients whose Securities are held by or at the relevant Sub-custodian, Clearing System or delegate and you may not receive your full entitlement. As a result, in the event of such a shortfall, there is a risk that not all Securities deposited by or with the Service Provider and held directly or indirectly with such Sub-custodian, Clearing System or delegate will be returned to the Service Provider or you. In addition, in certain jurisdictions, it may not be possible under applicable local law for Securities belonging to you and held by the Service Provider (directly or indirectly) at a Sub-custodian, Clearing System or delegate to be separately identifiable from the proprietary or other client assets of that Sub-custodian, Clearing System, delegate or the Service Provider.
- 3.12 In connection with the settlement of securities trades operated within an omnibus account structure, You consent to the Service Provider using, or authorising any Sub-custodian to use, your Securities for the account of another customer and vice versa. However, no Securities Account on the books of the Service Provider or a Sub-custodian shall hold Securities which are beneficially owned by the Service Provider or such Sub-custodian, as the case may be. The Service Provider may not use your Securities for its own account.

- 3.13 Where Securities are held outside the United Kingdom, different settlement, legal and regulatory requirements and different practices relating to the separate identification of those Securities may apply which are different from those in the United Kingdom.
- **3.14** The Service Provider hereby notifies you and you agree that your Securities may be registered:
  - (a) in your name;
  - (b) in the name of a Relevant Nominee Company, or a nominee company controlled by a Sub-custodian or a Clearing System on such terms and conditions as any of the foregoing may require; or
  - (c) in the name of the Service Provider, a Sub-custodian or a Clearing System where it is prevented from registering or recording legal title in the way set out in (a) or (b) where the Securities are subject to the law or market practice of a jurisdiction outside the United Kingdom and the Service Provider, Sub- Custodian or a Clearing System has determined that it is in the Your best interests to register or record it in that way, or that it is not feasible to do otherwise, because of the nature of the applicable law or market practice. Where Securities are registered or recorded in this manner, they may not be physically segregated from the assets of the Service Provider, the Subcustodian or Clearing System and in the event of the insolvency of the Service Provider, the Sub-custodian or the Clearing System (or any equivalent procedure in any relevant jurisdiction), as applicable, your assets may not be as well protected from claims made by the creditors of the Service Provider, the Sub-custodian or the Clearing System;
  - (d) or otherwise as permitted by the Client Asset Rules.

provided in each case that legal title to Securities shall be registered or recorded in any relevant record of legal entitlement in accordance with the applicable Client Asset Rules. The title of each Client Securities Account shall make it clear that the Securities in that Account do not belong to You or any affiliated company that is not being treated as Your arm's length client.

- 3.15 The Service Provider accepts the same level of responsibility to you for any Relevant Nominee Company with respect to the requirements of the Client Asset Rules.
- **3.16** The Service Provider may hold any documents of title to a Security:
  - (a) in the physical possession of the Service Provider;
  - (b) with a Sub-custodian in a safe custody account generally designated for the Service Provider's customers' Securities; or
  - (c) as you may otherwise direct with the consent of the Service Provider which may be withheld in its sole discretion. Where this paragraph applies, such documents of title shall be so held at your risk and expense, including the inapplicability of certain provisions of the Service Agreement.
- **3.17** The Service Provider assumes no obligation to review investments in the Accounts or to recommend the purchase, retention or sale of any Property.

#### 4 Duties and Responsibilities

4.1 General Standing Instructions

Subject to and in accordance with your Instructions, the Service Provider is authorised to, and to authorise and instruct Sub-custodians and Clearing Systems to:

- (a) receive and deliver Property, and settle the purchase and sale of Securities transactions, in accordance with the laws, rules, regulations, provisions, customs, practices and procedures in the relevant jurisdiction or market in which the transaction occurs;
- (b) receive all payments of principal and Distributions payable in respect of Property including presenting certificates, coupons and other appropriate documentation to the issuer of Securities or its paying agent and, subject to paragraph (e) below, where such payments or Distributions are received by the Service Provider in relation to Property held in a commingled account, the Service Provider shall apportion such payments or Distributions pro rata among the persons who are entitled to the Property held in such account;
- (c) exchange Securities in temporary or bearer form for Securities in definitive or registered form; effect an exchange of Securities pursuant to non-discretionary exchange offers or changes in the par value of Securities; surrender Securities at maturity or earlier when advised of a call for mandatory redemption and otherwise participate in non-discretionary corporate actions in accordance with customary or established rules, practices and procedures in the relevant jurisdiction or market provided that the Service Provider shall not be liable for failure to so exchange or surrender any Security or take other action:
  - (i) if notice of such exchange or call for redemption or other action was not actually received by the Service Provider from the issuer (in respect of Securities issued in the United States and the United Kingdom) or from one of the nationally or internationally recognised bond or corporate action services to which the Service Provider subscribes or from you; or
  - if, at the time of deposit, any Security so deposited is subject to call, exchange, redemption or similar action, unless specifically instructed to do so by you;
- (d) hold Property in certificated or non-certificated form with the issuer or at any other location;
- (e) pay to you for your own benefit any Cash proceeds arising from the required sale of Securities received as a Distribution in any omnibus account where such Cash proceeds are in respect of your fractional share entitlement<sup>1</sup>;
- (f) upon receipt of notification of a partial redemption, partial payment or other action affecting less than all Securities of a particular class, the Service Provider or the Subcustodian may select the Securities to be tendered in any non-discriminatory manner that either normally uses to make such selections;
- (g) hold any Security in bearer form;
- (h) accept documents in lieu of receipt of Securities;

.

- (i) make, execute, acknowledge and deliver as agent, any and all documents or instruments including but not limited to all declarations, affidavits and certificates of ownership that the Service Provider, in its sole discretion, may determine are necessary or appropriate in carrying out the purposes of the Service Agreement;
- (j) credit the Cash Accounts with all Distributions, provided that, unless the Service Provider in its discretion advances funds in accordance with Clause 7.2, the Cash Accounts will be credited only if the Service Provider or the relevant Sub-custodian has received funds which satisfy the definition of final payment in Clause 7.2 in an amount equal to the value of the sum to be so credited;
- (k) debit the Cash Accounts, even if, in the sole discretion of the Service Provider, such debit creates or increases any overdraft or net debit as the Service Provider determines is appropriate when instructed by you to receive Securities for the Custody Accounts against payment by the Service Provider on behalf of you as contemplated by Clause 7.2;
- (I) deduct or withhold any sum on account of any tax including money held in a Cash Account required or which in the Service Provider's view is required to be deducted or withheld or for which the Service Provider is or is in its view liable or accountable in connection with its services hereunder, by law or practice of any relevant revenue authority of any jurisdiction, and in each case in accordance with the Service Provider's or the Sub-custodian's usual and customary business practice;
- (m) employ suitable agents and consult with, and obtain advice from, suitable agents, including auditors and legal counsel (who may be counsel to you or to the Service Provider) or other advisers in respect of questions and issues relating to the Accounts, and, in the absence of its own fraud, wilful default or negligence, the Service Provider shall incur no liability in acting in accordance with the advice and opinion of such agents or advisers;
- (n) make any payments incidental to or in connection with this Clause 4; and
- (o) exercise all other rights and powers and to take any action it deems necessary or appropriate in carrying out the purposes of the Service Agreement.

#### 4.2 Settlement of Transactions

- (a) In order for the Service Provider or any Sub-custodian to receive and deliver any Security or to settle any Securities transaction in a timely manner, you shall provide the Service Provider with sufficient advance notice of such transaction and all necessary information, as specified by the Service Provider. Unless otherwise agreed by the Service Provider and subject to Clause 7.1, Property shall be credited to the Securities Account, or as the case may be, the Cash Account only when actually received by the Service Provider.
- (b) Settlement of and payment for Securities received for, and delivered from, the Securities Accounts may be made in accordance with the customary or established securities trading or securities processing practices and procedures in the jurisdiction or market in which the transaction occurs, including without limitation, the delivery of Securities to a purchaser, broker, dealer or their respective agents either

- against a receipt for future payment or without any payment (so-called "free delivery").
- (c) For the purpose of settling Securities or foreign exchange transactions, you shall provide the Service Provider with sufficient immediately available funds by such time and date as is required to settle such Securities or foreign exchange transactions in the country of settlement and in the currency to be used to settle such transaction. Without prejudice to Clause 4.1(k) and 4.2(d), in the event that you do not have sufficient funds in the currency required to settle the transaction, you shall deliver to the Service Provider immediately available funds in an amount sufficient to purchase the currency necessary to settle such Securities or foreign exchange transactions. However, the Service Provider shall have no obligation to advance its own funds for the settlement of such transaction.
- (d) The Service Provider may reverse any credit entries to the Securities Accounts or Cash Accounts valued back to the effective date of such credit. You shall repay to the Service Provider the amount advanced by the Service Provider as a result of such credit together with interest thereon calculated from the effective date of such credit.
- (e) The transfer of any Security from an Account to any other account on the Service Provider's books (whether or not such account is in your name) pursuant to your Oral Instructions or Written Instructions shall be conditional and such Security shall be deemed to be held in the Account for all purposes hereunder and subject to the provisions of Clause 7.4 hereof until the Service Provider determines that it has no loans, overdrafts or Losses on its books for you (or until the Service Provider has received satisfactory collateral for all such loans, overdrafts or Losses), all money payments are final and irreversible, and the Service Provider approves such transfer by making appropriate entries evidencing such approval on the Service Provider's books and records. The Service Provider making such entries on its books and records shall constitute the sole means by which such approval shall be given and such transfer shall become effective.

#### 4.3 Discretionary Corporate Actions

(a) Whenever Securities confer optional rights on you (including, but not limited to, warrants, options, conversion and subscription rights, takeovers and other forms of offer or capital reorganisation, redemptions, tenders, options to tender or non-mandatory puts or calls) or provide for discretionary action or alternative courses of action by you, you shall be responsible for making any decisions relating thereto, and for instructing the Service Provider to act. In order for the Service Provider to act, it must receive your Written Instructions at the Service Provider's offices, addressed as the Service Provider may request, by the deadline specified by the Service Provider in its sole discretion from time to time together with any amount which is required to be paid in carrying out any such action. If the Service Provider does not receive such Written Instructions together with any required amount prior to its specified deadlines, the Service Provider shall not be liable for failure to take any action relating to, or to exercise any rights conferred by, such Securities. The Service Provider will not perform any action or accept any instruction as described above in relation to, or which results in, the execution of any Order.

(b) The Service Provider shall endeavour to notify you of such rights or discretionary actions and of the date or dates by when such rights must be exercised or such action must be taken provided that the Service Provider has received, in respect of Securities issued in the United States and the United Kingdom, from the issuer, or, in respect of Securities issued in the United States, the United Kingdom and in any other country, from one of the nationally or internationally recognised bond or corporate action services to which the Service Provider subscribes, timely notice of such rights or discretionary action and of the date or dates on or by which such rights must be exercised or such action must be taken. If the Service Provider does not actually so receive such notice, the Service Provider shall have no liability for failing to so notify you.

#### 4.4 Voting

In relation to all Securities, however registered, the voting rights are to be exercised by you. In relation to all Securities, the Service Provider is authorised and permitted to employ suitable agents to provide global proxy services. Such agents shall be required to post to you any documents (including proxy statements, annual reports and signed proxies) relating to the exercise of such voting rights. Other than providing access to such provider of global proxy services, the Service Provider shall have no obligations in respect of voting for any Securities or any liability in relation to such provider's acts or omissions or for any Loss attributable to, or connected with, providing such access.

#### 4.5 Foreign Exchange Transactions

The Service Provider is authorised to enter into spot or forward foreign exchange contracts ("FX Transactions") with you in connection with the Accounts and may provide such foreign exchange services to you through a Service Provider Affiliates or through Sub-custodians or as may otherwise be agreed by the parties. FX Transactions may be entered into with the Service Provider, any Service Provider Affiliate or any Sub-custodian or any of its affiliated companies or associates acting as principal or otherwise through customary banking channels and they may retain any profits from such FX Transactions. Written Instructions, including standing Instructions, may be issued in relation to FX Transactions, but the Service Provider may establish rules or limitations concerning any foreign exchange facility made available to you. Where the Service Provider, any Service Provider Affiliate or a Subcustodian or any of its affiliated companies or associates enter into an FX Transaction relating to an Account, the terms and conditions of the relevant foreign exchange contracts or Instructions shall apply to such transaction. For the avoidance of doubt, the Service Agreement shall not apply to such transactions save where expressly required by its terms. Neither the Service Provider, nor any Service Provider Affiliate nor any Sub-custodian nor any of its affiliated companies or associates shall be liable for any fluctuations or changes in foreign exchange rates, which shall be the sole risk and liability of you, or be required to substitute one currency for any other currency in a Cash Account.

#### 4.6 Pricing and other data

(a) In providing Market Data related to the Accounts in connection with the Service Agreement, the Service Provider is authorised to use Data Providers. You acknowledge that certain pricing or valuation information may be based on calculated amounts rather than actual market transactions and may not reflect actual market values, and that the variance between such calculated amounts and actual market values may be material. The Service Provider may follow Authorised Person's Instructions in providing pricing or other Market Data, even if such Authorised Person's Instructions direct the Service Provider to override its usual procedures and Market Data sources. The Service Provider shall be entitled to rely without inquiry on all Market Data (and all Authorised Person's Instructions related to Market Data) provided to it. Without prejudice to Clause 4 (*Reports and Statements*) to the Operational Terms, the Service Provider shall not be liable for any Losses incurred as a result of errors or omissions with respect to any Market Data (including but not limited to the accuracy or completeness of such Market Data) used by the Service Provider or you hereunder, even if the Service Provider, in performing services for itself and others (including services similar to those performed for you) receives different valuations of the same or similar Securities of the same issuer. In the event that such pricing services are unable to provide a value or pricing information in respect of Securities and the Service Provider provides values and pricing information, the Service Provider shall so advise you, but shall have no other obligation or liability in respect of such valuation or pricing information.

(b) Market Data may be the intellectual property of the Data Providers, which may impose additional terms and conditions upon your use of the Market Data. The additional terms and conditions can be found within the Data Licensor Terms and you agree to those terms. Certain Data Providers may not permit a customer's directed price to be used. Performance and risk analytic services including, but not limited to certain analytic, accounting, compliance, reconciliation, asset pricing and other services with respect to the Accounts are available provided that you enter into the appropriate separate agreement with the relevant entity within the Service Provider. Performance measurement and analytic services (where subscribed to by you) may use different data sources than those used by the Service Provider to provide Market Data for the Account, with the result that different prices and other Market Data may apply.

#### 4.7 Funds Transfer Services

In relation to Instructions for a Cash transfer, if the Service Provider receives an Instruction to credit or pay a party containing both a name and a unique numeric or alpha-numeric identifier which is acceptable to the Service Provider, the Service Provider and any other bank participating in the Cash transfer may rely solely on such identifier, even if it identifies a party different from the party named. Such reliance on an identifier shall apply to beneficiaries named in such Instructions as well as any financial institution which is designated in such Instructions to act as an intermediary in a Cash transfer. The parties shall be bound by the rules of any Cash transfer system used to effect a Cash transfer.

#### **4.8** Tax Obligations on your Account.

#### (a) Tax Obligations

To the extent that the Service Provider has received the Tax Information within the time stipulated, the Service Provider, may in its sole discretion perform the following services with respect to Tax Obligations:

(i) Unless prohibited by law or regulation, at your reasonable request, the Service Provider will provide to you such information received by the Service Provider in its capacity as custodian that could, in the Service Provider's reasonable belief, assist you or your designee in the submission of any reports or returns with respect to Tax Obligations. An Authorised Person will inform the Service Provider in writing as to which party or parties will receive information from the Service Provider;

- (ii) The Service Provider will, upon receipt of sufficient Tax Information from you (as reasonably determined by the Service Provider), file claims for exemptions or refunds with respect to withheld taxes in those markets where it provides such services and subject to the Service Provider's service level description (in each case as made available to you from time to time). Where you (for whatever reason) fail or neglect to provide the Service Provider with or to review and confirm the Tax Information within the time stipulated by the Service Provider, then such failure or neglect may result in the disapplication of withholding tax relief or the obligation on you to immediately return amounts already refunded by a tax authority. You may, however, elect to appoint your own tax agent to file claims for exemptions or refunds in any or all markets, with advance notice to the Service Provider of such appointment and subject to such terms as separately agreed in writing between you and the Service Provider; and
- (iii) The Service Provider or the applicable Sub-custodian will withhold appropriate amounts, as required by applicable tax laws, with respect to amounts received and is authorised to debit the relevant Account in the amount of a Tax Obligation and to pay such amount to the appropriate taxing authority.

Your receipt of the foregoing services is dependent upon your subscription to the Service Provider's information reporting system, and you will be responsible for enrolling your designated Authorised Persons in such system. You acknowledge that the Service Provider may, at any time, amend the scope of its tax service offering and notice of such changes will be made available to the Service Provider's customers through its information reporting system. Such changes may require additional documentation, attestations or declarations to be entered into by you in order to continue receiving the relevant tax service in a particular market.

You acknowledge that the Service Provider is a provider of services and is not an economic beneficiary of any transaction.

You will be responsible for understanding your Tax Obligations, and will be solely responsible and liable for all Tax Obligations with respect to any Property held on behalf of you and any transaction related thereto.

You will provide the Service Provider with Tax Information to enable the Service Provider to comply with its obligations under any applicable tax laws or with any tax authority enquiry.

You acknowledge and agree that none of the Service Provider nor any Service Provider Affiliate is a tax adviser and none of the Service Provider nor any Service Provider Affiliate will, under any circumstances, provide tax advice to you. You will obtain your own independent tax advice for any tax-related matters or Tax Obligations.

#### (b) Payments

Where the Service Provider receives Instructions to make distributions or transfers out of an Account in order to pay your third party service providers, you acknowledge that in making such payments you are acting in an administrative capacity, and not as the payor, for tax information reporting and withholding purposes.

#### 4.9 Securities in default

The Service Provider shall not be under any duty or obligation to take action to procure collection of any amount if the Securities upon which such amount is payable are in default, or if payment is refused after due demand or presentation, unless and until it has been:

(a) directed to take such action by Written Instructions given by the you; and

### 5 assured to its satisfaction of reimbursement of its costs and expenses in connection with any such action.

#### 5.1 No other duties

The Service Provider's duties and responsibilities are solely those expressly set out in the Service Agreement and it shall not be obliged to perform any services or take any action not provided for in the Service Agreement unless specifically agreed in writing.

#### 6 General Authorisations

- 6.1 Subject to and in accordance with your Instructions, the Service Provider as your agent and for your Account, may place or negotiate orders to buy or sell Securities and other transactions. Such orders and transactions may be placed or negotiated through a Service Provider Affiliate. The Service Provider or any Service Provider Affiliate, acting as principal, may sell such Securities to, or buy such Securities from you. The terms and conditions of the relevant contract to buy or sell Securities or other transactions, as the case may be, shall apply to such transactions. For the avoidance of doubt, the Service Agreement shall not apply to such transactions save where otherwise expressly required by its terms.
- The Service Provider is authorised to disclose information concerning the Accounts and the Property to a Service Provider Affiliate and to Sub-custodians and other providers of services as may be necessary in connection with the administration of the Property or performance of the Service Agreement (including, without limitation, lawyers and accountants for the Service Provider). You agree to supply the Service Provider with any required information if it is not otherwise reasonably available to the Service Provider. The Service Provider may disclose to third parties that it is providing you the services contemplated by the Service Agreement. The Service Provider shall not be held responsible for information held by such persons or of which the Service Provider is not aware by virtue of restricted access or "Chinese Wall" arrangements. If the Service Provider becomes aware of confidential information which prevents it from effecting a particular transaction under the Service Agreement, then the Service Provider may refrain from effecting that transaction.

#### 7 Advances, Overdrafts and Indebtedness

7.1 You understand that when the Service Provider receives an Instruction to deliver Securities against payment or in exchange for Cash (for example in connection with the settlement of a Securities transaction or a redemption, exchange, tender offer or similar corporate action) such payment or exchange of Cash may not occur simultaneously with the delivery of Securities and that the Service Provider may deliver such Securities before actually receiving final payment for such delivery of Securities. Consequently, as a matter of bookkeeping convenience, the Service Provider may credit your Cash Account with Cash equal to the amount the Service Provider anticipates will be received by the Service Provider, a Subcustodian or a Clearing System prior to actual receipt by the Service Provider, a Sub-

custodian or Clearing System of the Cash by way of final payment for such delivery of Securities. The Service Provider may also credit your Cash Account with Cash equal to the amount the Service Provider anticipates will be received by the Service Provider, a Subcustodian or a Clearing System in relation to a Cash Distribution prior to actual receipt by the Service Provider, a Sub-custodian or Clearing System of the Cash Distribution by way of final payment of the Cash Distribution. The risk of non-receipt of payment shall be yours and the Service Provider shall have no liability for it.

- All credits to a Cash Account in anticipation of receipt of final payment of proceeds of sales and redemptions of, and similar corporate actions in respect of, Securities and Distributions shall be conditional upon receipt by the Service Provider of final payment and may be reversed to the extent that final payment is not received. In the event that the Service Provider in its discretion advances funds to you to facilitate the settlement of any transaction, or elects to permit you to use funds credited to a Cash Account in anticipation of final payment, or if you otherwise become indebted to the Service Provider (including indebtedness as a result of overdrafts in the Cash Accounts), upon demand you shall immediately reimburse the Service Provider for such amounts in the same currency if legally available plus any interest on such amounts. For the purposes of the Service Agreement, payment will not be "final" until the Service Provider has received immediately available funds which, under applicable local laws, regulations, rules, customs or practices, are not reversible and not subject to any encumbrance.
- 7.3 In addition to any rights which the Service Provider may have under applicable law or pursuant to other agreements, the Service Provider shall have the right to, and may, without notice to you, combine, consolidate or merge all or any of your Cash Accounts with, and liabilities to, the Service Provider and may set-off from or transfer any Cash in any currency held for you or standing to the credit of any such Accounts in or towards the satisfaction of any liability of yours (whether actual or contingent, present or future) to the Service Provider whether arising from or as a result of an FX Transaction, a Securities Transaction, the Service Agreement or otherwise, and may do so notwithstanding that Cash held for you or the balances of such Accounts may be held or deposited at different branches of the Service Provider or at any Sub-custodian and may not be expressed in the same currency as the currency of your liability to the Service Provider and the Service Provider may affect any necessary conversions at the Service Provider's own rate of exchange then prevailing.
- 7.4 To secure repayment of your obligations under the Service Agreement, as well as any other obligations of yours to the Service Provider, the Service Provider shall:
  - (a) in addition to any general lien or other rights to which the Service Provider may be entitled under any applicable laws, have a first lien on all Securities and shall (notwithstanding any other terms of the Service Agreement) have a right to withhold redelivery to or to the order of you of the Securities under the control of the Service Provider or of any Sub-custodian, Clearing System or agent, including without limitation, a general right of retention on all Securities in the Securities Accounts, to the extent only of the amount of such obligations for which you are from time to time liable to the Service Provider under or in connection with the Service Agreement or otherwise; and

- be entitled to sell, transfer or assign or otherwise realise the value of any such Securities and to apply the proceeds in satisfaction of such obligations.
- 8.1 You acknowledge and agree that Clearing Systems and Sub-custodians may have a lien, right of set off, security interest (statutory or otherwise) over or rights of retention and sale in respect of Securities credited to the Securities Accounts or Cash credited to the Cash Account in relation to claims for payment of obligations owed to them (including administration and safe custody charges) as provided in the applicable Clearing System agreement or Sub-custodian Agreement.
- You shall not take any action in relation to the Securities which is inconsistent with the rights granted to the Service Provider under the Service Agreement or any debenture under Clause 7.4 and shall take all appropriate action at the request of the Service Provider which may be required to assure the superiority or priority of any lien or other rights granted hereunder or the debenture in favour of the Service Provider.
- **8.3** The Service Provider is hereby authorised to effect any necessary currency conversions pursuant to the Service Agreement at the Service Provider's own rate of exchange then prevailing.
- 8.4 Unless required by law, neither the Service Provider nor its nominees shall be bound by or recognise any Security interest or similar entitlement to any Securities or Cash held for you for the benefit of any person, other than your entitlement under the Service Agreement.

#### 9 Instructions

- 9.1 Unless otherwise specified in the Service Agreement, Instructions shall be in English and may be given either orally or in writing (including electronic transmission which may include transmission through Computer Services). You accept that some methods of communication are not secure and the Service Provider shall incur no liability for receiving Instructions. All Instructions shall be delivered to the address set out in Schedule 3, Part B to the Operational Terms (Notices), which address may be changed upon five Business Days' prior written notice to the other party.
- 9.2 You shall provide the Service Provider with a certificate in the form set out Schedule 3 to the Operational Terms containing the names and specimen signatures of each Authorised Person. The Service Provider is authorised to comply with and rely upon any such Instructions believed by it to have been sent or given by an Authorised Person. You may amend such certificate or add any person to or delete any person from such certificate by delivering a replacement certificate to the Service Provider. However until the Service Provider actually receives such replacement certificate, the Service Provider may rely upon and shall incur no liability for relying upon the original certificate.
- 9.3 You shall use all reasonable endeavours to ensure that Instructions transmitted to the Service Provider pursuant to the Service Agreement are correct and complete. Any Instructions shall be conclusively deemed to be valid Instructions from you to the Service Provider for the purposes of the Service Agreement. The Service Provider may in its sole discretion decline to act upon any Instructions which are insufficient or incomplete or are not received by the Service Provider in sufficient time for the Service Provider to act upon such Instructions. If you elect to give the Service Provider Oral Instructions, and the Service Provider in its sole discretion elects to act upon such Oral Instructions, the Service Provider's understanding of such Oral Instructions shall be deemed controlling. The Service Provider

shall not be liable for any Losses arising directly or indirectly from the Service Provider's reliance upon and compliance with such Oral Instructions notwithstanding that such Oral Instructions conflict or are inconsistent with subsequent Written Instructions.

#### 9.4 You undertake:

- (a) to instruct the Service Provider to hold Client Securities in a Client Securities Account, Client Cash in a Client Cash Account, Proprietary Securities in a Proprietary Securities Account and Proprietary Cash in a Proprietary Cash Account; and
- 10 not to instruct or otherwise direct the Service Provider to hold or give Instructions which may result in the Service Provider holding Client Securities in a Proprietary Securities Account, Client Cash in a Proprietary Cash Account, Proprietary Securities in a Client Securities Account or Proprietary Cash in a Client Cash Account.

#### 11 Indemnity

You shall be liable for and agree to indemnify and keep indemnified and release and hold harmless the Service Provider and each of its nominee companies and the Service Provider Affiliates on demand (on an after-tax basis) against and from any and all Losses howsoever arising from or in connection with the Service Agreement or the performance of their duties under the Service Agreement including disputes between the parties or the enforcement of the Service Agreement, provided that nothing in the Service Agreement shall require that the Service Provider or its nominee companies or the Service Provider Affiliates be indemnified for their respective fraud, negligence or wilful default.

#### 12 Governing Law and Jurisdiction

- **12.1 Governing Law**: The Service Agreement and this Service Module and any non-contractual obligations arising out of or in connection with them are governed by, and shall be construed in accordance with, English law.
- 12.2 Jurisdiction: The courts of England are to have exclusive jurisdiction to settle any disputes that may arise out of or in connection with the Service Agreement and this Service Module (including non-contractual disputes or claims) and accordingly any legal action or proceedings arising out of or in connection with the Service Agreement and this Service Module ("Proceedings") shall be brought in such courts. The parties irrevocably submit to the exclusive jurisdiction of such courts and waive any objection to Proceedings in such courts whether on the ground of venue or on the ground that the Proceedings have been brought in an inconvenient forum.

## Schedule 1 to the Dealer Custody (English law) Service Module - Definitions

Whenever used in this Service Module, the following terms shall have the meaning set out below:

- "Accounts" the Cash Accounts and the Securities Accounts.
- "Authorised Person" any person who is designated in writing by you from time to time to give Instructions to the Service Provider under the terms of the Service Agreement.
- "Business Day" any day on which the Service Provider and any Sub-custodian or Clearing System is open for business in its respective markets.
- "Cash" the money and currency of any jurisdiction which the Service Provider accepts for deposit in a Cash Account.
- "Cash Account" a cash account as defined in Clause 3.3(b).
- "Clearing System" Clearstream (Luxembourg), Euroclear, Crest, the Federal Reserve/Treasury Book-Entry System, The Depository Trust and Clearing Corporation or securities depository, bookentry system or a recognised investment exchange (as defined in the Client Asset Rules) or safekeeping services with which the Service Provider or a Sub-custodian maintains accounts for the purpose of settling Transactions and holding Securities pursuant to the provisions of the Service Agreement.
- "Client Asset Rules" shall mean the client asset rules as set out in the Client Assets sourcebook of the FCA Rules.
- "Client Money Distribution and Transfer Rules" means the client money distribution and transfer rules as set out in Chapter 7A of the Client Asset Rules.
- "Client Money Rules" means the client money rules as set out in Chapter 7 of the Client Asset Rules.
- "Client Cash" Cash which you are required to hold pursuant to the Client Money Rules.
- "Client Cash Account" a Cash Account in which Client Cash is held.
- "Client Securities" Securities which are legally or beneficially owned by an underlying customer.
- "Client Securities Account" a Securities Account in which Client Securities are held.
- "Computer Services" those electronic access services and electronic communication facilities (if any), which may also allow an Authorised Person to transmit Instructions through an electronic access system offered by the Service Provider as may be agreed from time to time pursuant to the Service Agreement.
- "Data Licensor Terms" the set of terms and conditions (as may be amended by the Service Provider without notice to you) available at http://bnymellon.com/products/assetservicing/vendoragreement.pdf or any successor website the address of which is provided by the Service Provider to you.
- "Data Providers" pricing vendors, brokers, dealers, investment managers, Authorised Persons, Sub-custodians, Clearing System and any other person or entity providing Market Data to the Service Provider.

"Distribution" all interest, dividends and other income distributed or paid in respect of Cash and Securities.

**"FCA"** the UK Financial Conduct Authority (and any successor regulatory authority), whose registered office is at 12 Endeavour Square, London, E20 1JN.

"FCA Rules" the rules promulgated by the FCA under FSMA as amended or replaced from time to time.

"Financial Instrument" has the meaning ascribed to it in MiFID II.

"FSMA" the Financial Services and Markets Act 2000.

"FX Transactions" as defined in Clause 4.5.

"Instructions" Oral Instructions and Written Instructions.

"Losses" any and all claims, losses, liabilities, damages, costs, expenses and judgments (including reasonable legal fees and expenses) sustained by either party.

"Market Data" pricing and other data related to Securities and other assets including, but is not limited to, security identifiers, valuations, bond ratings, classification data, and other data received by the Service Provider from Data Providers.

"MiFID II" together the Markets in Financial Instruments Directive (Directive 2014/65/EU), Markets in Financial Instruments Regulation (Regulation 600/2014) and the associated EU regulatory and implementing technical standards and implementing law and regulation in the EEA states.

"Oral Instructions" verbal instructions or directions received by the Service Provider in accordance with Clause 9 (*Instructions*) from an Authorised Person or a person reasonably believed by the Service Provider to be an Authorised Person.

"Order" any Instructions received by the Service Provider from you or on your behalf or generated by the Service Provider on your behalf in relation to a Transaction.

"Property" Securities, Cash and Distributions.

"Proprietary Cash" Cash which is not Client Cash.

"Proprietary Cash Account" a Cash Account in which Proprietary Cash is held.

"Proprietary Securities" Securities which are not Client Securities.

"Proprietary Securities Account" a Securities Account in which Proprietary Securities are held.

"Relevant Nominee Company" a nominee company controlled by the Service Provider or a Service Provider Affiliate.

"Securities" all debt and equity securities and other instruments and intangible assets (including instruments representing the right to receive, purchase or subscribe to the foregoing or representing other rights or interests in the foregoing) as may be agreed upon from time to time by the Service Provider and you and which shall from time to time be delivered to or received by the Service Provider and/or any Sub-custodian for deposit in the Securities Accounts.

"Securities Account" a securities account as defined in Clause 3.3(a).

"Service Provider Affiliate" shall mean any entity in which The Bank of New York Mellon Corporation (a Delaware corporation with registered office at One Wall Street; New York, NY 10286,

U.S.A) controls (directly or indirectly) an interest of no less than 30% in the voting stock or interests in such entity.

"Sub-custodian" a bank or financial institution (other than Clearing System) identified in Schedule 3, Part D to the Operational Terms (*Global Custody Network Countries and Sub-Custodians*) (as amended from time to time) which is used by the Service Provider in connection with the purchase, sale, transfer or custody of Securities hereunder.

"Sub-custodian Agreement" a written agreement between the Service Provider and a Sub-custodian in relation to the Sub-custodian's appointment as such.

"Tax Information" shall mean all accurate, relevant and necessary information with respect to the Accounts or with respect to your identification or classification for purposes of Tax Obligations, in each case as may be required by applicable tax laws or by a tax authority inquiry, or as may be requested by the Service Provider in connection with the matters in clause 4.8

"Tax Obligations" taxes, withholding, certification and reporting requirements, claims for exemptions or refund, interest, penalties, additions to tax and other related expenses.

"Transaction" a transaction resulting in the acquisition or disposal of a Financial Instrument.

**"Written Instructions"** any written notices, directions or instructions received by the Service Provider in accordance with Clause 7 (*Instructions*) from an Authorised Person or from a person reasonably believed by the Service Provider to be an Authorised Person or received through the Computer Services.